UNITED STATES SECURITIES AND EXCHANGE COMMISSION Washington, D.C. 20549

FORM 8-K

CURRENT REPORT Pursuant to Section 13 OR 15(d) of The Securities Exchange Act of 1934

Date of report (Date of earliest event reported): June 25, 2013

ARCH THERAPEUTICS, INC.

(Exact name of registrant as specified in its charter)

Nevada333-17888346-0524102(State or other jurisdiction of incorporation)(Commission File Number)(I.R.S. Employer Identification No.)

One Broadway, 14th Floor Cambridge, Massachusetts

02142

(Zip Code)

(Address of principal executive offices)

Registrant's telephone number, including area code: (617) 475-5254

Pembroke House 28-32 Pembroke St Upper Dublin 2, Ireland

(Former name or former address, if changed since last report.)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions (see General Instruction A.2. below):
☐ Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)
□ Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)
☐ Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))
☐ Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))

FORWARD-LOOKING STATEMENTS

This Current Report on Form 8-K contains forward-looking statements that involve risks, uncertainties and assumptions. If such risks or uncertainties materialize or such assumptions prove incorrect, our results could differ materially from those expressed or implied by such forward-looking statements and assumptions. In some cases, you can identify forward-looking statements by terminology such as "may", "should", "expects", "plan", "anticipate", "believe", "estimate", "predict", "potential" or "continue" or the negative of these terms or other comparable terminology. All statements made in this Form 8-K other than statements of historical fact are statements that could be deemed forward-looking statements, including without limitation statements about our business plan, our plan of operations and our need to obtain future financing. These statements are only predictions and involve known and unknown risks, uncertainties and other factors, including the risks in the section entitled "Risk Factors" and the risks set out below, any of which may cause our or our industry's actual results, levels of activity, performance or achievements to be materially different from any future results, levels of activity, performance or achievements expressed or implied by these forward-looking statements. These risks include, by way of example and not in limitation, risks related to:

- General economic and business conditions;
- Our ability to continue as a going concern;
- Our ability to obtain financing necessary to operate our business;
- Our limited operating history;
- Our ability to recruit and retain qualified personnel;
- Our ability to manage future growth;
- Our ability to develop, obtain required approvals for and commercialize our product candidates;
- Our ability to maintain and protect our intellectual property;
- Our ability to successfully complete potential acquisitions and collaborative arrangements; and
- Other factors discussed under the section entitled "Risk Factors".

Although we believe that the expectations reflected in the forward-looking statements are reasonable, we cannot guarantee future results, levels of activity or performance. These forward-looking statements speak only as of the date of this Current Report on Form 8-K. Except as required by applicable law, we do not intend to update any of these forward-looking statements.

As used in this Current Report on Form 8-K, unless otherwise indicated the terms the "Company", "Arch Therapeutics", "we", "us" and "our" refer to Arch Therapeutics, Inc., a Nevada corporation, and its subsidiary, unless the context otherwise requires.

We have pending trademark applications for AC5TM, Crystal Clear SurgeryTM, NanoDrapeTM and NanoBioBarrierTM. All other trademarks, trade names and service marks included in this Current Report on Form 8-K are the property of their respective owners.

Item 2.01 Completion of Acquisition or Disposition of Assets.

The Merger and Related Transactions

The Merger

As previously disclosed, on May 10, 2013, we entered into an Agreement and Plan of Merger (the "Merger Agreement") with Arch Biosurgery, Inc. (formerly known as Arch Therapeutics, Inc.), a Massachusetts corporation ("ABS"), and Arch Acquisition Corporation, a Massachusetts corporation and our wholly-owned subsidiary formed for the purpose of the transaction ("Merger Sub"). The Merger Agreement provided for the merger of Merger Sub with and into ABS (the "Merger"), with ABS surviving the Merger as our wholly owned subsidiary, upon the terms and subject to the conditions set forth in the Merger Agreement.

On June 26, 2013, following the satisfaction or waiver of the conditions set forth in and otherwise in accordance with the terms of the Merger Agreement, the Merger was consummated and Merger Sub merged with and into ABS. As a result of the closing of the Merger, we have abandoned our prior business plan and we are now pursuing the operations of ABS as a life science company developing product candidates in the surgical hemostasis field.

The Merger Agreement includes customary representations, warranties and covenants made by us, Merger Sub and ABS as of specific dates. The assertions embodied in those representations and warranties were made solely for purposes of the Merger Agreement and are not intended to provide factual, business, or financial information about us, Merger Sub and ABS. Moreover, some of those representations and warranties (i) may not be accurate or complete as of any specified date, (ii) may be subject to a contractual standard of materiality different from those generally applicable to shareholders or different from what a shareholder might view as material, (iii) may have been used for purposes of allocating risk among us, Merger Sub and ABS, rather than establishing matters as facts, and/or (iv) may have been qualified by certain disclosures not reflected in the Merger Agreement that were made to the other party in connection with the negotiation of the Merger Agreement and generally were solely for the benefit of the parties to the Merger Agreement. The Merger Agreement should not be read alone, but should instead be read in conjunction with the other information regarding us and our business that has been, is or will be contained in, or incorporated by reference into, the Forms 10-K, Forms 10-Q, Forms 8-K, and other documents that we file with the Securities and Exchange Commission (the "SEC"). The description of the Merger Agreement set forth herein is qualified in its entirety by reference to the full text of the Merger Agreement, which is filed as Exhibit 2.1 to the Current Report Form 8-K we filed with the SEC on May 13, 2013 and is incorporated herein by reference.

The Coldstream Financing

In contemplation of the Merger, on April 19, 2013, we entered into a financing agreement (the "Financing Agreement") with Coldstream Summit Ltd. ("Coldstream"), pursuant to which we agreed to issue and sell, and Coldstream agreed to purchase or assist in securing the purchase of, \$2,000,000 worth of units in a private offering within the 12 month period following the closing of the Merger (the "Coldstream Financing"). Each unit issued in the Coldstream Financing is to be sold at a price of \$0.50 per share and is to consist of (i) one share of our common stock and (ii) one warrant to purchase one share of our common stock at an exercise price of \$0.75 per share and with a term of 12 months. As of the date of this Current Report on Form 8-K, we have issued and sold units consisting of 2,500,000 shares of our common stock and warrants to purchase 2,500,000 shares of our common stock in the Coldstream Financing, for aggregate gross proceeds of \$1,250,000. The proceeds of the Coldstream Financing are being used for the funding of our and ABS's ongoing business and operations. As previously disclosed, pursuant to the terms of the Merger Agreement, all such proceeds raised to date were advanced to ABS prior to the closing of the Merger.

Post-Merger Company Ownership

As set forth in the Merger Agreement, upon the closing of the Merger, all of the issued and outstanding capital stock and convertible notes and warrants of ABS were cancelled automatically and the holders thereof became entitled to receive an aggregate of 14,645,212 shares of the Company's common stock. That number of shares was negotiated and agreed to by the Company and ABS prior to entering into the Merger Agreement. Upon the closing of the Merger, the former shareholders of ABS are entitled to receive two and one-half shares of our common stock for each share of common stock of ABS held by them immediately prior to the closing of the Merger. After giving effect to the closing of the Merger and including the shares and warrants issued in the Coldstream Financing as of the date hereof and to be issued in the Coldstream Financing over the 12 month period following the closing of the Merger, the securities of the Company (on a fully diluted basis) are owned as follows:

- Former shareholders of ABS hold 5,645,212 shares of the Company's common stock, or approximately 7.8% of the Company on a fully diluted basis;
- Former holders of convertible promissory notes of ABS hold 9,000,000 shares of the Company's common stock, or approximately 12.5% of the Company on a fully diluted basis;
- Dr. Norchi and Dr. Dhillon, or their respective designees over which they hold a controlling interest, collectively hold 18,579,449 shares of the Company's common stock (including the shares of the Company's common stock they are entitled to receive as former shareholders and noteholders of ABS), or approximately 25.8% of the Company on a fully diluted basis;
- 7,825,388 shares of the Company's common stock initially reserved for issuance to employees, directors and consultants under the Arch Therapeutics, Inc. 2013 Stock Incentive Plan (the "Plan"), representing approximately 10.9% of the Company on a fully diluted basis:
- Stockholders of the Company prior to the closing of the Merger, including consultants of the Company that were issued an aggregate of 1,500,000 shares of our common stock on June 18, 2013 in restricted stock grants outside of the Plan, hold 21,500,000 shares of the Company's common stock, or approximately 29.9% of the Company on a fully diluted basis; and
- Current and future investors in the Coldstream Financing will hold 4,000,000 shares of the Company's common stock and warrants to acquire 4,000,000 shares of the Company's common stock, or approximately 11.1% of the Company on a fully diluted basis.

Lock-Up Restrictions

In connection with the Merger, shares of our common stock received by (i) substantially all of ABS's former shareholders and noteholders as a result of the Merger, including all shares held by Dr. Norchi and Dr. Dhillon (and their respective designees) that were received in connection with the Merger, (ii) recipients of restricted stock grants of an aggregate of 1,500,000 shares made outside of our Plan, and (iii) recipients of certain non-qualified stock options granted under our Plan to purchase an aggregate of 3,984,212 shares, are subject to certain lock-up restrictions that restrict the sale or other transfer of such shares for a certain period of time following the closing of the Merger. For the 18 months following the closing of the Merger, all such shares will be subject to those lock-up restrictions. Thereafter, 25% of such shares will be released from the lock-up restrictions.

Accounting Treatment of the Merger

For financial reporting purposes, the Merger represents a "reverse merger" rather than a business combination and ABS is deemed to be the accounting acquirer in the transaction. Consequently, the assets and liabilities and the historical operations that will be reflected in the Company's future financial statements will be those of ABS. The Company's assets, liabilities and results of operations will be consolidated with the assets, liabilities and results of operations of ABS after consummation of the Merger, and the historical financial statements of the Company before the Merger will be replaced with the historical financial statements of ABS before the Merger in all future filings with the SEC.

FORM 10 INFORMATION

Immediately prior to the closing of the Merger, we were deemed a shell company as defined in Rule 12b-2 promulgated under the Securities Exchange Act of 1934 (the "Exchange Act"). Item 2.01(f) of Form 8-K requires that, under those circumstances, a registrant must disclose the information that would be required if the registrant were filing a general form for registration of securities on Form 10 under the Exchange Act. Accordingly, we are providing such information for the combined enterprises of the Company and ABS below.

BUSINESS

Corporate Overview

We were incorporated under the laws of State of Nevada on September 16, 2009 as Almah, Inc. On May 10, 2013, we entered into the Merger Agreement with ABS and Arch Acquisition Corporation, our wholly owned subsidiary formed for the purpose of the transaction, pursuant to which Arch Acquisition Corporation merged with and into ABS and ABS thereby became our wholly owned subsidiary. In contemplation of the Merger, effective May 24, 2013 we increased our authorized common stock from 75,000,000 shares to 300,000,000 shares and effected a forward stock split, by way of a stock dividend, of our issued and outstanding shares of common stock at a ratio of 11 shares to each one issued and outstanding share, and effective June 5, 2013, we changed our name from Almah, Inc. to Arch Therapeutics, Inc. and changed the ticker symbol under which our common stock trades on the OTC Bulletin Board from "AACH" to "ARTH". All share amounts of our common stock referenced in this Current Report on Form 8-K give effect to the 11-for-1 forward stock split described above, including those applicable to periods prior to the forward stock split.

ABS was incorporated under the laws of Commonwealth of Massachusetts on March 6, 2006 as Clear Nano Solutions, Inc. On April 7, 2008, ABS changed its name to Arch Therapeutics, Inc., and on August 28, 2009, ABS increased its authorized common stock, no par value, from 275,000 shares to 1,275,000 shares. Effective upon the closing of the Merger, ABS changed its name from Arch Therapeutics, Inc. to Arch Biosurgery, Inc.

The Merger closed on June 26, 2013, and as a result we have abandoned our prior business plan and are now pursuing the business of ABS as our sole business. The following is a discussion of the business of ABS that we are now pursuing. References to "we", "us" and "our" in the following discussion refer to the Company and its subsidiary, ABS, as a combined enterprise.

Our Current Business

We are life science medical device company in the development stage with limited operations to date. We aim to develop products that make surgery and interventional care faster and safer by utilizing a novel approach to stop bleeding (referenced as "hemostasis"), control leaking (referenced as "sealant"), and provide other advantages during surgery and trauma care. Our core technology is based on a self-assembling peptide solution that creates a physical, mechanical barrier, which could be applied to bleeding organs or wounds to seal leaking blood and other fluids. We believe our technology could support an innovative platform of potential products in the field of stasis and barrier applications. Our first product candidate, AC5TM, is designed to achieve hemostasis in minimally invasive and open surgical procedures, and we hope to develop other product candidates in the future based on our technology platform aimed at stopping bleeding and sealing other leaking fluids during surgical and other procedures.

Our Core Technology

Our technology platform is based on self-assembling synthetic peptides. Our plan and business model is to develop products that apply that core technology to human bodily fluids and connective tissues.

Our primary product candidate, AC5, relies on this technology to achieve hemostasis during surgical procedures. We envision developing other product candidates in the future based on our core technology, examples of which could include, for instance, products for specialty surgery, burn and trauma care, wound care, military applications, and consumer care.

We have devoted much of our operations to date to the development of our core technology, including selecting our lead product composition, conducting initial safety and other related tests, generating scale-up, reproducibility and manufacturing methods, and developing and protecting the intellectual property rights underlying our technology platform. We have one key intellectual property licensor, the Massachusetts Institute of Technology ("MIT"), from which we license certain of our important intellectual property rights, and have made, and hope to continue to make, advances on our core technology to further refine and improve its use and functionality, further develop our intellectual property rights, and ultimately produce an expanded portfolio of potential product candidates.

AC5

Our first product in development, AC5, is a biocompatible synthetic peptide comprising naturally occurring amino acids. When applied to a wound, AC5 intercalates into the interstices of the connective tissue where it self-assembles into a physical, mechanical nanoscale structure that provides a barrier to leaking substances, such as blood.

The results of early data from preclinical animal tests have shown that AC5 achieves hemostasis quickly and effectively. AC5 can be directly applied as a liquid or sprayed, making it user-friendly and able to conform to irregular wound geometry, and is not sticky or glue-like, making it ideal for use in the setting of minimally invasive laparoscopic surgeries. Further, AC5 is transparent, which should make it easier for a surgeon or other healthcare providers to maintain a clear field of vision during a surgical procedure and prophylactically stop bleeding as it starts, which we call Crystal Clear SurgeryTM.

Completed Preclinical Development

We are in the early stages of our planned clinical program for AC5. To date, only preclinical animal tests have been performed. In order to achieve the approvals and certifications we need to market and sell AC5, significant additional testing, including conducting human clinical trials, will be required.

Preclinical testing to date has been conducted in a number of settings. One of the co-founders of ABS and a co-founding inventor of certain of our technology, Dr. Rutledge Ellis-Behnke, performed a significant portion of the preclinical animal experimentation conducted to date during his time at the Massachusetts Institute of Technology in the Department of Brain and Cognitive Sciences from 2001 through 2005 and the University of Hong Kong Faculty of Medicine in the Department of Anatomy from 2004 through 2009, with overlap between the two institutions in 2004 and 2005. Dr. Ellis-Behnke and his colleagues also outsourced certain experiments to third parties. Some of the most significant findings from Dr. Ellis-Behnke's studies have been published. Additionally, on a fee for service basis, ABS engaged a private third party facility in Massachusetts where certain preclinical animal experiments were performed with the assistance of ABS consultants. ABS also engaged a biomedical animal research company in Massachusetts to perform certain preclinical animal studies. Further, through collaboration with the National University of Ireland system, preclinical animal and tissue experiments have been performed in Dublin and Cork, Ireland.

In the preclinical animal tests conducted to date, AC5 has demonstrated improved average time to hemostasis ("TTH") when applied to animal brains, spinal cords and livers. Those tests have tested TTH when using AC5 during a range of surgical procedures compared to TTH when using a control substance, a saline control substance, a control peptide, and a cautery control substance during those same procedures. The results of those tests have shown a TTH of under 15 seconds when AC5 was applied, compared to a TTH ranging from 80 to 300 seconds when various control substances were applied, depending on the nature of the control substance and procedure performed. In tests to date, AC5 has also demonstrated biocompatibility and normal healing of tissue treated with the product. Further, animals whose liver, spleen, femoral artery, eye or brain was treated with AC5 have shown no ill-effects. We believe that the peptide degrades into the naturally occurring amino acids from which it was originally synthesized, which are molecules that already exist in large quantities in the body.

Our plans in the near-term are to focus our efforts on the development of AC5 by pursuing additional preclinical studies and preparing for future clinical trials.

Development and Commercialization Strategy

Our present business model is to operate with a relatively small internal team of key personnel and engage third party service providers to conduct larger scale research, development and manufacturing activities. Our internal team collectively has a broad range of expertise and experience working with and managing third party vendors. This general approach enables us to utilize the services of third party entities that are experts in each aspect of our operations, while preserving capital and efficiencies by avoiding certain internal scale-up costs and duplication of resources.

Research and Development; Manufacturing

Use of Third Party Relationships

To date, we have engaged third party laboratory facilities run by peptide experts in Europe and the U.S. to perform preclinical research and development activities. Those engagement have enabled us to properly develop our primary product candidate, as well as generate appropriate analytical methods, scale-up, and other procedures that we intend to use as a "blueprint" for a third party manufacturer to make the product on a larger scale for purposes of further clinical testing and ultimately commercialization.

We are currently preparing for that transition to traditional contract manufacturing and related organizations. We have commenced discussions with manufacturers operating with the current good manufacturing practices ("cGMP") required by applicable regulatory agencies, which we would engage to scale up and produce clinical formulation material to be used for final preclinical testing and clinical trials.

Manufacturing Methods

We believe that the manufacturing methods used for a product, including the type and source of ingredients and the burden of waste byproduct elimination, are important determinants of its opportunity for profitability. Industry is keenly aware of the downsides of technologies that rely on expensive biotechnology techniques and facilities for manufacture, onerous and expensive programs to eliminate complex materials, or ingredients that are sourced from the complicated process of human or other animal plasma separation, since those products typically are expensive, burdensome to produce, and at greater risk for failing regulatory oversight.

The manufacturing methods that we envision would be utilized to produce AC5 and other potential future product candidates rely on synthetic organic chemistry. The technology, skill and know-how involved with those methods are important, but the required manufacturing equipment is widely available. Furthermore, improvements in relevant synthetic manufacturing techniques in the past several years have reduced their complexity and cost, while increasing large scale cGMP capacity. In addition, as a result of increased demand for amino acids in recent years, the cost of obtaining amino acid raw materials has decreased. Further, our planned product candidates, including AC5, will be synthesized of naturally occurring ingredients that are not sourced from humans or other animals, but do exist in humans in their natural state. That type of ingredient is often more likely to be categorized as "generally recognized as safe", or "GRAS", by the U.S. Food and Drug Administration ("FDA"), and can convey a lower risk of adverse effects.

We believe that our pursued manufacturing methods and ingredients will make our choice of third party manufacturers important, as we will need to select service providers with sufficient expertise with synthetic organic chemistry manufacturing, but will benefit from the lack of expensive equipment, technology and materials required and the naturally occurring ingredients used in the manufacturing process.

Regulatory

Medical Device Classification

Although the FDA and other regulatory authorities or related bodies will finally determine the classification of AC5, we believe that our primary product candidate meets the criteria for a medical device. Generally, a product is a medical device if it requires neither metabolic nor chemical activity to achieve the desired effect. Furthermore, a medical device can achieve its desired effects without requiring a body (animal/human), whereas a drug or a biologic requires a body. The AC5 mechanism of assembly into a barrier can occur outside of a body and is accordingly consistent with the medical device definition.

Medical devices in the European Union ("EU") and the U.S. are classified along a spectrum. We anticipate that AC5 will be a Class III medical device in these jurisdictions, subject to the process for obtaining a CE mark in the EU and the premarketing authorization process in the U.S. While the Class III status is a higher-level classification than for devices not comprised of novel materials and involves additional procedure and regulatory scrutiny of the product candidate to obtain approvals, it provides less regulatory ambiguity.

Biocompatibility Tests and Clinical Trials

Before initiating any human clinical trials, we will need to assess the biocompatibility of AC5. Standard required tests to assess biocompatibility, as set forth in ISO 10993 issued by the International Organization for Standardization, include:

- in vitro cytotoxicity;
- in vitro blood compatibility;
- in vitro Ames assay (mutagenic activity);
- irritation / intracutaneous reactivity;
- sensitization (allergenic reaction);
- implantation (performed on devices that contact the body's interior);
- pyrogenicity;
- · systemic toxicity; and
- in vitro chromosome aberration assay (structural chromosome changes).

We have not commenced formal biocompatibility studies for AC5. However, Dr. Ellis-Behnke and his colleagues, on a fee for service basis, engaged a third party Massachusetts-based facility to perform certain in vitro and in vivo biocompatibility and toxicology studies on what was an earlier version of our composition; such tests illustrated no evidence of toxicity and portions of the results have been published. Further, with the assistance of ABS personnel and consultants, certain large relative dose pilot tests were performed in rodents at a private third party facility in Massachusetts, and no abnormal behavior or pathology was observed from such tests.

Following completion of biocompatibility tests for AC5, assuming successful results of those tests, we expect that we will focus on conducting required human clinical trials. We currently plan to conduct the First in Human clinical trial on AC5 in Europe. Assuming successful results of the trial, we expect that we will then pursue a CE mark, the required European approval to market and commercialize a medical device such as AC5, prior to pursuing approval by the U.S. FDA. Based on precedent, we believe that the EU will require one clinical trial to obtain a CE mark for AC5.

When properly harmonized, the FDA may accept non-U.S. jurisdiction clinical trial data for a product in support of a FDA application for the same product, and we hope to use the data from our planned initial clinical trial to be conducted in the EU in this fashion. Similarly, any subsequent American clinical trials could help to broaden the scope and indications of any European label for AC5 that we may achieve.

In order to obtain a broad label for AC5 in the U.S., we believe that the FDA will require safety and efficacy data in three different tissue types. We hope to utilize the data from our planned initial clinical trial in the EU to contribute to the satisfaction of some of those FDA requirements.

We also intend to pursue other potential indications for AC5 and/or other potential product candidates based on our technology platform, which we may pursue either opportunistically or once regulatory approval is obtained for our initial surgical hemostasis product candidate.

Commercialization

We are in the process of developing a long-term commercialization plan for our product candidates. That plan could entail entering into one or more strategic partnerships in connection with product commercialization, our direct performance of commercialization activities, or some combination of those alternatives. Based on our current general approach and strategy of utilizing the expertise and resources of third party service providers while maintaining a small internal team, we currently expect that we may pursue some degree of strategic collaborations or partnerships with third parties, which could include licensing arrangements, distribution and supply partnerships, engagement of external regulatory experts and/or marketing and sales teams, among other types of potential relationships. We presently believe that partnerships or collaboration relationships could improve our ability to obtain regulatory approval for our product candidates and attain market acceptance for and profitable sales of those product candidates, and that our current and planned activities and milestones relating to AC5 are well-aligned with the needs of the market and potential partners and collaborators that wish to enter or expand their presence in our target markets.

We envision the potential future customers in the marketplace for AC5 and any other hemostatic or sealant agent we may pursue will include surgeons and other doctors, government agencies such as the Department of Defense, hospital and operating room management and ambulance and other trauma specialists.

Plan of Operations

Our long-term business plan includes the following goals:

- conducting successful clinical trials on AC5;
- obtaining regulatory approval or certification of AC5 in the EU, the U.S., and other jurisdictions;
- expanding our intellectual property portfolio;
- developing appropriate third party relationships to manufacture, distribute, market and otherwise commercialize AC5; and
- developing additional product candidates in the hemostatic and sealant field.

With respect to our goals relating to AC5, we currently project requiring between \$6,000,000 and \$8,000,000 of additional capital to complete the milestones to obtain regulatory approval in Europe and launch AC5 in the European market. We expect that obtaining regulatory approvals and launching AC5 in the U.S., including conducting additional required clinical trials, would require at least an additional \$9,000,000 in capital.

In furtherance of our long-term business goals, we expect to focus on the following activities during the remainder of calendar year 2013 and calendar year 2014:

- conducting formal biocompatibility studies;
- participating in EU and, subsequently, U.S. regulatory meetings;
- preparing for initial clinical trials, including developing clinical trial protocols;
- engaging a large scale manufacturing partner to produce cGMP product for clinical trials;
- further developing and securing our intellectual property rights; and
- · commencing human clinical trials.

We anticipate that our operating and other expenses will increase following the closing of the Merger as we and ABS implement our business plan as a combined enterprise. After giving effect to the funds received in the recent equity and debt financings and certain committed funding over the next 12 months from the Coldstream Financing, and assuming our use of that funding at the rate we presently anticipate, as of the date of this Current Report on Form 8-K we expect to have sufficient funds to operate our business for the next 12 months. We could spend our financial resources much faster than we expect, in which case our current funds may not be sufficient to operate our business for that period.

Our estimates of the amount of cash necessary to operate our business and attain our near-term and long-term business goals may prove to be wrong, due to increased costs to achieve milestones and/or additional expenses if we encounter unanticipated difficulties or other reasons, in which case additional funding than projected would be needed. Other than the funding committed under the Coldstream Financing, we have no firm commitments for future capital. Even after giving effect to those additional committed funds, we will require significant additional financing to fund our planned operations, including further research and development relating to our primary product candidate, seeking regulatory approval of that or any other product candidate we may choose to develop, commercializing any product candidate for which we are able to obtain regulatory approval or certification, seeking to license or acquire new assets or business, and maintaining our intellectual property rights and pursuing rights to new technologies. We may not be able to obtain additional financing on commercially reasonable or acceptable terms when needed, or at all. If we cannot raise the money that we need in order to continue to develop our business, we will be forced to delay, scale back or eliminate some or all of our proposed operations. If any of these were to occur, there is a substantial risk that our business would fail and our stockholders could lose all of their investment.

Since inception we have funded our operations primarily through equity and debt financings and we expect to continue to seek to do so in the future. If we obtain additional financing by issuing equity securities, our existing stockholders' ownership will be diluted. The terms of securities we may issue in future capital-raising transactions may be more favorable for our new investors. Further, newly issued securities may include preferences, superior voting rights and the issuance of warrants or other derivative securities, which may have additional dilutive effects. If we obtain additional financing by incurring debt, we may become subject to significant limitations and restrictions on our operations pursuant to the terms of any loan or credit agreement governing the debt. Further, obtaining any loan, assuming a loan would be available when needed on acceptable terms, would increase our liabilities and future cash commitments. We may also seek funding from collaboration or licensing arrangements in the future, which may require that we relinquish potentially valuable rights to our product candidates or proprietary technologies or grant licenses on terms that are not favorable to us. Moreover, regardless of the manner in which we seek to raise capital, we may incur substantial costs in those pursuits, including investment banking fees, legal fees, accounting fees, printing and distribution expenses and other related costs.

Industry and Competition

According to a 2012 report produced by MedMarket Diligence, LLC, approximately 114 million surgical and procedure-based wounds occur annually worldwide, including 36 million from surgery in the U.S. We estimate that 20-25% of those surgeries are performed laparoscopically. Additionally, there are many minor procedures and operations that may not be included in those figures. Those surgeries and other procedures could benefit from sealants and hemostatic agents, as surgical and trauma patients are at significant risk for morbidity and mortality from bleeding and/or leaking body fluid.

Additional trends that support a demand for hemostatic and sealant products include the following:

- overall procedure volume growth;
- ambulatory same day surgery volume growth of approximately 5%;
- laparoscopic procedure volume growth; and
- efforts to reduce operating room time.

As a result of this demand, use of hemostatic agents and sealants is increasing. According to MedMarket Diligence, the market for these products achieved approximately \$3.4 billion in 2010 worldwide sales and is projected to reach \$4.5 billion in 2013 and surpass \$6.5 billion in 2017. Over two-thirds of those sales are for hemostats. The growth rate for sealants is even higher than that for hemostats due to a general lack of available products and potentially larger unmet need.

In spite of the large size of the market for these products, many available hemostatic and sealant agents possess a combination of limitations, including slow onset of action, general unreliability, user-unfriendliness, and risk for adverse effects, such as healing problems, adhesion formation, infection and other safety concerns. Many of the deficiencies of currently available hemostatic and sealant agents are the same as those of their first-generation counterparts, as revolutionary advances in underlying technologies have been elusive.

The hemostatic and sealant market is currently comprised of large companies, such as Johnson & Johnson and its affiliated companies, Covidien plc and Baxter Healthcare Corporation, as well as a number of smaller companies. Although some companies are developing new products in the hemostatic and sealant space, they appear to be mostly geared toward focused, niche applications and not on broad surgical applications. For instance, a glue-like composition may be effective for sealing an air leak in the lung or attaching two bleeding blood vessels, but it may not easily stop bleeding and enable normal healing in the liver. AC5 is envisioned as a general hemostatic agent that serves as one tool to replace narrower alternatives.

In the course of developing AC5, we engaged commercial strategy and marketing consultants to understand the routines and needs of potential customers and to assess market preferences. Although better efficacy and reliability were identified as important to those customers, it was discovered that other product features are also critical to achieving broad market acceptance. Surgeons, operating room managers, sales representatives for competitive products, and hospital administrator decision-makers identified the following characteristics as desirable features of a hemostatic agent, which we carefully considered in developing AC5 and which we believe are well satisfied by our primary product candidate:

- laparoscopic friendly;
- easily handled and applied;
- promotes a clear field of vision and does not obstruct view;
- non-viscous and flowable;
- non-sticky (to tissue or equipment);
- enables normal healing;
- indifferent to status of coagulation cascade or "blood thinning" drugs;
- non-toxic; and
- does not contain human blood product or animal components.

We hope that AC5 will meet particular market demands, and we anticipate its use in laparoscopic surgery as well as open surgery. While open surgery represents the more established market for hemostatic agents, approximately one-quarter of surgeries are laparoscopic, and that number is growing. Less invasive laparoscopic procedures produce shorter recovery times, faster discharges, less scarring, less pain and less need for pain medications. Many of the hemostasis products currently available do not possess certain features and handling characteristics required for use in a laparoscopic setting. For instance, most available products are difficult to use laparoscopically because they tend to be sticky, powdery, fabric-based or are otherwise difficult to control and/or insert into the small tubes used during laparoscopic procedures. We believe that the novel features and differentiating characteristics of AC5 will make it more suitable for laparoscopic surgeries than presently available alternatives.

Further, there seems to be increased pressure to perform more complex surgeries at reduced costs, including conducting operations in less expensive outpatient settings. Although accurate current statistics are difficult to obtain, a National Health Statistics Report from 2006 and updated in 2009 indicates that outpatient surgery volume is increasing approximately 5% annually, and a 2009 report covering U.S. surgical procedures suggests that inpatient surgery volume is declining 1% per year. A motivating factor of this trend is the increased costs associated with hospital inpatient procedures performed in operating rooms, which, according to MedMarket Diligence, have been estimated to cost between \$2,000 and \$10,000 per hour. These costs motivate increased operating room throughput and increased volume of procedures performed in outpatient settings. Both of those trends highlight the need for highly effective hemostatic and sealant products that can decrease operating room time for inpatient procedures and help to increase the safety of performing more types of procedures in less expensive outpatient settings.

Commercially available products in the hemostasis field with which we expect AC5 will compete can cost between \$50 and \$500 per procedure, with the higher value added products generally priced at the upper end of that range. Production costs of many products are significant, as they may require biotechnology or plasma separation technologies to manufacture, and they may require ingredients or other materials that are expensive to obtain. We believe that AC5 is well positioned to complete against currently available products as a result of its broad applicability in various types of surgical settings and its features that address drawbacks seen in many available hemostatic agents, as well as our planned use of a manufacturing method to produce the product that we expect will be relatively simple and cost effective compared to competing products, which could enable sales at competitive price points within the market range.

Potential Disadvantages of AC5 Compared to the Competition

Some potential disadvantages of AC5 compared to the hemostatic agents currently on the market with which we expect AC5 will compete are as follows:

- The favorable handling characteristics of AC5 are the result of its non-sticky or glue-like nature. However, if a surgeon or healthcare provider requires a product to adhere tissues together, or provide similar glue-like action, then AC5 in its current form would not achieve that desired effect.
- While we project that AC5 will be relatively economical to manufacture at scale, it will not be able to compete from a price perspective with inexpensive means to stop bleeding, such as application of pressure or use of bandages or other inexpensive hemostatic agents.
- We have not completed preclinical and clinical human trials relating to AC5, whereas marketed competition has done so. Accordingly,
 the safety and efficacy of AC5 has not been demonstrated or accepted by required regulatory agencies, and we will require significant
 resources in order to conduct the required trials and other tests to attempt to obtain such approvals.

Research and Development Expenditures

Our research and development expenses to date have primarily included costs to develop our core technology and AC5. During the year ended September 30, 2011, we incurred \$122,738 on research and development expenses, as compared to \$87,021 incurred during the year ended September 30, 2012. We expect our research and development activities and expenses to increase significantly as we execute on our business plan and pursue clinical trials.

Regulation by the FDA and Similar Foreign Agencies

Our research, development and clinical programs, as well as our manufacturing and marketing operations that may be performed by us or third party service providers on our behalf, are subject to extensive regulation in the U.S. and other countries. Most notably, we believe that AC5 will be subject to regulation as a medical device under the U.S. Food Drug and Cosmetic Act (the "FDCA") as implemented and enforced by the FDA and equivalent regulations enforced by foreign agencies in countries in which we desire to pursue commercialization. The FDA and its foreign counterparts generally govern the following activities that we do or will perform or that will be performed on our behalf, to ensure that products we may manufacture, promote and distribute domestically or export internationally are safe and effective for their intended uses:

- product design, preclinical and clinical development and manufacture;
- product premarket clearance and approval;
- product safety, testing, labeling and storage;
- record keeping procedures;
- product marketing, sales and distribution; and
- post-marketing surveillance, complaint handling, medical device reporting, reporting of deaths, serious injuries or device malfunctions and repair or recall of products.

Pre-Marketing Regulation by the U.S. FDA

Medical Device Classification

As described above, we expect that AC5 will be classified as a medical device because it does not depend on a body for metabolic or chemical activity. The FDA classifies medical devices into one of the following three classes on the basis of the amount of risk associated with the medical device and the controls deemed necessary to reasonably ensure their safety and effectiveness:

- Class I, requiring general controls, including labeling, device listing, reporting and, for some products, adherence to good
 manufacturing practices through the FDA's quality system regulations and pre-market notification;
- Class II, requiring general controls and special controls, which may include performance standards and post-market surveillance; or
- Class III, requiring general controls and approval of a premarket approval application ("PMA"), which may include post-approval conditions and post-market surveillance.

Class III devices are those that are deemed by the FDA to pose the greatest risks, such as life-sustaining, life-supporting or implantable devices, or that have a new intended use, or use advanced technology that is not substantially equivalent to that of a legally marketed device. As a result of the intended use of AC5 and the novel technology on which is it based, we anticipate that it will be classified as a Class III medical device by the FDA.

PMA Approval Process

A PMA must be submitted to the FDA if a device cannot be cleared through another approval process or is not otherwise exempt from the FDA's premarket clearance and approval requirements, and is required for most Class III medical devices. A PMA must generally be supported by extensive data, including without limitation technical, preclinical, clinical trial, manufacturing and labeling data, to demonstrate to the FDA's satisfaction the safety and efficacy of the device for its intended use. During the review period, the FDA will typically request additional information or clarification of the information previously provided. Also, an advisory panel of experts from outside the FDA may be convened to review and evaluate the PMA and provide recommendations to the FDA as to the approvability of the device, although the FDA may or may not accept the panel's recommendation. In addition, the FDA will generally conduct a pre-approval inspection of the manufacturing facility or facilities involved with producing the device to ensure compliance with the cGMP regulations. Upon approval of a PMA, the FDA may require that certain conditions of approval, such as conducting a post-market approval clinical trial, be met.

The PMA approval process can be lengthy and expensive and requires an applicant to demonstrate the safety and efficacy of the device based, in part, on data obtained from clinical trials, described below. The PMA process is estimated to take from one to three years or longer, from the time the PMA application is submitted to the FDA until an approval is obtained.

Further, if post-approval modifications are made that affect the safety or efficacy of the device, including, for example, certain types of modifications to the device's indication for use, manufacturing process, labeling or design, then new PMAs or PMA supplements would be required. PMA supplements often require submission of the same type of information as a PMA, except that the supplement is typically limited to information needed to support the changes from the device covered by the original PMA and accordingly may not require as extensive clinical and other data.

We expect that we will need to obtain PMA approval in order to sell AC5 in the U.S., but we have not submitted to the FDA any PMA covering AC5 or commenced the required clinical trials. If we are able to conduct successful preclinical studies and submit a PMA, the FDA may not grant PMA approval of AC5 for the desired indications of use, on a timely basis, or at all. Our inability to achieve regulatory approval for AC5 in the U.S., a large market for hemostatic products, would materially adversely affect our ability to grow our business.

Clinical Trials

Obtaining PMA approval requires the completion of human clinical trials that produce successful results demonstrating the safety and efficacy of the product. Clinical trials for a Class III medical device typically require an application for an investigational device exemption ("IDE"), which would be approved in advance by the FDA for a specified number of patients and study sites. Human clinical trials are subject to extensive monitoring, recordkeeping and reporting requirements, and must be conducted under the oversight of an institutional review board ("IRB") for the relevant clinical trial sites and comply with applicable FDA regulations, including those relating to good clinical practices ("GCP").

Prior to conducting a clinical trial, we also would be required to obtain the patient's informed consent in a form and substance that complies with both FDA requirements and state and federal privacy and human subject protection regulations. We, the FDA or the IRB could suspend a clinical trial at any time for various reasons, including a belief that the risks to the subjects of the trial outweigh the anticipated benefits. Even if a trial is completed, the results of clinical testing may not adequately demonstrate the safety and efficacy of the device or may otherwise not be sufficient to obtain FDA clearance or approval to market the product in the U.S.

We have not yet commenced any human clinical trials. We also have not yet commenced certain biocompatibility studies, described above under the heading "—Development and Commercialization Strategy—Regulatory—Biocompatibility Tests and Clinical Trials", that are typically completed prior to commencing clinical trials. We will require significant additional funding and preparation before we are able to initiate the first clinical trial for AC5 and in order to complete all required trials to obtain marketing approval in the U.S.

Medical Device Classification

Similar to the U.S., the EU recognizes different class of medical devices. The EU recognizes Class I, Class IIa, Class IIb or Class III medical devices. Medical devices in the EU are classified into one of those classes on the basis of the amount of potential risk to the patient associated with the medical device. Classification involves rules found in the EU's Medical Device Directive. Key questions of relevance include the degree of the device's contact with the patient, invasiveness, active nature, and indications for use. The medical device classes recognized in the EU are as follows:

- Class I, which are considered low risk devices, such as wheelchairs and stethoscopes, and require pre-market notification prior to
 placing the devices onto the EU market;
- Class IIa, which are considered low-medium risk devices and require certification by a Notified Body;
- Class IIb, which are considered medium-high risk devices and require certification by a Notified Body; and
- Class III, which are considered high-risk devices and require certification by a Notified Body.

CE Mark Approval Process

The EU has adopted numerous directives and standards regulating the design, manufacture, clinical trials, labeling, and adverse event reporting for medical devices. Each EU member state has implemented legislation applying these directives and standards at a national level. Other countries outside of the EU have also voluntarily adopted laws and regulations that mirror those of the EU with respect to medical devices.

A CE mark is a symbol placed on a product that declares the product's compliance with the essential requirements of applicable EU health, safety and environmental protection legislation. In order to receive a CE mark for a product candidate, a company must select a country in which to apply. Each country in the EU has one competent authority ("CA") that implements the national regulations by interpreting the EU directives. The CA in each country also designates and regulates Notified Bodies, which are private commercial entities designated by the national government of a member state as being competent to make independent judgments about whether a device complies with applicable regulatory requirements. An assessment by a Notified Body in the selected country within the EU is required in order to commercially distribute the device. In addition, compliance with ISO 13485 issued by the International Organization for Standardization, among other standards, establishes the presumption of conformity with the essential requirements for CE marking. Certification to the ISO 13485 standard demonstrates the presence of a quality management system that can be used by a manufacturer for design and development, production, installation and servicing of medical devices and the design, development and provision of related services.

Devices that comply with the requirements of the laws of the selected member state applying the applicable EU directive are entitled to bear a CE mark and can be distributed throughout the member states of the EU, as well as in other countries that have mutual recognition agreements with the EU or have adopted the EU's regulatory standards.

We have preliminarily selected Ireland as the country through which we will pursue a CE mark for AC5. The CA in that country has a strong record of compliance, a relatively rapid approval process, and is reputed to be trusted by the FDA. Our hope is that the selection of this country will prove helpful if and when we are able to attain a CE mark for AC5 and subsequently pursue approval with the FDA, by potentially permitting us to include data from the CE mark approval process in a PMA and/or IDE. Alternative countries have also been identified.

Clinical Trials

As with U.S. Class III medical device approval, EU Class III medical device approval requires the successful completion of human clinical trials. However, there are several key differences between the jurisdictions with respect to the approvals and processes. Obtaining a CE mark is not equivalent to obtaining FDA approval, in that a CE mark confirms the safety, but not the effectiveness, of a product. Furthermore, a CE mark affixed to a product serves as a declaration by the responsible party that the product conforms to applicable provisions and that relevant conformity assessment procedures have been completed with respect to the product. Accordingly, we anticipate that the required EU clinical trial(s) for AC5 will be smaller, faster, and less expensive than what we expect will be required for AC5 to obtain approvals in the U.S.

Post-Approval Regulation

After a medical device obtains approval from the applicable regulatory agency and is launched in the market, numerous regulatory requirements continue to apply. Many of those requirements are similar in the U.S. and in member states of the EU, and include:

- product listing and establishment registration;
- requirements that manufacturers, including third-party manufacturers, follow stringent design, testing, control, documentation and other quality assurance procedures during all aspects of the design and manufacturing process;
- labeling and other advertising regulations, including prohibitions against the promotion of products for uncleared, unapproved or offlabel use or indication:
- approval of product modifications that affect the safety or effectiveness of one of our approved devices;
- post-approval restrictions or conditions, including post-approval study commitments;
- post-market surveillance regulations, which apply, when necessary, to protect the public health or to provide additional safety and effectiveness data for the device;
- the recall authority of the applicable government agency and regulations pertaining to voluntary recalls; and
- reporting requirements, including reports of incidents in which a products may have caused or contributed to a death or serious injury
 or in which a product malfunctioned, and notices of corrections or removals.

Failure by us or by our third-party manufacturers and other suppliers to comply with applicable regulatory requirements could result in enforcement action by various regulatory authorities, which may result in monetary fines, the imposition of operating restrictions, product recalls, criminal prosecution or other sanctions.

Regulation by Other Foreign Agencies

International sales of medical devices are subject to government regulations in each country in which the device is marketed and sold, which vary substantially from country to country. The time required to obtain approval by a foreign country may be longer or shorter than that required for FDA or CE mark clearance or approval, and the requirements may substantially differ.

Other Governmental Regulations and Environmental Matters

We are or may become subject to various laws and regulations regarding laboratory practices and the use of animals in testing, as well as environmental laws and regulations governing, among other things, any use and disposal by us of hazardous or potentially hazardous substances in connection with our research. At this time, costs attributable to environmental compliance are not currently material. In each of these areas, applicable U.S. and foreign government agencies have broad regulatory and enforcement powers, including, among other things, the ability to levy fines and civil penalties, suspend or delay issuance of approvals, seize or recall products, and withdraw approvals, any one or more of which could have a material adverse effect on our business. Additionally, if we are able to successfully obtain approvals for and commercialize our product candidates, then we and our products may become subject to various federal, state and local laws targeting fraud, abuse, privacy and security in the healthcare industry.

Intellectual Property

We are focused on the development of self-assembling compositions, particularly self-assembling peptide compositions, and methods of making and using such compositions in medical and non-medical applications. Suitable applications of these compositions include limiting or preventing the movement of bodily fluids and contaminants within or on the human body, preventing adhesions, treatment of leaky or damaged tight junctions, and reinforcement of weak or damages vessels, such as aneurysms. Our strategy to date has been to develop an intellectual property portfolio in high-value jurisdictions with a track record of upholding intellectual property rights.

We have filed 10 patent applications for self-assembling peptides and methods of use thereof in 5 jurisdictions, all of which are pending. We have also entered into a license agreement with MIT pursuant to which we have been granted exclusive rights under one portfolio of patents and non-exclusive rights under another portfolio of patents. The portfolio exclusively licensed from MIT includes one issued patent in one jurisdiction that expires in 2026, and 18 pending patent applications in 10 jurisdictions. The portfolio non-exclusively licensed from MIT includes 11 issued patents in eight jurisdictions that expire between 2016 and 2026, and six pending patent applications in four jurisdictions.

Our license agreement with MIT imposes certain diligence, capital raising, and other obligations on us, including obligations to raise certain amounts of capital by specific dates. Additionally, we are responsible for all patent prosecution and maintenance fees under that agreement. Our breach of any material terms of our license agreement with MIT could permit the counterparty to terminate the agreement, which could result in our loss of some or all of our rights to use certain intellectual property that is material to our business and our lead product candidate. Our loss of any of the rights granted to us under our license agreement with MIT could materially harm our product development efforts and could cause our business to fail.

We also have been granted a non-exclusive sub-license of a patent assigned to MIT and in turn licensed by MIT to the sub-licensing third party, which patent is due to expire in 2014. The sub-license is a fully-paid and royalty-free and does not provide any outbound license grant to any ABS owned or exclusively licensed intellectual property. We presently do not anticipate any material impact on our business or operations resulting from the expected expiration of this patent in 2014.

We have pending trademark applications for $AC5^{TM}$, $Crystal\ Clear\ Surgery^{TM}$, $NanoDrape^{TM}$ and $NanoBioBarrier^{TM}$.

Employees

We presently have one full-time employee and three part-time employee, and make extensive use of third party contractors, consultants, and advisors to perform many of our present activities. We expect to increase the number of our employees significantly as we increase our operations.

Properties

We currently maintain our corporate office at One Broadway, 14th Floor, Cambridge, Massachusetts 02142 under a month-to-month property rental agreement, pursuant to which we are obligated to pay monthly rent of approximately \$2,800. We currently do not own any real property. We believe our present offices are suitable for our current and planned near-term operations.

Legal Proceedings

We are not aware of any material pending legal proceedings to which we or our subsidiary is a party or of which any of our property is the subject.

RISK FACTORS

Investment in our common stock involves a high degree of risk. The risk factors described below summarize some of the material risks inherent in and affecting our business. You should carefully consider the following risk factors before making an investment decision. If any of the following risks and uncertainties actually occurs, our business, financial condition, and results of operations could be negatively impacted and you could lose all or part of your investment.

Risks Related to our Business

Both we and ABS have incurred significant losses since inception. We expect to continue to incur losses for the foreseeable future as we pursue our operations as a combined enterprise, and we may never generate revenue or achieve or maintain profitability.

We and ABS have incurred losses in each year since our inception and we expect that losses will continue to be incurred in the foreseeable future in the operation of our new business. Our net losses and ABS's net losses were \$36,611 and \$576,911, respectively, for the year ended September 30, 2012, and \$5,037 and \$5,037 and \$573,196 for the year ended September 30, 2011. As of March 31, 2013, we and ABS had a total deficit accumulated of \$41,648 and \$3,150,911, respectively. To date, we and ABS have financed our respective operations entirely through investments by founders and other investors, and we expect to continue to do so in the foreseeable future. ABS's losses from its operations, which we are pursuing as of the closing of the Merger, have resulted principally from costs incurred in research and development programs and from general and administrative expenses, including significant costs associated with maintaining its intellectual property rights. ABS has devoted substantially all of its time, money and efforts to date to the advancement of its technology, and expects to continue to devote significant time, money and efforts to such activities going forward.

We expect to continue to incur significant expenses as we pursue ABS's business plan, and we anticipate that those expenses and losses may increase in the foreseeable future as we seek to:

- develop our principal product candidate, AC5TM;
- conduct clinical trials relating to AC5 and any other product candidate we seek to develop;
- attempt to gain regulatory approvals for any product candidate that successfully completes clinical trials;
- invest in product and process development through contract manufacturing partners;
- maintain, expand and protect our intellectual property portfolio;
- seek to commercialize selected product candidates for which we may obtain regulatory approval;
- hire additional regulatory, clinical, quality control, scientific and management consultants and personnel; and
- add operational, financial, accounting, facilities engineering and information systems consultants and personnel, consistent with expanding our operations and becoming a public company as a result of the closing of the Merger.

To become and remain profitable, we must succeed in developing and eventually commercializing product candidates with significant market potential. This will require us to be successful in a range of challenging activities, including successfully completing preclinical testing and clinical trials of product candidates, obtaining regulatory approval for our product candidates and manufacturing, marketing and selling any products for which we may obtain regulatory approval. We are only in the preliminary stages of most of those activities. We may never succeed in those activities and may never generate revenues or achieve profitability. Even if we do generate revenues sufficient to achieve profitability, we may not be able to sustain or increase profitability. Our failure to generate revenues or become and remain profitable would impair our ability to raise capital, expand our business or continue our operations, all of which would depress the price of our common stock. A decline in the prices of our common stock could cause our stockholders to lose all or a part of their investment in the Company.

There is substantial doubt about our ability to continue as a going concern.

Neither we nor ABS has generated any revenue from operations since inception, and have each incurred substantial net losses to date. Further, our operating expenses will likely increase in the foreseeable future, as we seek to increase operations in our new field as a life sciences medical device company. Moreover, our and ABS's combined cash position is vastly inadequate to support our business plans and substantial additional funding will be needed in order to pursue those plans, which include research and development of our primary product candidate, seeking regulatory approval for that product candidate, and pursuing its commercialization in the U.S., Europe and other markets. Those circumstances raise substantial doubt about our ability to continue as a going concern, and an explanatory paragraph to that effect has been included in the audited financial statements of ABS set forth in this Current Report on Form 8-K and in our audited financial statements for the year ended September 30, 2012, which are included in our Annual Report on Form 10-K for the annual period ended September 30, 2012 filed with the SEC on December 31, 2012.

We will need substantial additional funding and may be unable to raise capital when needed, which would force us to delay, reduce or eliminate our product development programs or commercialization efforts and could cause our business to fail.

We are a development stage company with no commercial products. Our primary product candidate is in the process of being developed, and will require significant additional clinical development and additional investment before it could potentially be commercialized. We anticipate that none of our product candidates will be commercially available for several years, if at all.

We currently believe that proceeds we expect to receive from current funding commitments will be sufficient to meet our anticipated cash requirements for the next 12 months. However, our plans may change and/or we may use our capital resources more rapidly than we currently anticipate. We presently expect that our expenses will increase in connection with our ongoing activities, particularly as we commence preclinical and clinical development for our lead product candidate, AC5, and that we will need to raise significant additional funds to continue operations. Our future capital requirements will depend on many factors, including:

- the scope, progress and results of our research and preclinical development activities;
- the scope, progress, results, costs, timing and outcomes of any clinical trials conducted for any of our product candidates;
- the timing of entering into, and the terms of, any collaboration agreements with third parties relating to any of our product candidates;
- the timing of and the costs involved in obtaining regulatory approvals for our product candidates;
- the costs of operating, expanding and enhancing our operations to support our clinical activities and, if our product candidates are approved, commercialization activities;
- the costs of maintaining, expanding and protecting our intellectual property portfolio, including potential litigation costs and liabilities:
- revenues, if any, received from sales of our product candidates, if any are approved by the FDA or other applicable regulatory agencies; and

 the costs of additional general and administrative personnel, including accounting and finance, legal and human resources employees. As a result of these and other factors, we expect that we will need substantial additional funding in the future. We would likely seek such funding through public or private securities offerings, incurrence of indebtedness, or some combination. We may also seek funding through collaborative arrangements if we determine them to be necessary or appropriate. Additional funding may not be available when needed on acceptable terms, or at all. If we obtain capital through collaborative arrangements, these arrangements could require us to relinquish rights to our technology or product candidates and could result in our receipt of only a portion of the revenues associated with the partnered product. If we raise capital through the sale of equity, or securities convertible into equity, it would result in dilution to our then existing stockholders, which could be significant depending on the price at which we may be able to sell our securities. If we raise additional capital through the incurrence of indebtedness, we would likely become subject to covenants restricting our business activities, and holders of debt instruments may have rights and privileges senior to those of our equity investors. In addition, servicing the interest and principal repayment obligations under debt facilities could divert funds that would otherwise be available to support research and development, clinical or commercialization activities.

If we are unable to obtain adequate financing on a timely basis or on acceptable terms in the future, we would likely be required to delay, reduce or eliminate one or more of our product development activities, which could cause our business to fail.

Our short operating history may hinder our ability to successfully meet our objectives.

We are a development stage company subject to the risks, uncertainties and difficulties frequently encountered by early-stage companies in evolving markets. Our operating subsidiary, ABS, commenced operations in 2006, and its operations to date have been primarily limited to organizing and staffing, developing and securing its technology and undertaking or funding preclinical studies of its lead product candidate. It has not demonstrated its ability to successfully complete large-scale, pivotal clinical trials, obtain regulatory approvals, manufacture a commercial scale product or arrange for a third party to do so on its behalf, or conduct sales and marketing activities necessary for successful product commercialization.

Following the completion of the Merger, we are pursuing ABS's business plan, and the management of ABS presently serves as our management. Because of our and ABS's limited operating histories, we have limited insight into trends that may emerge and affect our business, and errors may be made in developing an approach to address those trends and the other challenges faced by development stage companies. Failure to adequately do so could cause our business, results of operations and financial condition to suffer or fail. Further, our limited operating history may make it difficult for our stockholders to make any predictions about our likelihood of future success or viability.

If we are not able to attract and retain qualified management and scientific personnel, we may fail to develop our technologies and product candidates.

Our future success depends to a significant degree on the skills, experience and efforts of the principal members of our scientific and management personnel. These members include Dr. Terrence Norchi, MD, our President and CEO. The loss of Dr. Norchi or any of our other key personnel could harm our business and might significantly delay or prevent the achievement of research, development or business objectives. Further, our operation as a public company will require that we attract additional personnel to support the establishment of appropriate financial reporting and internal controls systems. Competition for personnel is intense. We may not be able to attract, retain and/or successfully integrate qualified scientific, financial and other management personnel, which could materially harm our business.

If we fails to properly manage any growth we may experience, our business could be adversely affected.

We anticipate increasing the scale of our operations as we seek to develop our product candidates, including hiring and training additional personnel and establishing appropriate systems for a company with larger operations. The management of any growth we may experience will depend, among other things, upon our ability to develop and improve our operational, financial and management controls, reporting systems and procedures. If we are unable to manage any growth effectively, our operations and financial condition could be adversely affected.

Risks Related to the Development and Commercialization of our Product Candidates

Our current business plan is dependent on the success of one product candidate.

Our business is currently focused almost entirely on the development and commercialization of one product candidate, AC5. Our reliance on one primary product candidate means that, if we are not able to obtain regulatory approvals and market acceptance of that product, our chances for success will be significantly reduced. We are also less likely to withstand competitive pressures if any of our competitors develops and obtains regulatory approval or certification for a similar product faster than we can or that is otherwise more attractive to the market than AC5. Our current dependence on one product candidate increases the risk that our business will fail if our development efforts for that product candidate experience delays or other obstacles or are otherwise not successful.

Our principal product candidate is inherently risky because it is based on novel technologies.

We are subject to the risks of failure inherent in the development of products based on new technologies. The novel nature of AC5 creates significant challenges with respect to product development and optimization, manufacturing, government regulation and approval, third-party reimbursement and market acceptance. Our failure to overcome any one of those challenges could harm our operations and overall chances for success.

If the FDA or similar foreign agencies or intermediaries impose requirements or an alternative product classification more onerous than we anticipate, our business could be adversely affected.

The development plan for our initial product candidate is based on our anticipation of pursuing the medical device regulatory pathway. However, the FDA and other applicable foreign agencies will have authority to finally determine the regulatory route for our product candidates in their jurisdictions. If the FDA or similar foreign agencies or intermediaries deem our product to be a member of a category other than a medical device, such as a drug or biologic, or impose additional requirements on our pre-clinical and clinical development, financing needs would increase, the timeline for product approval would lengthen, the program complexity and resource requirements world increase, and the probability of successfully commercializing a product would decrease. Any or all of those circumstances would adversely affect our business.

If we are not able to secure and maintain relationships with third parties that are capable of conducting clinical trials on our product candidates, our product development efforts could be adversely impacted.

Our management has limited experience in conducting preclinical development activities and clinical trials. As a result, we will need to rely on research institutions and other third party clinical investigators to conduct our preclinical and clinical trials. If we are unable to reach agreement with qualified research institutions and clinical investigators on acceptable terms, or if any resulting agreement is terminated prior to the completion of our clinical trials, then our product development efforts could be materially delayed or otherwise harmed. Further, our reliance on third parties to conduct our clinical trials will provide us with less control over the timing and cost of those trials and the ability to recruit suitable subjects to participate in the trials. Moreover, the U.S FDA and other regulatory authorities require that we to comply with standards, commonly referred to as good clinical practices, or "GCP", for conducting, recording and reporting the results of our preclinical development activities and our clinical trials, to assure that data and reported results are credible and accurate and that the rights, safety and confidentiality of trial participants are protected. Additionally, we and any third party contractor performing preclinical and clinical studies are subject to regulations governing the treatment of human and animal subjects in performing those studies. Our reliance on third parties that we do not control does not relieve us of those responsibilities and requirements. If those third parties do not successfully carry out their contractual duties, meet expected deadlines or conduct our preclinical development activities or clinical trials in accordance with regulatory requirements or stated protocols, we may not be able to obtain, or may be delayed in obtaining, regulatory approvals for our product candidates and will not be able to, or may be delayed in our efforts to, successfully commercialize our product candidates. Any of those circumstances would materially harm our business a

Any clinical trials that are conducted on our product candidates may fail.

Clinical trials are lengthy, complex and extremely expensive processes with uncertain expenditures and results and frequent failures. Any clinical trials that are commenced for one of our product candidates could be delayed, limited or fail for a number of reasons, including if:

• the FDA or other regulatory authorities do not grant permission to proceed or places a trial on clinical hold due to safety concerns or other reasons;

- sufficient suitable subjects do not enroll or remain in our trials;
- · we fail to produce necessary amounts of product candidate;
- subjects experience an unacceptable rate of efficacy of the product candidate;
- subjects experience an unacceptable rate or severity of adverse side effects, demonstrating a lack of safety of the product candidate;
- any portion of the trial or related studies produces negative or inconclusive results or other adverse events;
- reports from preclinical or clinical testing on similar technologies and products raise safety and/or efficacy concerns;
- third-party clinical investigators lose their licenses or permits necessary to perform our clinical trials, do not perform their clinical
 trials on their anticipated schedule or consistent with the clinical trial protocol, GCP or regulatory requirements, or other third
 parties do not perform data collection and analysis in a timely or accurate manner;
- inspections of clinical trial sites by the FDA or Institutional Review Boards ("IRBs") or other applicable regulatory authorities find violations that require us to undertake corrective action, suspend or terminate one or more testing sites, or prohibit us from using some or all of the data in support of our marketing applications with the FDA or other applicable agencies;
- manufacturing facilities of our third party manufacturers are ordered by the FDA or other government or regulatory authorities to temporarily or permanently shut down due to violations of cGMP or other applicable requirements;
- third-party contractors become debarred or suspended or otherwise penalized by FDA or other government or regulatory authorities for violations of regulatory requirements;
- the FDA or other regulatory authorities impose requirements on the design, structure or other features of the clinical trials for our product candidates that we and/or its third party contractors are unable to satisfy;
- one or more IRBs refuses to approve, suspends or terminates a trial at an investigational site, precludes enrollment of additional subjects, or withdraws its approval of the trial;
- the FDA or other regulatory authorities seek the advice of an advisory committee of physician and patient representatives that may view the risks of our product candidates as outweighing the benefits;
- the FDA or other regulatory authorities require us to expand the size and scope of the clinical trials, which we may not be able to do; or
- the FDA or other regulatory authorities impose prohibitive post-marketing restrictions on any of our product candidates that attains regulatory approval.

Any delay or failure of one or more of our clinical trials may occur at any stage of testing. Any such delay could cause our development costs to materially increase, and any such failure could significantly impair our business plans, which would materially harm our financial condition and operations.

We cannot market and sell any product candidate in the U.S. or in any other country or region if we fail to obtain the necessary regulatory approvals or certifications from applicable government agencies.

We cannot sell our product candidates in any country until regulatory agencies grant marketing approval or other required certifications. The process of obtaining such approval is lengthy, expensive and uncertain. If we are able to obtain such approvals for our lead product candidate or any other product candidate we may pursue, which we may never be able to do, it would likely be a process that takes many years to achieve.

To obtain marketing approvals in the U.S. for our product candidates, we must, among other requirements, complete carefully controlled and well-designed clinical trials sufficient to demonstrate to the FDA that the product candidate is safe and effective for each indication for which we seek approval. As described above, many factors could cause those trials to be delayed or to fail.

We believe that the pathway to marketing approval for our lead product candidate will likely require the FDA's approval of a PMA for the product, which likely will be classified as a Class III medical device and is based on novel technologies. This approval pathway can be lengthy and expensive, and is estimated to take from one to three years or longer from the time the PMA application is submitted to the FDA until an approval is obtained, if an approval can be obtained at all.

Similarly, to obtain approval to market our product candidates outside of the U.S., we will need to submit clinical data concerning our product candidates and receive marketing approval or other required certifications from governmental agencies in those countries, which in certain countries includes approval of the price we intend to charge for a product. In order to obtain the certification needed to market our lead product candidate in the EU, we believe that we will need to obtain a CE mark for the product, which entails scrutiny by applicable regulatory agencies and bears some similarity to the PMA process, as well as completion of at least one successful clinical trial.

We may encounter delays or rejections if changes occur in regulatory agency policies, if difficulties arise within regulatory or related agencies such as, for instance, any delays in their review time, or if reports from preclinical and clinical testing on similar technology or products raise safety and/or efficacy concerns during the period in which we develop a product candidate or during the period required for review of any application for marketing approval or certification.

Any difficulties we encounter during the approval or certification process for any of our product candidates would have a substantial adverse impact on our operations and financial condition and could cause our business to fail.

Any product for which we obtain required regulatory approvals could be subject to post-approval regulation, and we may be subject to penalties if we fail to comply with such post-approval requirements.

Any product for which we are able to obtain marketing approval or other required certifications, along with approval of the manufacturing processes, post-approval clinical data, labeling, advertising and promotional activities for such product, will be subject to continual requirements of and review by the FDA and comparable foreign regulatory authorities, including through periodic inspections. These requirements include, without limitation, submissions of safety and other post-marketing information and reports, registration requirements, cGMP requirements relating to quality control, quality assurance and corresponding maintenance of records and documents. Maintaining compliance with any such regulations that may be applicable to us in the future would require significant time, attention and expense. Even if marketing approval of a product is granted, the approval may be subject to limitations on the indicated uses for which the product may be marketed or other conditions of approval, or may contain requirements for costly and time consuming post-marketing approval testing and surveillance to monitor the safety or efficacy of the product. Discovery after approval of previously unknown problems with any approved product candidate or related manufacturing processes, or failure to comply with regulatory requirements, may result in consequences to us such as:

- restrictions on the marketing or distribution of a product, including refusals to permit the import or export of products;
- · warning letters or untitled letters;
- warning labels on the products;
- withdrawal or recall of the products from the market;
- refusal by the FDA or other regulatory agencies to approve pending applications or supplements to approved applications that we
 may submit;
- · suspension of any ongoing clinical trials;
- fines, restitution or disgorgement of profits or revenue;
- · suspension or withdrawal of marketing approvals or certifications; or
- civil or criminal penalties.

The occurrence of any such consequences if any of our product candidates achieves required regulatory marketing approvals or certifications in the future would materially adversely affect our business and operations.

Current or future legislation may make it more difficult and costly for us to obtain marketing approval or other certifications of our product candidates.

In 2007, the Food and Drug Administration Amendments Act of 2007 (the "FDAAA") was adopted. This legislation grants significant powers to the FDA, many of which are aimed at assuring the safety of medical products after approval. For example, the FDAAA grants the FDA authority to impose post-approval clinical study requirements, require safety-related changes to product labeling and require the adoption of complex risk management plans. Pursuant to the FDAAA, the FDA may require that a new product be used only by physicians with specialized training, only in specified designated health care settings, or only in conjunction with special patient testing and monitoring. The legislation also included requirements for disclosing clinical study results to the public through a clinical study registry, and renewed requirements for conducting clinical studies to generate information on the use of products in pediatric patients. Under the FDAAA, companies that violate these laws are subject to substantial civil monetary penalties. The requirements and changes imposed by the FDAAA, or any other new legislation, regulations or policies that grant the FDA or other regulatory agencies additional authority that further complicates the process for obtaining marketing approval and/or further restricts or regulates post-marketing approval activities, could make it more difficult and more costly for us to obtain and maintain approval of any of our product candidates.

Public perception of ethical and social issues may limit or discourage the type of research we conduct.

Our clinical trials will involve human subjects, and we and third parties with whom we contract also do research involving animal subjects. Governmental authorities could, for public health or other purposes, limit the use of human or animal research or prohibit the practice of our technology. Further, ethical and other concerns about our or our third party contractors' methods, particularly the use of human subjects in clinical trials or the use of animal testing, could delay our research and preclinical and clinical trials, which would adversely affect our business and financial condition.

Use of third parties to manufacture our product candidates may increase the risk that clinical development and potential commercialization of our product candidates could be delayed, prevented or impaired.

We have limited personnel with experience in medical device development and manufacturing, do not own or operate manufacturing facilities, and generally lack the resources and the capabilities to manufacture any of our product candidates on a clinical or commercial scale. We currently intend to outsource all or most of the manufacturing and packaging of our preclinical and clinical product candidates and products to third parties. However, we do not currently have agreements with any third party manufacturers for the supply of any of our product candidates. There are a limited number of manufacturers that operate under cGMP regulations and that are capable of and willing to manufacture our product candidates utilizing the manufacturing methods that are required to produce our lead product candidate, and our product candidates will compete with other product candidates for access to qualified manufacturing facilities. In the near term, if we have difficulty locating third party manufacturers to develop our product candidates for clinical work, then our product development programs will experience delays and otherwise suffer. We may also be unable to enter into agreements for the commercial supply of products with third party manufacturers in the future, or may be unable to do so when needed or on acceptable terms. Any such results could materially harm our business

Reliance on third party manufacturers entails risks to our business, including:

- the failure of the third party to maintain regulatory compliance, quality assurance, and general expertise in advanced manufacturing techniques and processes that may be necessary for the manufacture of our product candidates;
- limitations on supply availability resulting from capacity and scheduling constraints of the third parties;
- failure of the third party manufacturers to meet the demand for the product candidate, either from future customers or for clinical trial needs;
- the possible breach of the manufacturing agreement by the third party; and
- the possible termination or non-renewal of the agreement by the third party at a time that is costly or inconvenient for us.

The failure of any of our contract manufacturers to maintain high manufacturing standards could result in harm to clinical trial participants or patients using products. Such failure could also result in product liability claims, product recalls, product seizures or withdrawals, delays or failures in testing or delivery, cost overruns or other problems that could seriously harm our business or profitability. Further, our contract manufacturers will be required to adhere to FDA and other applicable regulations relating to manufacturing practices. Those regulations cover all aspects of the manufacturing, testing, quality control and recordkeeping relating to our product candidates and any products that we may commercialize in the future. The failure of our third party manufacturers to comply with applicable regulations could result in sanctions being imposed on us, including fines, injunctions, civil penalties, failure of regulatory authorities to grant marketing approval or other required certifications of our product candidates, delays, suspension or withdrawal of approvals, license revocation, seizures or recalls of product candidates or products, operating restrictions and criminal prosecutions, any of which could significantly and adversely affect our business, financial condition and operations.

Materials necessary to manufacture our product candidates may not be available on commercially reasonable terms, or at all, which may delay or otherwise hinder the development and commercialization of those product candidates.

We will rely on the manufacturers of our product candidates to purchase from third party suppliers the materials necessary to produce the compounds for preclinical and clinical studies, and may rely on those other manufacturers for commercial distribution if we obtain marketing approval or other required certifications for any of our product candidates. The materials to produce our products may not be available when needed or on commercially reasonable terms, and the prices for such materials may be susceptible to fluctuations. We do not have any control over the process or timing of the acquisition of these materials by our manufacturers. Moreover, we currently do not have any agreements for the commercial production of any of these materials. If these materials cannot be obtained for our preclinical and clinical studies, product testing and potential regulatory approval of our product candidates would be delayed, which would significantly impact our ability to develop our product candidates and materially adversely affect our ability to meet our objectives and obtain operations success.

We may not be successful in maintaining or establishing collaborations, which could adversely affect our ability to develop and, if required regulatory approvals are obtained, commercialize, our product candidates.

We are collaborating with physicians, patient advocacy groups, foundations and government agencies to assist with the development of our product candidates. If required regulatory approvals are obtained for any of our product candidates, then we may consider entering into selective collaboration arrangements with leading pharmaceutical or biotechnology companies and/or seek to establish strategic partnerships with marketing partners for the sale, marketing and distribution of our products within or outside of the U.S. If we elect to seek collaborators in the future but are unable to reach agreements with suitable collaborators, then we may fail to meet our business objectives for the affected product or program. Moreover, collaboration arrangements are complex and time consuming to negotiate, document and implement, and we may not be successful in our efforts, if any, to establish and implement collaborations or other alternative arrangements. The terms of any collaborations or other arrangements that we establish may not be favorable to us, and the success of any such collaborations will depend heavily on the efforts and activities of our collaborators. Any failure to engage successful collaborators could cause delays in our product development and/or commercialization efforts, which could harm our financial condition and operational results.

We compete with other pharmaceutical and medical device companies, including companies that may develop products that make our product candidates less attractive or obsolete.

The medical device, pharmaceutical and biotechnology industries are highly competitive. If our product candidates become available for commercial sale, we will compete in that competitive marketplace. There are several products on the market or in development that could be competitors with our lead current product candidate. While our management, which is familiar with these other products, believes that our lead product candidate could be safer and possibly more effective than those competitors, those beliefs may turn out to be wrong. Further, most of our competitors have greater resources or capabilities and greater experience in the development, approval and commercialization of medical devices or other products than we do. We may not be able to compete successfully against them. We also compete for funding with other companies in our industry that are focused on discovering and developing novel improvements in surgical bleeding prevention.

We anticipate that competition in our industry will increase. In addition, the healthcare industry is characterized by rapid technological change, resulting in new product introductions and other technological advancements. Our competitors may develop and market products that render our lead product candidate or any future product non-competitive or otherwise obsolete. Any such circumstances could cause our operations to suffer.

If we fail to generate market acceptance of our product candidates and establish programs to educate and train surgeons as to the distinctive characteristics of our product candidates, we will not be able to generate revenues on those product candidates.

Acceptance in the marketplace of our lead product candidate depends in part on our and our third party contractors' ability to establish programs for the training of surgeons in the proper usage of that product candidate, which will require significant expenditure of resources. Convincing surgeons to dedicate the time and energy necessary to properly train to use new products and techniques is challenging, and we may not be successful in those efforts. If surgeons are not properly trained, they may ineffectively use our product candidates. Such misuse could result in unsatisfactory patient outcomes, patient injury, negative publicity or lawsuits against us. Accordingly, even if our product candidates are superior to alternative treatments, our success will depend on our ability to gain and maintain market acceptance for those product candidates among certain select groups of the population that will use those products. If we fail to do so, we will not be able to generate revenue from product sales and our business, financial condition and results of operations will be adversely affected.

We face uncertainty related to pricing, reimbursement and healthcare reform, which could reduce our potential revenues.

If our product candidates are approved for commercialization, any sales will depend in part on the availability of coverage and reimbursement from third-party payors such as government insurance programs, including Medicare and Medicaid, private health insurers, health maintenance organizations and other healthcare related organizations. If our product candidates are approved for commercialization, pricing and reimbursement may be uncertain. Both the federal and state governments in the U.S. and foreign governments continue to propose and pass new legislation affecting coverage and reimbursement policies, which are designed to contain or reduce the cost of healthcare. Further, federal and state proposals and healthcare reforms could limit the prices that can be charged for the product candidates that we may develop and may further limit our commercial opportunity. Adoption of our product candidates by the medical community may be limited if doctors and hospitals do not receive adequate partial or full reimbursement for use of our products, if any are commercialized. As a result, any denial of private or government payor coverage or inadequate reimbursement for procedures performed using our products, if any are commercialized, could harm our business and reduce our prospects for generating revenue.

In addition, the U.S. Congress recently adopted legislation regarding health insurance. As a result of this new legislation, substantial changes could be made to the current system for paying for healthcare in the U.S., including modifications to the existing system of private payors and government programs, such as Medicare, Medicaid and State Children's Health Insurance Program, creation of a government-sponsored healthcare insurance source, or some combination of those, as well as other changes. Restructuring the coverage of medical care in the U.S. could impact the reimbursement for medical devices such as our product candidates. If reimbursement for our approved product candidates, if any, is substantially less than we expect, or rebate obligations associated with them are substantially increased, our business could be materially and adversely impacted.

The use of our product candidates in human subjects may expose us to product liability claims, and we may not be able to obtain adequate insurance or otherwise defend against any such claims.

We face an inherent risk of product liability claims and do not currently have product liability insurance coverage. We will need to obtain insurance coverage if and when we begin clinical trials and commercialization of any of our product candidates. We may not be able to obtain or maintain product liability insurance on acceptable terms with adequate coverage. If claims against us exceed any applicable insurance coverage we may obtain, then our business could be adversely impacted. Regardless of whether we would be ultimately successful in any product liability litigation, such litigation could consume substantial amounts of our financial and managerial resources, which could significantly harm our business.

Risks Related to our Intellectual Property

If we are unable to obtain and maintain protection for our intellectual property rights, the value of our technology and products will be adversely affected.

Our success will depend in large part on our ability to obtain and maintain protection in the U.S. and other countries for the intellectual property rights covering or incorporated into our technology and products. The patent situation in the field of medical devices generally is highly uncertain and involves complex legal, technical, scientific and factual questions. We may not be able to obtain and maintain patent protection relating to our technology or products. Even if issued, patents issued or licensed to us may be challenged, narrowed, invalidated, held to be unenforceable or circumvented, or determined not to cover our product candidates or our competitors' products, which could limit our ability to stop competitors from marketing identical or similar products. Further, we cannot be certain that we were the first to make the inventions claimed in the patents we own or license, or that protection of the inventions set forth in those patents was the first to be filed in the U.S. Third parties that have filed patents or patent applications covering similar technologies or processes may challenge our claim of sole right to use the intellectual property rights covered by the patents we own or exclusively license. Moreover, changes in applicable intellectual property laws or interpretations thereof in the U.S. and other countries may diminish the value of our intellectual property rights or narrow the scope of our patent protection. Any failure to obtain or maintain adequate protection for the intellectual property rights we use would materially harm our business, product development programs and prospects.

In addition, our proprietary information, trade secrets and know-how are important components of our intellectual property rights. We seek to protect our proprietary information, trade secrets, know-how and confidential information, in part, with confidentiality agreements with our employees, corporate partners, outside scientific collaborators, sponsored researchers, consultants and other advisors. We also have confidentiality and invention or patent assignment agreements with our employees and certain consultants and advisors. If our employees or consultants breach those agreements, we may not have adequate remedies for any of those breaches. In addition, our proprietary information, trade secrets and know-how may otherwise become known to or be independently developed by others. Enforcing a claim that a party illegally obtained and is using our proprietary information, trade secrets and know-how is difficult, expensive and time consuming, and the outcome is unpredictable. In addition, courts outside the U.S. may be less willing to protect trade secrets. Costly and time consuming litigation could be necessary to seek to enforce and determine the scope of our intellectual property rights, and failure to obtain or maintain protection thereof could adversely affect our competitive business position and results of operations.

If we lose certain intellectual property rights owned by third parties and licensed to us, our business could be materially harmed.

We have entered into certain in-license agreements with MIT and with certain other third parties, and may seek to enter into additional in-license agreements relating to other intellectual property rights in the future. To the extent we and our product candidates rely heavily on any such in-licensed intellectual property, we are subject to our and the counterparty's compliance with the terms of such agreements in order to maintain those rights. Presently, we, our lead product candidate and our business plans are dependent on the patent and other intellectual property rights that are licensed to us under our license agreement with MIT. Although that agreement has a durational term through the life of the licensed patents, it also imposes certain diligence, capital raising, and other obligations on us, our breach of which could permit the counterparty to terminate the agreement. Further, we are responsible for all patent prosecution and maintenance fees under that agreement, and a failure to pay such fees on a timely basis could also entitle the counterparty to terminate the agreement. Any failure by us to satisfy our obligations under our license agreement with MIT or any other dispute or other issue relating to that agreement could cause us to lose some or all of our rights to use certain intellectual property that is material to our business and our lead product candidate, which would materially harm our product development efforts and could cause our business to fail.

If we infringe or are alleged to infringe the intellectual property rights of third parties, our business and financial condition could suffer.

Our research, development and commercialization activities, as well as any product candidates or products resulting from those activities, may infringe or be accused of infringing a patent or other intellectual property under which we do not hold a license or other rights. Third parties may own or control those patents or other rights in the U.S. or abroad. The third parties that own or control those intellectual property rights could bring claims against us that would cause us to incur substantial time, expense, and diversion of management attention. If a patent or other intellectual property infringement suit were brought against us, we could be forced to stop or delay research, development, manufacturing or sales, if any, of the applicable product or product candidate that is the subject of the suit. In order to avoid or settle potential claims with respect to any of the patent or other intellectual property rights of third parties, we may choose or be required to seek a license from a third party and be required to pay license fees or royalties or both. Any such license may not be available on acceptable terms, or at all. Even if we or our future collaborators were able to obtain a license, the rights granted to us could be non-exclusive, which could result in our competitors gaining access to the same intellectual property rights and materially negatively affecting the commercialization potential of our planned products. Ultimately, we could be prevented from commercializing one or more product candidates, or be forced to cease some aspects of our business operations, if, as a result of actual or threatened infringement claims, we are unable to enter into licenses on acceptable terms or at all or otherwise settle such claims. Further, if any such claims were successful against us, we could be forced to pay substantial damages. Any of those results could significantly harm our business, prospects and operations.

Risks Related to the Merger and our Common Stock

The Company may have material liabilities that are not discovered until after the closing of the Merger.

The Company may have material liabilities that are not discovered until after the consummation of the Merger. We could experience losses as a result of any such undisclosed liabilities that are discovered following the Merger, which could materially harm our business and financial condition. Although the Merger Agreement contains customary representations and warranties from the Company concerning its assets, liabilities, financial condition and affairs, there may be limited or no recourse against the Company's current owners or principals in the event those prove to be untrue. As a result, the stockholders of the Company following the closing of the Merger will bear some of the risks relating to any such unknown or undisclosed liabilities.

Certain of our directors and officers own a significant percentage of our capital stock as a result of the Merger and are able to exercise significant influence over the Company.

Certain of our executive officers and directors own a significant percentage of our outstanding capital stock following the closing of the Merger. As described under the heading "The Merger and Related Transactions—Post-Merger Company Ownership" under Item 2.01 of this Current Report on Form 8-K, as of immediately following the Merger, Dr. Terrence W. Norchi, our President, Chief Executive Officer and a director, and Dr. Avtar Dhillon, the Chairman of our Board of Directors, collectively hold or control over 25% of our outstanding shares of common stock. Accordingly, these members of our Board of Directors and management team have substantial voting power to approve matters requiring stockholder approval, including without limitation the election of directors in the future, and have significant influence over our affairs. This concentration of ownership could have the effect of delaying or preventing a change in control of the Company after the Merger.

There is not now, and there may not ever be, an active market for our common stock, which trades in the over-the-counter market in low volumes and at volatile prices.

There currently is a limited market for our common stock. Although our common stock is quoted on the OTC Bulletin Board ("OTCBB"), an over-the-counter quotation system, trading of our common stock is extremely limited and sporadic and generally at very low volumes. Further, the price at which our common stock may trade is volatile and we expect that it will continue to fluctuate significantly in response to various factors, many of which are beyond our control. The stock market in general, and securities of small-cap companies driven by novel technologies in particular, has experienced extreme price and volume fluctuations in recent years. Continued market fluctuations could result in further volatility in the price at which our common stock may trade, which could cause its value to decline. To the extent we seek to raise capital in the future through the issuance of equity, those efforts could be limited or hindered by low and/or volatile market prices for our common stock.

We do not now, and are not expected to in the foreseeable future, meet the initial listing standards of the Nasdaq Stock Market or any other national securities exchange. We presently anticipate that our common stock will continue to be quoted on the OTCBB or another over-the-counter quotation system. In those venues, our stockholders may find it difficult to obtain accurate quotations as to the market value of their shares of our common stock, and may find few buyers to purchase their stock and few market makers to support its price.

A more active market for our common stock may never develop. As a result, investors must bear the economic risk of holding their shares of our common stock for an indefinite period of time.

Our common stock is a "penny stock."

The SEC has adopted regulations that generally define "penny stock" as an equity security that has a market price of less than \$5.00 per share, subject to specific exemptions. The market price of our common stock is, and is expected to continue to be in the near term, less than \$5.00 per share and is therefore a "penny stock." Brokers and dealers effecting transactions in "penny stock" must disclose certain information concerning the transaction, obtain a written agreement from the purchaser and determine that the purchaser is reasonably suitable to purchase the securities. Those rules may restrict the ability of brokers or dealers to sell our common stock and may affect the ability of our stockholders to sell their shares of our common stock. In addition, if our common stock continues to be quoted on the OTCBB as we expect, then our stockholders may find it difficult to obtain accurate quotations for our stock, and may find few buyers to purchase our stock and few market makers to support its price.

FINRA sales practice requirements may limit a stockholder's ability to buy and sell our stock.

In addition to the "penny stock" rules described above, FINRA has adopted rules that require that, in recommending an investment to a customer, a broker-dealer must have reasonable grounds for believing that the investment is suitable for that customer. Prior to recommending speculative low priced securities to their non-institutional customers, broker-dealers must make reasonable efforts to obtain information about the customer's financial status, tax status, investment objectives and other information. Under interpretations of these rules, FINRA has indicated its belief that there is a high probability that speculative low priced securities will not be suitable for at least some customers. These FINRA requirements make it more difficult for broker-dealers to recommend that at least some of their customers buy our common stock, which may limit the ability of our stockholders to buy and sell our common stock and could have an adverse effect on the market for our shares.

There may be additional risks because the business of ABS is going public by means of a reverse merger transaction.

Additional risks may exist because the business of ABS is becoming a public company through a "reverse merger" transaction. Securities analysts of major brokerage firms may not provide coverage of the Company following the Merger because there may be little incentive to brokerage firms to recommend the purchase of our common stock. There may also be increased scrutiny by the SEC and other government agencies and holders of our securities prior to the Merger due to the nature of the transaction, as there has been increased focus on transactions such as the Merger in recent years. Further, since the Company existed as a "shell company" under applicable rules of the SEC up until the closing of the Merger on June 26, 2013, there will be certain restrictions and limitations on the Company going forward relating to any potential future issuances of additional securities to raise funding and compliance with applicable SEC rules and regulations.

The elimination of monetary liability against our directors and officers under Nevada law and the existence of indemnification rights held by our directors, officers and employees may result in substantial expenditures by the Company and may discourage lawsuits against our directors, officers and employees.

Our Articles of Incorporation eliminates the personal liability of our directors and officers to our Company and our stockholders for damages for breach of fiduciary duty as a director or officer to the extent permissible under Nevada law. Further, our amended and restated bylaws provide that we are obligated to indemnify any of our directors or officers to the fullest extent authorized by the Nevada law and, subject to certain conditions, advance the expenses incurred by any director or officer or director in defending any action, suit or proceeding prior to its final disposition. Those indemnification obligations could result in the Company incurring substantial expenditures to cover the cost of settlement or damage awards against our directors or officers, which we may be unable to recoup. These provisions and resultant costs may also discourage us from bringing a lawsuit against any of our current or former directors or officers for breaches of their fiduciary duties, and may similarly discourage the filing of derivative litigation by our stockholders against our directors and officers even though such actions, if successful, might otherwise benefit us or our stockholders.

We are subject to the reporting requirements of federal securities laws, compliance with which involves significant time, expense and expertise.

We are a public reporting company in the U.S., and, accordingly, are subject to the information and reporting requirements of the Exchange Act and other federal securities laws, including the obligations imposed by the Sarbanes-Oxley Act. The costs associated with preparing and filing annual, quarterly and current reports, proxy statements and other information with the SEC in the ordinary course, as well as preparing and filing audited financial statements, will cause our operational expenses to be higher than ABS's operational expenses would have been if it remained privately held and did not effect the Merger.

Our present management team, which consists of ABS's former management team, has never operated a publicly-traded company. It will be time consuming, difficult and costly for our management team to acquire expertise and experience in operating a public company, and to develop and implement the internal controls and reporting procedures required by Sarbanes-Oxley and other applicable securities laws. As described in our reports filed with the SEC, including Item 9A of our Annual Report on Form 10-K for the fiscal year ended September 30, 2012 and Part I, Item 4 of our Form 10-Q for the quarterly period ended March 31, 2013, we have identified material weaknesses in our internal controls and procedures relating to insufficient resources and personnel, the lack of a separate standing audit committee, our management team's lack of formal training in this area, and insufficient segregation of duties on our internal team. As a result, we have concluded that our disclosure controls and procedures were not effective as of the end of the period covered by those reports. We will need to hire additional financial reporting, internal controls and other finance staff in order to develop and implement appropriate internal controls and reporting procedures as required by applicable securities regulations for public companies, which we may not be able to do on a timely basis or at all. Additionally, any of our efforts to improve our internal controls and design, implement and maintain an adequate system of internal controls will require that we expend significant resources. Moreover, even if we are able to hire and retain such additional personnel and are able implement other measures aimed at remediating the current and any future additional material weaknesses identified in our internal controls and procedures, we may nonetheless fail to remediate all weaknesses and fail to establish and/or maintain adequate internal controls and procedures.

Shares of our common stock that have not been registered under federal securities laws, regardless of whether such shares are restricted or unrestricted, are subject to resale restrictions imposed by Rule 144, including those set forth in Rule 144(i) which apply to a "shell company." In addition, any shares of our common stock that are held by affiliates, including any received in a registered offering, will be subject to the resale restrictions of Rule 144(i).

Pursuant to Rule 144 ("Rule 144") of the Securities Act of 1933, as amended (the "Securities Act"), a "shell company" is defined as a company that has no or nominal operations and either no or nominal assets; assets consisting solely of cash and cash equivalents; or assets consisting of any amount of cash and cash equivalents and nominal other assets. As such, we may be deemed a "shell company" pursuant to Rule 144 prior to the closing of the Merger, and as such, sales of our securities pursuant to Rule 144 are not permitted until a period of at least 12 months has elapsed from the date on which this Current Report on Form 8-K, reflecting our status as a non-"shell company", is filed with the SEC. Therefore, any restricted securities we sell in the future or issue to consultants or employees in consideration for services rendered or for any other purpose will have no liquidity until and unless such securities are registered under the Securities Act and/or until a year after the date of the filing of this Current Report on Form 8-K, provided that we and the selling stockholder are in compliance with the other requirements of Rule 144. As a result, it will be more difficult for us to raise funding to support our operations through the sale of debt or equity securities unless we agree to register such securities under the Securities Act, which could cause us to expend additional time and cash resources. Further, it may be more difficult for us to compensate our employees and consultants with our securities instead of cash. Our previous status as a "shell company" could also limit our use of our securities to pay for any acquisitions we may seek to pursue in the future (although none are currently planned), which could cause the value of our securities, if any, to decline in value or become worthless. In addition, any shares held by affiliates, including shares received in any registered offering, will be subject to the resale restrictions of Rule 144(i).

We do not intend to pay cash dividends on our capital stock in the foreseeable future.

We have never declared or paid any dividends on our shares and do not anticipate paying any such dividends in the foreseeable future. Any future payment of cash dividends would depend on our financial condition, contractual restrictions, solvency tests imposed by applicable corporate laws, results of operations, anticipated cash requirements and other factors and will be at the discretion of the our Board of Directors. Our stockholders should not expect that we will ever pay cash or other dividends on our outstanding capital stock.

We are at risk of securities class action litigation that could result in substantial costs and divert management's attention and resources.

In the past, securities class action litigation has been brought against companies following periods of volatility of its securities in the market place, particularly following a company's initial public offering. Due to the volatility of our stock price, we could be the target of securities litigation in the future. Securities litigation could result in substantial costs and divert management's attention and resources.

MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion should be read in conjunction with ABS's historical financial statements and the pro forma financial statements filed with this Current Report on Form 8-K. The following discussion and analysis contains forward-looking statements that involve risks and uncertainties, as described under the heading "Forward-Looking Statements" in this Current Report on Form 8-K. Actual results could differ materially from those projected in the forward-looking statements. For additional information regarding these risks and uncertainties, please see the disclosure under the heading "Risk Factors" elsewhere in this Current Report on Form 8-K.

Overview

Arch Therapeutics, Inc.

Arch Therapeutics, Inc. (the "Company") was incorporated under the laws of State of Nevada on September 16, 2009 as Almah, Inc. On May 10, 2013, the Company entered into an Agreement and Plan of Merger (the "Merger Agreement") with Arch Biosurgery, Inc. (formerly known as Arch Therapeutics, Inc.), a Massachusetts corporation ("ABS"), and Arch Acquisition Corporation, the Company's wholly owned subsidiary formed for the purpose of the transaction, pursuant to which Arch Acquisition Corporation merged with and into ABS and ABS thereby became the wholly owned subsidiary of the Company (the "Merger"). Upon the closing of the Merger, the Company has abandoned its prior business plan and is now pursuing the business and plan of operations of ABS. In contemplation of the Merger, effective May 24, 2013, the Company increased its authorized common stock from 75,000,000 shares to 300,000,000 shares and effected a forward stock split, by way of a stock dividend, of its issued and outstanding shares of common stock at a ratio of 11 shares to each one (1) issued and outstanding share, and effective June 5, 2013, the Company changed its name from Almah, Inc. to Arch Therapeutics, Inc. and changed the ticker symbol under which its common stock trades on the OTC Bulletin Board from "AACH" to "ARTH". The Merger closed on June 26, 2013.

For a discussion and analysis of the Company's financial condition and results of operations prior to the Merger, please refer to the information set forth under the heading "Management's Discussion and Analysis of Financial Condition and Results of Operations," as well as the related financial statements, in the Company's Annual Report on Form 10-K for the fiscal year ended September 30, 2012 filed with the Securities and Exchange Commission ("SEC") on December 31, 2012 and in the Company's Quarterly Report on Form 10-Q for the quarter ended March 31, 2013 filed with the SEC on May 20, 2013, which information and financial statements are incorporated herein by reference.

Arch Biosurgery, Inc. (referred to in this discussion and analysis set forth below as "ABS", "we", "us", or "our") was incorporated under the laws of Commonwealth of Massachusetts on March 6, 2006 as Clear Nano Solutions, Inc. On April 7, 2008, we changed our name to Arch Therapeutics, Inc., and on August 28, 2009, we increased our authorized common stock, no par value, from 275,000 shares to 1,275,000 shares. Effective upon the closing of the Merger, we changed our name from Arch Therapeutics, Inc. to Arch Biosurgery, Inc.

All information relating to ABS set forth in this Current Report on Form 8-K, including the information in this discussion and analysis and the related financial statements filed herewith, is presented as of the years ended September 30 in all respects.

This discussion and analysis is a discussion of the financial condition and results of operations of ABS solely, and not of financial condition and results of operations of the Company.

ABS is a life science medical device company in the development stage with limited operations to date. We aim to develop products that make surgery and interventional care faster and safer by utilizing a novel approach to stop bleeding (referenced as "hemostasis"), control leaking (referenced as "sealant"), and provide other advantages during surgery and trauma care. Our core technology is based on a self-assembling peptide solution that creates a physical, mechanical barrier, which could be applied to bleeding organs or wounds to seal leaking blood and other fluids. We believe our technology could support an innovative platform of potential products in the field of stasis and barrier applications. Our first product candidate, AC5TM, is designed to achieve hemostasis in minimally invasive and open surgical procedures, and we hope to develop other hemostatic or sealant product candidates in the future based on our self-assembling peptide technology platform. Our plan and business model is to develop products that apply that core technology to human bodily fluids and connective tissues.

Our primary product candidate, AC5, relies on this technology to achieve hemostasis during surgical procedures. AC5 is a biocompatible synthetic peptide comprising naturally occurring amino acids. When applied to a wound, AC5 intercalates into the interstices of the connective tissue where it self-assembles into a physical, mechanical nanoscale structure that provides a barrier to leaking substances, such as blood. The results of early data from preclinical animal tests have shown that AC5 achieves hemostasis quickly and effectively. AC5 can be directly applied as a liquid or sprayed, making it user-friendly and able to conform to irregular wound geometry, and is not sticky or glue-like, making it ideal for use in the setting of minimally invasive laparoscopic surgeries. Further, AC5 is transparent, which should make it easier for a surgeon or other healthcare provider to maintain a clear field of vision during a surgical procedure and prophylactically stop bleeding as it starts, which we call Crystal Clear SurgeryTM.

We have devoted much of our operations to date to the development of our core technology, including selecting our lead product composition, conducting initial safety and other related tests, generating scale-up, reproducibility and manufacturing methods, and developing and protecting the intellectual property rights underlying our technology platform.

Our long-term business plan includes the following goals:

- conducting successful clinical trials on AC5;
- obtaining regulatory approval or certification of AC5 in the European Union (the "EU"), the U.S., and other jurisdictions;
- expanding our intellectual property portfolio;
- developing appropriate third party relationships to manufacture, distribute, market and otherwise commercialize AC5; and
- develop additional product candidates in the hemostatic and sealant field.

In furtherance of our long-term business goals, we expect to focus on the following activities during the remainder of calendar year 2013 and calendar year 2014:

- · conducting formal biocompatibility studies;
- participating in EU and, subsequently, U.S. regulatory meetings;
- preparing for initial clinical trials, including developing clinical trial protocols;
- engaging a large scale manufacturing partner to produce cGMP product for clinical trials;
- further developing and securing our intellectual property rights; and
- commencing human clinical trials.

Results of Operations

The following discussion of our results of operations should be read together with the financial statements included in this Current Report on Form 8-K. The period to period comparisons of our annual and interim results of operations that follow are not necessarily indicative of future results of ABS or the Company.

Six Months Ended March 31, 2013 and 2012

	March 31, 2013	March 31, 2012	Increase (Decrease)
Revenue	<u>\$</u> -	\$ -	\$ -
Operating Expenses			
General and Administrative	278,411	231,097	47,314
Research and Development	11,290	17,912	(6,622)
Loss from Operations	(289,701) (249,009)	40,692
Other Income (Expense)	(88,193) (75,006)	13,187
Net Income (Loss)	\$ (377,893) \$ (324,015)	\$ 53,879

Revenue

We did not generate any revenue in either of the six months ended March 31, 2013 or 2012.

General and Administrative Expense

We incurred general and administrative expense during the six months ended March 31, 2013 in the amount of \$278,410, compared to general and administrative expense incurred during the six months ended March 31, 2012 in the amount of \$231,097 (an increase of \$47,313). Our general and administrative expenses during those periods primarily included legal fees, patent prosecution costs, payroll related expenses and office overhead. The increase in general and administrative expense period over period is primarily attributable to increased costs associated with patent prosecution and maintenance.

General and administrative expenses are generally expected to increase following the closing of the Merger as a result of plans to ramp up operations and requirements to comply with public company reporting obligations. We expect increased expenses related to plans to hire additional personnel and consultants and expected incurrence of additional legal fees.

Research and Development Expense

We incurred research and development expense during the six months ended March 31, 2013 in the amount of \$11,290, compared to research and development expense incurred during the six months ended March 31, 2012 in the amount of \$17,912 (a decrease of \$6,622). Our research and development expenses primarily relate to our activities to develop our primary product candidate, and are comprised mostly of payroll related expenses. The decrease in research and development expense between periods is primarily attributable to reduction in payroll related expenses.

Research and development expenses are expected to increase following the closing of the Merger as a result of plans to pursue additional preclinical and clinical studies and otherwise relating to development of ABS's primary product candidate.

Other Income (Expense)

We incurred total other expenses during the six months ended March 31, 2013 in the amount of \$88,193, compared to total other expenses incurred during the six months ended March 31, 2012 in the amount of \$75,006 (an increase of \$13,187). Other expenses during those periods were primarily interest accrued on debt. The increase in other expense between periods is attributable to additional interest associated with increased amounts of outstanding debt.

	September 30, 2012		September 30, 2011	Increase (Decrease)	
Revenue	\$	-	\$ -	\$ -	
Operating Expenses					
General and Administrative		333,503	362,096	(28,593)	
Research and Development		87,021	122,738	(35,717)	
Loss from Operations		(420,524)	(484,834)	(64,310)	
Other Income (Expense)		(156,387)	(88,362)	68,025	
Net Income (Loss)	\$	(576,911)	\$ (573,196)	\$ 3,715	

Revenue

We did not generate any revenue in either of the years ended September 30, 2012 or 2011.

General and Administrative Expense

We incurred general and administrative expense during the year ended September 30, 2012 in the amount of \$333,503, compared to general and administrative expense incurred during the year ended September 30, 2011 in the amount of \$362,096 (a decrease of \$28,593). Our general and administrative expenses during those periods primarily included legal fees, patent prosecution costs, payroll related expenses, license maintenance fees, professional fees and office overhead. The decrease in general and administrative expense between periods is primarily attributable to reduction in payroll related expenses.

General and administrative expenses are generally expected to increase following the closing of the Merger as a result of plans to ramp up operations and requirements to comply with public company reporting obligations. We expect increased expenses related to plans to hire additional personnel and consultants and expected incurrence of additional legal fees.

Research and Development Expense

We incurred research and development expense during the year ended September 30, 2012 in the amount of \$87,021, compared to research and development expense incurred during the year ended September 30, 2011 in the amount of \$122,738 (a decrease of \$35,717). Our research and development expenses primarily relate to our activities to develop our primary product candidate, and are comprised of payroll related expenses, advisor fees and cost of materials. The decrease in research and development expense between periods is primarily attributable to reduction in payroll related expenses.

Research and development expenses are expected to increase following the closing of the Merger as a result of plans to pursue additional preclinical studies and clinical studies and otherwise relating to development of ABS's primary product candidate.

Other Income (Expense)

We incurred total other expenses during the year ended September 30, 2012 in the amount of \$156,387, compared to total other expenses incurred during the year ended September 30, 2011 in the amount of \$88,362 (an increase of \$68,025). Other expenses during those periods were primarily interest accrued on debt. The increase in other expense between periods is attributable to additional interest associated with increased amounts of outstanding debt.

Liquidity and Capital Resources

Working Capital

Our working capital as of March 31, 2013 and September 30, 2012 is summarized as follows:

	March 31, 2013	September 30, 2012	
Total Current Assets	\$ 2,002	\$ 20,447	
Total Current Liabilities	2,712,392	2,552,439	
Working Capital	<u>\$ (2,710,390</u>)	\$ (2,531,992)	

As of March 31, 2013, total current assets were \$2,002, compared to total current assets of \$20,447 as of September 30, 2012 (a decrease of \$18,445). The decrease was due to a decrease in cash balances and amortization of prepaid expenses. Our total current assets as of March 31, 2013 were comprised primarily of cash and prepaid expenses.

As of March 31, 2013, total current liabilities were \$2,712,392, compared to total current liabilities of \$2,552,439 as of September 30, 2012 (an increase of \$159,953). The increase was primarily due to an increase in the current maturities of outstanding debt, current portion of accrued interest on debt and accounts payable and a decrease in accrued expenses. Our total current liabilities as of March 31, 2013 were comprised primarily of current maturities of debt, current portion of interest accrued on debt, accounts payable and accrued expenses.

As a result, on March 31, 2013, we had negative working capital of \$2,710,390.

Cash Flow

Our cash on-hand as of March 31, 2013 was \$881, compared to cash on-hand as of September 30, 2012 of \$17,139 (a decrease of \$16,258). The decrease was primarily due to operating expenditures that exceeded funds provided by financing activities.

Cash Used in Operating Activities

Cash used in operating activities during the six months ended March 31, 2013 was \$266,257, compared to cash used in operating activities during the six months ended March 31, 2012 of \$85,565 (an increase of \$180,692). The increase was primarily due to an increase in general and administrative expense attributable to increased costs associated with patent prosecution and a reduction in the balance of accounts payable period over period.

Cash used in operating activities during the year ended September 30, 2012 was \$254,636, compared to cash used in operating activities during the year ended September 30, 2011 of \$375,065 (a decrease of \$120,429). The decrease was primarily due to a decrease in operating expenses resulting from a reduction in payroll related expenses and an increase in the balance of accounts payable year over year.

Cash Used in Investing Activities

There was no cash used in investing activities during the six months ended March 31, 2013 or during the six months ended March 31, 2012.

There was no cash used in investing activities during the year ended September 30, 2012 or during the year ended September 30, 2011.

Cash Provided by Financing Activities

Cash provided by financing activities during the six months ended March 31, 2013 was \$250,000, compared to cash provided by financing activities during the six months ended March 31, 2012 of \$60,000 (an increase of \$190,000). Cash provided by financing activities during the year ended September 30, 2012 was \$235,000, compared to cash provided by financing activities during the year ended September 30, 2011 of \$401,200 (a decrease of \$166,200).

All cash provided by financing activities during the periods presented was obtained from (i) loans to us made by the President and Chief Executive Officer of ABS between February 2009 and February 2011 for an aggregate amount of \$275,200 (the "CEO Loans") and our related issuance of a promissory note therefor, and (ii) in connection with issuances to various investors in ABS of convertible promissory notes bearing interest at rates ranging from 6% to 10% (the "Convertible Notes") and related warrants. Those Convertible Notes and related warrants were originally issued to investors on various dates during and before the periods presented in bridge loan transactions in expectation of potential financings of our capital stock. In contemplation of the Merger, any such potential financing of the capital stock of ABS was abandoned and such securities were amended and restated to provide for (i) the conversion of all amounts owed under all outstanding Convertible Notes into the right to receive an aggregate of 9,000,000 shares of the Company's common stock upon the closing of the Merger, calculating to approximately one share of the Company's common stock for each \$0.27 outstanding under the Convertible Notes, and (ii) the cancellation of the warrants in full upon the closing of the Merger.

Sources of Capital

Prior to the closing of the Merger, we have primarily funded our operations through the financing activities described under the heading "— Cash Flow—Cash Provided By Financing Activities" above. Other than such financing activities and the Coldstream Financing effected in contemplation of the Merger and described below, we have had no sources of material funding to date. Since inception through March 31, 2013, we have received an aggregate of \$275,200 from the CEO Loans and an aggregate of \$1,985,000 from our issuance of the Convertible Notes and related warrants.

In contemplation of the Merger, the Company obtained a financing commitment totaling \$2,000,000 to fund our combined operations. Of that amount, gross proceeds of \$1,250,000 have been received to date and the remaining \$750,000 is to be provided over the 12 months following the closing of the Merger. The financing commitment is set forth in a financing agreement (the "Financing Agreement") dated April 19, 2013 between the Company and Coldstream Summit Ltd. ("Coldstream"), pursuant to which the Company agreed to issue and sell, and Coldstream agreed to purchase or assist in securing the purchase of, a total of \$2,000,000 worth of units in a private offering (the "Coldstream Financing"). Each unit issued in the Coldstream Financing is to be sold at a price of \$0.50 per share and is to consist of (i) one share of the Company's common stock and (ii) one warrant to purchase one share of the Company's common stock at an exercise price of \$0.75 per share and with a term of 12 months. The Company advanced to us prior to the closing of the Merger the aggregate gross proceeds of \$1,250,000 that have been received in the Coldstream Financing to date, which are being used to fund our present operations.

Since inception we have funded our operations primarily through equity and debt financings and we expect to continue to seek to do so in the future. If additional financing is obtained by issuing equity securities of the Company, its existing stockholders' ownership will be diluted. Further, the terms of securities that may be issued in future capital-raising transactions may be more favorable for new investors, and newly issued securities may include preferences, superior voting rights and the issuance of warrants or other derivative securities, which may have additional dilutive effects. If additional financing is obtained by incurring debt, our operations may become subject to significant limitations and restrictions pursuant to the terms of any loan or credit agreement governing the debt. Further, obtaining any loan, assuming a loan would be available when needed on acceptable terms, would increase the liabilities and future cash commitments of us and the Company as a combined enterprise. Funding for our operations may also be sought from collaboration or licensing arrangements, which could require that we and the Company relinquish potentially valuable rights to our product candidates or proprietary technologies or grant licenses on terms that are not favorable. Moreover, regardless of the manner in which we and the Company seek to raise capital, substantial costs may be incurred in those pursuits, including investment banking fees, legal fees, accounting fees, printing and distribution expenses and other related costs.

Cash Requirements

As described above, we anticipate that our operating and other expenses will increase following the closing of the Merger as we and the Company implement our business plan as a combined enterprise and a public reporting company. After giving effect to the funds received in the recent equity and debt financings and certain committed funding over the next 12 months from the Coldstream Financing, as of the date of this Current Report on Form 8-K we estimate we will have sufficient funds to operate the business for the next 12 months. However, these estimates could differ if we encounter unanticipated difficulties, in which case our current funds may not be sufficient to operate our business for that period. In addition, our estimates of the amount of cash necessary to operate our business may prove to be wrong, and we could spend our available financial resources much faster than we currently expect.

Other than the funding committed under the Coldstream Financing, neither we nor the Company have any firm commitments for future capital. Even after giving effect to those additional committed funds, significant additional financing will be required to fund our planned operations in the near term and in future periods, including research and development activities relating to our principal product candidate, seeking regulatory approval of that or any other product candidate we may choose to develop, commercializing any product candidate for which we are able to obtain regulatory approval or certification, seeking to license or acquire new assets or business, and maintaining our intellectual property rights and pursuing rights to new technologies. We do not presently have, nor do we expect in the near future to have, revenue to fund our business from our operations, and will need to obtain all of our necessary funding from external sources in the near term. We may not be able to obtain additional financing on commercially reasonable or acceptable terms when needed, or at all. If we cannot raise the money that we need in order to continue to develop our business, we will be forced to delay, scale back or eliminate some or all of our proposed operations. If any of these were to occur, there is a substantial risk that our business would fail and our stockholder could lose all of their investment.

Going Concern

We have not received revenues from sales of products or services, and have recurring losses from operations. As of March 31, 2013, we had incurred a net loss of \$3,150,911 since our inception. In their report on the annual financial statements for the year ended September 30, 2012, our independent auditors included an explanatory paragraph regarding concerns about our ability to continue as a going concern. The financial statements included in this Current Report on Form 8-K contain note disclosures describing the circumstances that resulted in the inclusion of that explanatory paragraph. The continuation of our business as a going concern is dependent upon raising additional capital and eventually attaining and maintaining profitable operations. The financial statements included in this Current Report on Form 8-K do not include any adjustments relating to the recoverability of assets that might be necessary should operations discontinue.

Contractual Obligations

The table below outlines payments due under our significant contractual obligations over the periods shown, exclusive of interest. The table below includes our obligations as of March 31, 2013 and does not reflect any changes in our obligations that have occurred after that date. Reference Note 10 of our audited financial statements for the year ended September 30, 2012 and Note 4 of our unaudited interim financial statements for the period ended March 31, 2013 included in this Current Report on Form 8-K.

	Payments Due By Period					
	Less Than One Year	One to Three Years	T	hree to Five Years	 Iore Than live Years	Total
Contractual Obligations at March 31, 2013:						
Long Term Obligations (1)	\$25,000	\$ 80,000	\$	100,000	\$ (2)	\$ 205,000

- (1) Represents certain license maintenance fees and patent prosecution costs we are obligated to pay to the Massachusetts Institute of Technology ("MIT") under the terms of our license agreement with MIT.
- (2) Annual license maintenance obligations extend through the life of the patents subject to the license. For each year that the agreement is in effect after 2017, the annual license maintenance fee commitment would be \$50,000. In addition, MIT is entitled to royalties on applicable future product sales, if any. The annual license maintenance payments may be applied towards royalties payable to MIT for that year for product sales.

Critical Accounting Policies and Significant Judgments and Estimates

Pursuant to certain disclosure guidance issued by the SEC, the SEC defines "critical accounting policies" as those that require the application of management's most difficult, subjective or complex judgments, often as a result of the need to make estimates about the effect of matters that are inherently uncertain and may change in subsequent periods.

Our critical accounting policies that we anticipate will require the application of our most difficult, subjective or complex judgments are as follows:

Basis of Presentation — Development Stage Company

We have not earned any revenue from operations. Accordingly, our activities have been accounted for as those of a "Development Stage Company" as set forth in Financial Accounting Standards Board ("FASB") ASC 915. Among the disclosures required by ASC 915 are that our financial statements be identified as those of a development stage company, and that the statements of operations, stockholders' deficit and cash flows disclose activity since the date of our inception.

Income Taxes

In accordance with FASB ASC 740, Income Taxes, we recognize deferred tax assets and liabilities for the expected future tax consequences or events that have been included in our financial statements and/or tax returns. Deferred tax assets and liabilities are based upon the differences between the financial statement carrying amounts and the tax bases of existing assets and liabilities and for loss and credit carryforwards using enacted tax rates expected to be in effect in the years in which the differences are expected to reverse. Deferred tax assets are reduced by a valuation allowance if it is more likely than not that some portion or all of the deferred tax asset will not be realized.

We provide reserves for potential payments of tax to various tax authorities related to uncertain tax positions when management determines that it is probable that a loss will be incurred related to these matters and the amount of the loss is reasonably determinable. We have no reserves related to uncertain tax positions as of March 31, 2013 and September 30, 2012.

Off-Balance Sheet Arrangements

We have no significant off-balance sheet arrangements that have or are reasonably likely to have a current or future effect on our financial condition, changes in financial condition, revenues or expenses, results of operations, liquidity, capital expenditures or capital resources that is material to stockholders.

SECURITY OWNERSHIP OF CERTAIN BENEFICIAL OWNERS AND MANAGEMENT

The following tables sets forth certain information regarding the beneficial ownership of our common stock by (i) each person who, to our knowledge, owns more than 5% of our common stock, (ii) each of our directors and named executive officers, and (iii) all of our current executive officers and directors as a group. Unless otherwise indicated in the footnotes to the following table, the address of each person named in the table is: c/o Arch Therapeutics, Inc., One Broadway, 14th Floor, Cambridge, Massachusetts 02412. The information set forth in the first table below is based on 44,000,000 shares of our common stock issued and outstanding on June 26, 2013 immediately prior to giving effect to the closing of the Merger, and the second table below is based on 58,645,212 shares of our common stock issued and outstanding on June 26, 2013 as of immediately after giving effect to the closing of the Merger. Shares of our common stock subject to options, warrants, or other rights currently exercisable or exercisable within 60 days of June 26, 2013, are deemed to be beneficially owned and outstanding for computing the share ownership and percentage of the person holding such options, warrants or other rights, but are not deemed outstanding for computing the percentage of any other person.

Beneficial Ownership Immediately Prior to Giving Effect to the Merger:

	Number of Shares	Percentage Beneficially	
	Beneficially	Owned	
Name of Beneficial Owner	Owned	(1)	
5%+ Stockholders			
Fitzroy Limited	2,500,000	5.68%	
Twelve Pins Partners, LLC (2)	10,000,000	22.7%	
Walk on Water Ventures, LLC	2,750,000	6.3%	
Directors and Named Executive Officers:			
Avtar Dhillon	7,000,000	15.9%	
Terrence W. Norchi (2)	10,000,000	22.7%	
Joey Power (3)	0	*	
Current Directors and Executive Officers as a Group (2 persons)	17,000,000	38.6%	

^{*} Less than 1%

- (1) Except as otherwise indicated, we believe that the beneficial owners of the common stock listed above, based on information furnished by such owners, have sole investment and voting power with respect to such shares, subject to community property laws where applicable. Beneficial ownership is determined in accordance with the rules of the SEC and generally includes voting or investment power with respect to securities.
- (2) Dr. Norchi is the sole member of Twelve Pins Partners, LLC and has sole voting and investment control with respect to the shares it holds. Dr. Norchi disclaims beneficial ownership of these securities except to the extent of his pecuniary interest therein.

(3) Mr. Power was our sole named executive officer and director during our fiscal year ended September 30, 2012 and during the subsequent period until April 23, 2013. His employment with us terminated and he resigned as a director in April 2013 in anticipation of the consummation of the Merger.

Beneficial Ownership Immediately After Giving Effect to the Merger:

	Number of	Percentage Beneficially	
	Shares Beneficially		
Name of Beneficial Owner	Owned	(1)	
5%+ Stockholders			
Twelve Pins Partners, LLC (2)	10,000,000	17.1%	
Directors and Named Executive Officers:			
Avtar Dhillon	7,160,373	12.2%	
Terrence W. Norchi (3)	11,419,076	19.5%	
Arthur Rosenthal	58,400	*	
Joey Power (4)	0	*	
Current Directors and Executive Officers as a Group (3 persons)	18,637,849	31.8%	

- * Less than 1%
- (1) Except as otherwise indicated, we believe that the beneficial owners of the common stock listed above, based on information furnished by such owners, have sole investment and voting power with respect to such shares, subject to community property laws where applicable. Beneficial ownership is determined in accordance with the rules of the SEC and generally includes voting or investment power with respect to securities.
- (2) Dr. Norchi is the sole member of Twelve Pins Partners, LLC and has sole voting and investment control with respect to the shares it holds. Dr. Norchi disclaims beneficial ownership of these securities except to the extent of his pecuniary interest therein.
- (3) Represents (a) 10,000,000 shares of our common stock held by Twelve Pins Partners, LLC, with respect to which Dr. Norchi holds sole voting and investment control, and (b) 1,419,076 shares issued to Dr. Norchi upon the closing of the Merger in exchange for the cancellation of the shares of common stock and convertible notes of ABS owned by him immediately prior to the closing of the Merger. As stated in footnote (2) above, Dr. Norchi disclaims beneficial ownership of the securities held by Twelve Pins Partners, LLC except to the extent of his pecuniary interest therein.
- (4) Mr. Power was our sole named executive officer and director during our fiscal year ended September 30, 2012 and during the subsequent period until April 23, 2013. His employment with us terminated and he resigned as a director in April 2013 in anticipation of the consummation of the Merger.

Changes in Control

We are unaware of any arrangement the operation of which may at a subsequent date result in a change in control of the Company.

DIRECTORS AND EXECUTIVE OFFICERS

The following individuals serve as our current directors and executive officers:

Name	Position	Age	Director/Officer Since
Dr. Avtar Dhillon	Chairman of the Board of Directors	52	April 2013
Dr. Arthur Rosenthal	Director	66	June 2013
Dr. Terrence W. Norchi	President, Chief Executive Officer and Director	48	April 2013
Alan T. Barber	Chief Financial Officer	59	June 2013

Business Experience

The following is a brief account of the education and business experience of our current directors and executive officers:

Dr. Avtar Dhillon. Dr. Dhillon has served as the Chairman of our Board of Directors since April 2013 and has been on the Board of Directors of ABS since May 2011. Previously, Dr Dhillon was the President and Chief Executive Officer of Inovio Pharmaceuticals, Inc. (formerly Inovio Biomedical Corporation) (NYSE Euronext: INO) from October 2001 to June 2009, as President and Chairman of Inovio from June 2009 until October 2009, as Executive Chairman until August 2011, and as Chairman from September 2011. During his tenure at Inovio, Dr. Dhillon led the successful turnaround of the company through a restructuring, acquisition of technology from several European and North American companies, and a merger with VGX Pharmaceuticals to develop a vertically integrated DNA vaccine development company with one of the strongest development pipelines in the industry. Dr. Dhillon led multiple successful financings for Inovio and concluded several licensing deals that included global giants, Merck and Wyeth (now Pfizer). Prior to joining Inovio, Dr. Dhillon was vice president of MDS Capital Corp. (now Lumira Capital Corp.), one of North America's leading healthcare venture capital organizations. In July 1989, Dr. Dhillon started a medical clinic and subsequently practiced family medicine for over 12 years. Dr. Dhillon has been instrumental in successfully turning around struggling companies and influential as an active member in the biotech community. From March 1997 to July 1998, Dr.

Dhillon was a consultant to Cardiome Pharma Corp. (NASDAQ: CRME), where he lead a turnaround based on three pivotal financings, establishing a clinical development strategy, and procuring a new management team. In his role as a founder and board member of companies, Dr. Dhillon has been involved in several early stage healthcare focused companies listed on U.S. or Canadian stock exchanges, which have successfully matured through advances in their development pipeline and subsequent M&A transactions. Most recently, he was a founding board member (May 2003) of Protox Therapeutics, Inc. (TSX-V: SHS) (now Sophiris Bio Inc.), a publicly traded specialty pharmaceutical company. Dr. Dhillon maintained his board position until the execution of a financing of up to \$35 million with Warburg Pincus in November 2010. Dr. Dhillon currently sits on the Board of Directors of BC Advantage Funds, a Venture Capital Corporation in British Columbia, and since March 2012 has been the Chairman of the Board of Directors of Stevia First Corp. (OTCQB: STVF), an agricultural biotechnology

company engaged in the cultivation and harvest of stevia leaf and the development of stevia products. Since March 2011, Dr. Dhillon has also served as the Chairman of the Board of Directors of of OncoSec Medical, Inc. (OTCQB: ONCS), a company developing its advanced-stage ImmunoPulse DNA-based immunotherapy to treat solid tumor and metastatic cancers. Dr. Dhillon adds value to our Board of Directors with his extensive experience as a member of boards of directors and senior management of other public companies and with his experience in company building, financing, and licensing with large industry partners.

Dr. Arthur Rosenthal. Dr. Rosenthal has been appointed as a director of the Company upon the consummation of the Merger, and has served as the Chairman of the Board of ABS since April 2011. He has served for 40 years in senior research and product development executive roles for medical technology companies and in those roles has successfully directed commercialization efforts for hundreds of novel medical products. He was Chief Scientific Officer at Boston Scientific from January, 1994 to January, 2005, Vice President of Research and Development at Johnson and Johnson Medical Products, Inc. from April, 1990 to January, 1994 and more recently Chief Executive Officer of two start-up companies, Labcoat, Ltd. and Cappella, Inc., both developing cardiovascular medical devices. He is currently, and has been since January 2010, a Professor of Practice in Translational Research in Boston University's College of Engineering, where he oversees biomedical engineering innovation. Dr. Rosenthal received his Ph.D. in biochemistry from the University of Massachusetts, Amherst, 1973. Currently, Dr. Rosenthal serves as Non-Executive Director and Chairman and as a member of the Compensation Committee and Audit Committee for Cyberonics, Inc. (NASDAQ: CYBX), having joined its Board of Directors in January 2007. Dr. Rosenthal is a valuable member of our Board of Directors because of his high-ranking roles in private and public medical device companies, his extensive experience overseeing research and development and commercialization of a large number of products in the medical field, and his company-building acumen.

Dr. Terrence W. Norchi. Dr. Terrence W. Norchi commenced service as our President, Chief Executive Officer and Interim Chief Financial Officer and a director on our Board of Directors on April 23, 2013. As a result of the appointment of Alan T. Barber as the Company's Chief Financial Officer concurrently with the closing of the Merger, Dr. Norchi no longer serves as the Company's Interim Chief Financial Officer. Dr. Norchi also serves as the President and Chief Executive Officer and a director of ABS, and has served in those positions since cofounding ABS in 2006. Prior to founding ABS, Dr. Norchi was a portfolio manager and pharmaceutical analyst at Putnam Investments from April 2002 to September 2004. Prior to that he served as the senior global biotech and international pharmaceutical equity analyst at Citigroup Asset Management from January 2000 to March 2002, and as a sell-side analyst covering non-U.S. pharmaceutical equities at Sanford C. Bernstein in New York City from September 1996 to December 1999. Dr. Norchi earned an M.B.A. from the Massachusetts Institute of Technology, Sloan School of Management in 1996. Dr. Norchi earned an M.D. degree in 1990 from Northeast Ohio Medical University and completed his internal medicine residency in 1994 at Baystate Medical Center, Tufts University School of Medicine, where he was selected to serve as Chief Medical Resident. Dr. Norchi brings to our Board of Directors invaluable experience and knowledge of our core technology and proposed product candidates as a result of his first-hand experience with the development of that technology, having ushered it from the research laboratory to its current stage of development, and also contributes his investing experience as a former public company analyst and a portfolio manager.

Alan T. Barber. Mr. Barber has been appointed as the Chief Financial Officer of the Company effective as of the consummation of the Merger in June 2013, and has served as the Chief Financial Officer of ABS since August 2008. He has over 30 years of financial management experience and has been since September 2005, and continues to be, an independent consultant on financial matters. Prior to that Mr. Barber was the Chief Financial Officer for a number of technology and life science start-up companies including Biotrove, Inc. from April 2004 to September 2005, Omnisonics Medical Technologies, Inc. from October 2001 to April 2004, Innovation Chain, Inc. from October 2000 to September 2001, MyWay.com from December 1999 to October 2000, Medical Foods, Inc. from November 1997 to October 1999 and Ergo Science, Inc. from October 1993 to November 1997. Prior to that Mr. Barber was a Partner with the international accounting firm of PricewaterhouseCoopers (formerly Coopers & Lybrand) from July 1979 to October 1993 where he was elected as a Partner in the firm in July 1986. Prior to that he worked for the international accounting firm KPMG from May 1975 to July 1979. Mr. Barber received a Bachelor of Science degree in Accounting from the Florida State University, Rovetta School of Business, and is a Certified Public Accountant.

Term of Office of Directors

Our directors are elected at each annual meeting of stockholders and serve until the next annual meeting of stockholders or until their successor has been duly elected and qualified, or until their earlier death, resignation or removal.

Family Relationships

No family relationships exist between any of our current or former directors or executive officers.

Involvement in Certain Legal Proceedings

No director, executive officer, significant employee or control person of the Company has been involved in any legal proceeding listed in Item 401(f) of Regulation S-K in the past 10 years.

Meetings of Directors; Committees of the Board

Our Board of Directors held no formal meetings during the fiscal year ended September 30, 2012. All proceedings of the Board of Directors were conducted by resolutions consented to in writing by the directors and filed with the minutes of the proceedings of the directors. Such resolutions consented to in writing by the directors entitled to vote on such resolutions at a meeting of the directors are, according to the Nevada Revised Statutes and the bylaws of the Company, as valid and effective as if they had been approved at a meeting of the directors duly called and held. We do not presently have a policy regarding director attendance at meetings of directors or meetings of our stockholders.

We do not currently have separate standing audit, nominating or compensation committees, or committees performing similar functions. Due to the present and prior size of our Board of Directors, our Board of Directors believes that it is not necessary to have separate standing audit, nominating or compensation committees at this time because the functions of each such committee are adequately performed by our full Board of Directors. However, it is anticipated that our Board of Directors will form separate standing audit, nominating and compensation committees, with the audit committee including an audit committee financial expert, when our Board of Directors determines that the establishment of such committees is advisable if and when we seek additional qualified and value-adding directors to serve on our Board of Directors and as we further develop our business and operations. We do not presently have an audit, nominating or compensation committee charter as we have not established any such committees.

Audit Committee

Our Board of Directors has not established a separate standing audit committee within the meaning of Section 3(a)(58)(A) of the Exchange Act. Instead, the entire Board of Directors presently acts as the audit committee within the meaning of Section 3(a)(58)(B) of the Exchange Act and will continue to do so upon the appointment of any new directors and until such time as a separate audit committee has been established.

Section 16(a) Beneficial Ownership Reporting Compliance

Section 16(a) of the Exchange Act requires our directors, executive officers, and stockholders holding more than 10% of our outstanding common stock to file with the SEC initial reports of ownership and reports of changes in ownership of our common stock. Executive officers, directors, and persons who own more than 10% of our common stock are required by SEC regulations to furnish us with copies of all Section 16(a) reports they file. We were not subject to the reporting requirements of Section 16(a) of the Exchange Act prior to the closing of the Merger.

Nominations to the Board of Directors

Director candidates are considered based upon various criteria, including without limitation their broad-based business and professional skills and experiences, their knowledge of the industry in which we operate and ability to add perspectives relating to that industry, concern for the long-term interests of our stockholders, diversity, and personal integrity and judgment. In addition, directors must have time available to devote to the activities of our Board of Directors and to understand and enhance their knowledge of our industry and business plans. Our Board of Directors has a critical role in guiding our strategic direction and overseeing the management of the Company, and accordingly we seek to attract and retain highly qualified directors who have sufficient time to attend to their substantial duties and responsibilities as a director of the Company.

In carrying out its responsibilities, our Board of Directors will consider director candidates suggested by stockholders. If a stockholder wishes to formally place a director candidate's name in nomination, then he or she must do so in accordance with the provisions of our amended and restated bylaws, which provide certain advance notice and other requirements in order for our stockholders to nominate a director candidate. Proposed nominations of director candidates must be timely sent to the Secretary of the Company, c/o Arch Therapeutics, Inc., One Broadway, 14th Floor, Cambridge, Massachusetts 02142. To be timely, notice of a proposed director nominee must be delivered to or mailed and received at the Company's address set forth above not less than 90 days prior to the date of the meeting at which the proposed director nominee would be up for election. Further, the stockholder's notice relating to a director nomination must set forth the following information about each person whom the stockholder proposes to nominate for election or re-election as a director: (i) the name, age, business address and residence address of the person, (ii) the principal occupation or employment of the person, (iii) the class and number of shares of our common stock that are beneficially owned by the person, and (iv) any other information relating to the person that is required to be disclosed in solicitations for proxies for election of directors pursuant to Regulation 14A under the Exchange Act. Further, the stockholder must also provide the following information about itself and certain of persons associated with, controlling, controlled by or acting on concert with the stockholder: (i) the name and record address of the stockholder, (ii) the class and number of shares of our common stock which are beneficially owned by the stockholder; and (iii) certain information specified in our amended and restated bylaws regarding any hedge transactions entered into, derivative instruments beneficially owned by, or rights to dividends on the shares of our common stock beneficially owned by such persons. Pursuant to the terms of our amended and restated bylaw, the Company may also require any proposed nominee to furnish such other information as may reasonably be required by the Corporation to determine the eligibility of such proposed nominee to serve as a director on our Board of Directors.

Board Leadership Structure and Role in Risk Oversight

Terrence W. Norchi, M.D. currently serves as our principal executive officer and a director, however, an independent director, Dr. Avtar Dhillon, serves as the Chairman of the Board of Directors. The Board of Directors has determined that leadership structure to be in the best interests of the Company and its stockholders. That leadership structure enables Dr. Norchi to focus on carrying out the day to day direction and long term strategic goals of the Company, and also provide valuable input regarding the functions and operations of our business to the Board of Directors. It also enables the Chairman of the Board to focus specifically on the activities of the Board of Directors, and better enables the Board of Directors to provide effective guidance to and oversight and accountability of management. The Board of Directors will continue to evaluate our leadership structure and modify it as appropriate based on the size, resources and operations of the Company.

Subsequent to the closing of the Merger, it is anticipated that our Board of Directors will establish procedures to determine an appropriate role for the Board of Directors in our risk oversight function.

Compensation Committee Interlocks and Insider Participation

The Company has no compensation committee, and during its last completed fiscal year, its former sole director and officer participated in deliberations of our Board of Directors regarding officer compensation. During the last completed fiscal year, no executive officer of our Company (i) served as a member of the compensation committee (or other committee performing equivalent functions or, in the absence of any such committee, the entire board of directors) of another entity, one of whose executive officers served on our Company's Board of Directors, (ii) served as a director of another entity, one of whose executive officers served on our Company's Board of Directors, or (iii) served as a member of the compensation committee (or other committee performing equivalent functions or, in the absence of any such committee, the entire board of directors) of another entity, one of whose executive officers served as a director of our Company.

EXECUTIVE COMPENSATION

Executive Compensation Relating to the Company

Compensation Information Before the Closing of the Merger

There has been no compensation awarded, earned or paid by the Company to its former President and Chief Executive Officer and sole director, Mr. Joey Power, during the term of his service in such positions (although he was reimbursed for any out-of-pocket expenses that he incurred on the Company's behalf in connection with such service). Mr. Power performed such service as a consultant and not an employee, and did not have any employment agreement with the Company during the term of his service in such positions. Reference is made to the information set forth under the heading "Executive Compensation" in our Annual Report on Form 10-K for the fiscal year ended September 30, 2012 filed with the SEC on December 31, 2012, which is incorporated herein by reference.

Compensation Information After the Closing of the Merger

In connection with the Dr. Terrence Norchi's position as our President and Chief Executive Officer, on June 25, 2013 we entered into an executive employment agreement with Dr. Norchi with an effective date of June 26, 2013. In addition, in connection with the appointment of Alan Barber as our Chief Financial Officer, on June 26, 2013 we entered into an employment agreement with Mr. Barber. Reference is made to the description of that agreement set forth in Item 5.02 of this Current Report on Form 8-K, which are incorporated herein by reference.

We presently have no formal plan for compensating our non-employee directors for their service as our directors, and none of our current or former directors has received compensation for their service as of the end of our last-completed fiscal year.

Executive Compensation Relating to ABS

ABS became our wholly owned subsidiary as a result of the closing of the Merger on June 26, 2013. The following table summarizes all compensation earned in each of ABS's years ended September 30, 2012 and 2011 by (i) its principal executive officer, and (ii) its next most highly compensated executive officer other than its principal executive officer serving as an executive officer as of September 30, 2012 and whose total compensation exceeded \$100,000 in during the year ended September 30, 2012 (of which there were none).

Summary Compensation Table

	Year ended		All Other				
Name and Principal Position	September 30,	Salary		Compensation		Total	
Dr. Terrence W. Norchi,	2012	\$	200,000	_	\$	200,000	
President, Chief Executive Officer (1)	2011	\$	125,000	_	\$	125,000	

(1) Dr. Norchi has been the President and Chief Executive Officer of ABS since its inception in 2006, and was appointed as our President, Chief Executive Officer and Interim Chief Financial Officer on April 23, 2013. All amounts reflected in this table were paid to Dr. Norchi in connection with his service as an officer of ABS during the periods presented.

Employment Agreements

In connection with our entry into an executive employment agreement with Dr. Norchi on June 26, 2013, Dr. Norchi's former employment agreement with ABS was terminated effective as of the same date, pursuant to a termination agreement and release between Dr. Norchi and ABS. That termination agreement and release is attached as Exhibit 10.7 to this Current Report on Form 8-K and is incorporated herein by reference.

Director Compensation

The directors of ABS during the year ended September 30, 2012 did not receive any compensation for their service as directors of ABS during such period.

Potential Payments Upon Termination or Change-in-Control

SEC regulations state that we must disclose information regarding agreements, plans or arrangements that provide for payments or benefits to our named executive officers in connection with any termination of employment or change in control of the Company.

Prior to the closing of the Merger, there were no agreements, plans or arrangements between the Company and its sole director and executive officer named above that would have provided for any such payments or benefits to such named executive officer upon a termination of employment of change of control of the Company. Further, prior to the closing of the Merger, there were no agreements, plans or arrangements between ABS and its named executive officer set forth above that would have provided for any such payments or benefits to such named executive officer upon a termination of employment of change of control of ABS.

In connection with the closing of the Merger, we have entered into an executive employment agreement with each of Dr. Terrence W. Norchi and Alan T. Barber. Of those agreements, only our agreement with Dr. Norchi provides for payments or benefits in connection with any termination of employment or change in control of the Company.

Pursuant to the terms of Dr. Norchi's employment agreement with us, if, as of the last day of our last-completed fiscal year, (i) Dr. Norchi had been an executive officer of the Company, (ii) his employment agreement had been in effect, and (iii) he had been terminated For Cause or had terminated his employment for Good Reason (as such terms are defined in Dr. Norchi's employment agreement), then Dr. Norchi would have been entitled to receive salary continuation totaling \$275,000 over a 12-month period, plus the payment of Dr. Norchi's premiums to continue his group health coverage under COBRA until the earlier of (a) the end of the 12 months following the date of such termination, or (b) the date Dr. Norchi were to become covered under another employer's health plan.

Reference is made to the description of the terms of Dr. Norchi's employment agreement with the Company set forth in Item 5.02(e) of this Current Report on Form 8-K and the full text of that employment agreement, which is attached hereto as Exhibit 10.8 and is incorporated by reference herein.

CERTAIN RELATIONSHIPS AND RELATED TRANSACTIONS, AND DIRECTOR INDEPENDENCE

Certain Relationships and Transactions

Except for Dr. Terrence Norchi, our President, Chief Executive Officer, former Interim Chief Financial Officer and a director, and Dr. Dhillon, the Chairman of our Board of Directors, who each became executive officers and/or directors of our Company shortly following the Company's and ABS's entry into a binding letter of intent regarding the terms of the Merger (the "LOI"), none of the current directors and executive officers were directors or executive officers of the Company prior to the closing of the Merger, nor did any hold any position with the Company prior to the closing of the Merger, nor have any been involved in any material proceeding adverse to the Company or any transactions with the Company or any of its directors, executive officers, affiliates or associates that are required to be disclosed pursuant to the rules and regulations of the SEC.

Review, Approval or Ratification of Transactions with Related Persons

We have not adopted a code of ethics applicable to our principal executive officer, principal financial officer, principal accounting officer or controller, or persons performing similar functions that establishes, among other things, procedures for handling actual or apparent conflicts of interest. Due to the small size of our Company, at this time we have determined to rely on our full Board of Directors to review related party transactions and identify and prevent conflicts of interest. Our Board of Directors reviews a transaction in light of the affiliations of the director, officer, employee or stockholder and the affiliations of such person's immediate family. Transactions are presented to our Board of Directors for approval before they are entered into or, if that is not possible, for ratification after the transaction has occurred. If our Board of Directors finds that a conflict of interest exists, then it will determine the appropriate remedial action, if any. Our Board of Directors approves or ratifies a transaction if it determines that the transaction is consistent with the best interests of the Company and its stockholders. The procedures described above have been approved by resolutions adopted by our Board of Directors.

Related Party Transactions

Dr. Terrence Norchi and Dr. Avtar Dhillon were appointed to their officer and director positions with us on April 23, 2013, shortly following the entry into the LOI between the Company and ABS relating to the Merger. Each of Dr. Avtar Dhillon and Dr. Terrence Norchi also held, and continue to hold, positions with ABS, with Dr. Norchi serving as the President, Chief Executive Officer and a director of ABS and Dr. Dhillon serving as a director of ABS. As a result, each of Dr. Norchi and Dr. Dhillon were directors and/or officers of us and of ABS upon the signing of the Merger Agreement on May 10, 2013. Further, it was a condition to the closing of the Merger that Dr. Norchi and Dr. Dhillon each receive, on or before the date of the closing, 10,000,000 shares of our common stock in private transfers from the former holders thereof. As a result of those transfers and other shares of our common stock to which Dr. Norchi and Dr. Dhillon became entitled in exchange for their former shares and notes of ABS, as of the closing of the Merger Dr. Norchi and Dr. Dhillon collectively hold or otherwise control approximately 25.8% of our common shares on a fully diluted basis and approximately 30.5% of our outstanding common shares. The number of shares of our common stock received by Dr. Norchi and Dr. Dhillon in connection with the Merger was negotiated by the parties to the LOI and was determined without input from any independent third party.

As a result of his ownership of 23,260 shares of ABS immediately prior to the closing of the Merger, Arthur Rosenthal became entitled to receive an aggregate of 58,400 shares of the Company's common stock upon the closing of the Merger.

On June 19, 2013, Dr. Terrence Norchi purchased from ABS an aggregate amount of \$15,397 of certain convertible promissory note and warrant positions (the "Repurchased Securities"). The Repurchased Securities had originally been issued by ABS to third parties in June 2009, were repurchased by ABS from the original holders on April 30, 2013, and were resold to Dr. Norchi and other third party purchasers effective June 19, 2013. The Repurchased Securities were first issued by ABS to the original holders thereof in a bridge loan transaction in expectation of potential financings of ABS's capital stock. In contemplation of the Merger, any such potential financing of the capital stock of ABS was abandoned and such Repurchased Securities were amended and restated to provide for (i) the conversion of all amounts owed under the convertible promissory notes into the right to receive an aggregate of 1,349,614 shares of the Company's common stock upon the closing of the Merger, calculating to approximately one share of the Company's common stock for each \$0.27 outstanding under the notes, and (ii) the cancellation of the warrants in full upon the closing of the Merger. Accordingly, Dr. Norchi became entitled to receive 56,103 shares of the Company's common stock upon the closing of the Merger as a result of his purchase of \$15,397 worth of the Repurchased Securities.

Pursuant to the terms of Dr. Norchi's former employment agreement with ABS, Dr. Norchi was entitled to receive a cash bonus in the amount of \$500,000 and certain warrants to acquire ABS's capital stock upon the closing of a capital raise by ABS of at least \$1,000,000. Dr. Norchi agreed to defer his right to receive such cash bonus and warrants at the time they became due and issuable upon ABS's satisfaction of that capital raise condition. In connection with the closing of the Merger on June 26, 2013 and the concurrent entry into an executive employment agreement with the Company, Dr. Norchi and ABS entered into a termination agreement and release pursuant to which Dr. Norchi's employment agreement with ABS has been terminated by mutual agreement and Dr. Norchi effective as of the closing of the Merger and Dr. Norchi has agreed to waive in full any and all right to receive such cash bonus and warrants. The termination agreement and release is filed as Exhibit 10.7 to this Current Report on Form 8-K, and the description set forth above is qualified in its entirety by the full text of that agreement, which is incorporated herein by reference.

Commencing in February 2009, Dr. Norchi loaned ABS an aggregate amount of \$275,200 in several installments. On January 21, 2010, ABS issued a promissory note to Dr. Norchi in exchange for that loan in principal amount of \$275,200, which promissory note, as amended, bears interest at the rate of 6% per annum through December 31, 2009 and at the rate of 10% per annum thereafter, is due upon demand and is unsecured. On June 24, 2013, ABS paid to Dr. Norchi all amounts due and owing under such promissory note, which totaled \$373,488 as of such date.

On July 11, 2011, we issued a total of 4,000,000 shares of common stock to our then-President, Chief Executive Officer and sole director Mr. Joey Power at a price of \$0.005 per share for an aggregate amount of \$20,000 in a private offering.

Director Independence

Our Board of Directors has determined that Dr. Avtar Dhillon and Dr. Arthur Rosenthal would qualify as "independent" as that term is defined by Nasdaq Listing Rule 5605(a)(2). Further, although we do not presently have established separately designated audit, nominating or compensation board committees, Dr. Dhillon and Dr. Rosenthal would qualify as "independent" under Nasdaq Listing Rules applicable to such board committees. Dr. Terrence W. Norchi would not qualify as "independent" under Nasdaq Listing Rules applicable to the Board of Directors generally or to separately designated board committees because he currently serves as our President and Chief Executive Officer. Mr. Joey Power, our sole director during our last completed fiscal year, also did not qualify as "independent" under such Nasdaq Listing Rules because he served as our President, Chief Executive Officer and Chief Financial Officer during that period.

Subject to some exceptions, Nasdaq Listing Rule 5605(a)(2) provides that an independent director is a person other than an executive officer or other employee of the Company or any other individual having a relationship which, in the option of our Board of Directors, would interfere with the exercise of independent judgment in carrying out the responsibilities of a director, provided that a director will not be independent if (a) the director is, or in the past three years has been, an employee of ours; (b) a member of the director's immediate family is, or in the past three years has been, an executive officer of ours; (c) the director or a member of the director's immediate family has received more than \$120,000 per year in direct compensation from us within the preceding three years, other than for service as a director or benefits under a tax-qualified retirement plan or non-discretionary compensation (or, for a family member, as a non-executive employee); (d) the director or a member of the director's immediate family is a current partner of our independent public accounting firm, or has worked for such firm in any capacity on our audit at any time during the past three years; (e) the director or a member of the director's immediate family is, or in the past three years has been, employed as an executive officer of a company where one of our executive officers serves on the compensation committee; or (f) the director or a member of the director's immediate family is an executive officer, partner or controlling shareholder of a company that makes payments to, or receives payments from, us in an amount which, in any twelve-month period during our past three fiscal years, exceeds the greater of 5% of the recipient's consolidated gross revenues for that year or \$200,000 (except for payments arising solely from investments in our securities or payments under non-discretionary charitable contribution matching programs).

MARKET PRICE OF AND DIVIDENDS ON OUR COMMON EQUITY AND RELATED STOCKHOLDER MATTERS

Market Information

Our common stock is quoted on the OTCBB over-the-counter quotation system. Our common stock began quotation on the OTCBB on June 26, 2012 under the trading symbol "AAHC.OB". Effective June 5, 2013, in connection with the change of our name to Arch Therapeutics, Inc., our trading symbol changed to "ARTH.OB". There was no trading of our common stock on the OTCBB or any other over-the-counter market prior to January 2, 2013. Although our common stock is quoted on the OTCBB, there is a limited trading market for our common stock and there have been few trades in our common stock to date. Because our common stock is thinly traded, any reported sale prices may not be a true market-based valuation of our common stock.

The table below sets forth the high and low closing bid quotations for our common stock for the fiscal quarters indicated as reported on the OTCBB. The quotations reflect inter-dealer prices, without retail mark-up, mark-down or commission and may not represent actual transactions.

	High	Low
Fiscal Year Ended September 30, 2011		
Quarter ended December 31, 2010*	_	_
Quarter ended March 31, 2011*	_	_
Quarter ended June 30, 2011*	_	_
Quarter ended September 30, 2011*	_	_
Fiscal Year Ended September 30, 2012		
Quarter ended December 31, 2011*	_	_
Quarter ended March 31, 2012*	_	_
Quarter ended June 30, 2012*	_	_
Quarter ended September 30, 2012*	_	_
Fiscal Year Ending September 30, 2013		
Quarter ended December 31, 2012*	_	_
Quarter ended March 31, 2013	\$ 1.01	\$ 1.01
Quarter ending June 30, 2013 (as of June 25, 2013)	\$ 6.00	\$ 0.54

Transfer Agent

The transfer agent and registrar for our common stock is Empire Stock Transfer, 1859 Whitney Mesa Drive, Henderson, Nevada 89014.

Holders of Common Stock

As of June 26, 2013 and after giving effect to the closing of the Merger, there were 20 holders of record of our common stock.

Dividends

We have never declared or paid any cash dividends or distributions on our capital stock. We currently intend to retain our future earnings, if any, to support operations and to finance expansion and we do not anticipate paying any cash dividends on our common stock in the foreseeable future.

Securities Authorized for Issuance under Equity Compensation Plans

We had not adopted any equity compensation plan as of the end of our last completed fiscal year on September 30, 2012.

On June 18, 2013, our Board of Directors approved and adopted the Arch Therapeutics, Inc. 2013 Stock Incentive Plan (the "Plan"), and authorized management to submit the Plan to our stockholders for approval. On June 18, 2013, a majority of our stockholders executed a written consent approving and adopting the Plan. Pursuant to the approval of our Board of Directors and our stockholders, the adoption of the Plan became effective on June 18, 2013.

The Plan permits us to grant a variety of forms of awards, including stock options, stock appreciation rights, restricted stock, restricted stock units, and dividend equivalent rights, to allow us to adapt our incentive compensation program to meet our needs. The number of shares of our common stock initially reserved for issuance under the Plan to employees, directors and/or consultants in such awards is 7,825,388 shares. The number of shares of our common stock reserved for issuance under the Plan for all awards except for incentive stock option awards will be subject to increase on an annual basis, on the first business day of our fiscal year commencing in 2013, by an amount equal to the lesser of (i) 3,000,000 shares, ii) four percent of the number of shares outstanding on the last day of our immediately preceding fiscal year, or (iii) such lesser number of shares as determined by the administrator of the Plan. Our Board of Directors currently serves as the administrator of the Plan. As of the date of this Current Report on Form 8-K, non-qualified stock options to purchase an aggregate of 3,825,388 shares of our common stock have been granted under the Plan.

RECENT SALES OF UNREGISTERED SECURITIES

By the Company

On July 11, 2011, we issued a total of 4,000,000 shares of common stock to our then-President, Chief Executive Officer and sole director Mr. Joey Power at a price of \$0.005 per share for an aggregate amount of \$20,000. The issuance of those shares has not been registered under the Securities Act, and such shares have been issued in reliance upon an exemption from registration under Section 4(2) of the Securities Act and Regulation S promulgated thereunder. Such shares may not be offered or sold in the United States absent registration under or exemption from the Securities Act and any applicable state securities laws. In determining that the issuance of such shares qualified for an exemption under Section 4(2) of the Securities Act and Regulation S promulgated thereunder, we relied on the following facts: the recipient of the shares represented that he is not a "U.S. Person" as defined in Rule 902 promulgated under the Securities Act; we used no advertising or general solicitation in connection with the issuance of such shares; and the shares were issued as restricted securities.

On June 18, 2013, pursuant to the approval of our Board of Directors, we issued an aggregate of 1,500,000 shares of our common stock pursuant to restricted stock awards granted outside of the Plan to two consultants performing services for the Company. The issuance of those shares has not been registered under the Securities Act, and such shares have been issued in reliance upon an exemption from registration under Section 4(2) of the Securities Act. Such shares may not be offered or sold in the United States absent registration under or exemption from the Securities Act and any applicable state securities laws. In determining that the issuance of such shares qualified for an exemption under Section 4(2) of the Securities Act, we relied on the following facts: the recipients of the shares represented that were provided with or had access to information regarding the Company sufficient to provide a basis for an informed investment decision and that they had such knowledge and experience in financial and business matters that Recipient is capable of evaluating the merits and risk of this investment; the recipients of the shares represented that they are acquiring the shares for investment purposes and without a view toward disposition of the shares; we used no advertising or general solicitation in connection with the issuance of such shares; and the shares were issued as restricted securities.

^{*} There was no market for our common stock during this period.

In contemplation of the Merger, on April 19, 2013, we entered into the Financing Agreement with Coldstream, pursuant to which we agreed to issue and sell, and Coldstream agreed to purchase or assist in securing the purchase of, \$2,000,000 worth of units in a private offering within the 12 month period following the closing of the Merger in the Coldstream Financing. Each unit issued in the Coldstream Financing is to be sold at a price of \$0.50 per share and is to consist of (i) one share of our common stock and (ii) a warrant to purchase one share of our common stock at an exercise price of \$0.75 per share and with a term of 12 months. As of the date of this Current Report on Form 8-K, we have issued and sold units consisting of 2,500,000 shares of our common stock and warrant to purchase 2,500,000 shares of our common stock in the Coldstream Financing, for aggregate gross proceeds of \$1,250,000. The proceeds of the Coldstream Financing are being used for the funding of our and ABS's ongoing business and operations. As previously disclosed, pursuant to the terms of the Merger Agreement, the amount of such proceeds raised to date was advanced to ABS prior to the closing of the Merger. The issuance of securities in the Coldstream Financing has not been registered under the Securities Act, and such securities have been issued in reliance upon an exemption from registration under Section 4(2) of the Securities Act and Regulation S promulgated thereunder. Such securities may not be offered or sold in the United States absent registration under or exemption from the Securities Act and any applicable state securities laws. In determining that the issuance of such securities qualifies for an exemption under Section 4(2) of the Securities Act and Regulation S promulgated thereunder, we have relied on the following facts: the recipients of the securities represented that they are not a "U.S. Person" as defined in as defined in Rule 902 promulgated under the Securities Act and are "accredited investors" as defined in Rule 501 under the Securities Act; and the securities were issued as restricted securities.

Upon the closing of the Merger, we issued an aggregate of 14,645,212 shares of our common stock to stakeholders of ABS in exchange for the cancellation of their shares, or rights to acquire shares, of ABS. Reference is made to the description of the Merger and the securities issued in connection therewith, as set forth under the heading "The Merger and Related Transactions—The Merger" in Item 2.01 above. The issuance of shares in connection with the Merger to stakeholders of ABS has not been registered under the Securities Act, and such shares have been issued in reliance upon an exemption from registration under Section 4(2) of the Securities Act and Rule 506 of Regulation D promulgated thereunder. Such shares may not be offered or sold in the United States absent registration under or exemption from the Securities Act and any applicable state securities laws. In determining that the issuance of such shares qualifies for an exemption under Section 4(2) of the Securities Act and Rule 506 of Regulation D promulgated thereunder, we have relied on the following facts: the recipients of the shares represented that they are acquiring the shares for investment purposes and without a view toward disposition of the shares; the recipients of the shares represented that they are "accredited investors" as defined in Rule 501 under the Securities Act or otherwise financially sophisticated; we used no advertising or general solicitation in connection with the issuance of such shares; and the shares were issued as restricted securities.

By ABS

During the three-year period preceding the date of this Current Report on Form 8-K, ABS issued convertible promissory notes in aggregate principal amount of \$1,280,397 and bearing interest at rates ranging from 6% to 10% together with related warrants to a total of 26 purchasers. Those securities were originally issued to the purchasers thereof on various dates during and prior to the past three-year period in bridge loan transactions in expectation of potential financings of ABS's capital stock. In contemplation of the Merger, any such potential financing of the capital stock of ABS was abandoned and all such securities were amended and restated to provide for (i) the conversion of all amounts owed under all outstanding convertible promissory notes into the right to receive an aggregate of 9,000,000 shares of the Company's common stock upon the closing of the Merger, calculating to approximately one share of the Company's common stock for each \$0.27 outstanding under the notes, and (ii) the cancellation of the warrants in full upon the closing of the Merger. The issuance of such convertible promissory notes and related warrants was not been registered under the Securities Act, and such securities were issued in reliance upon an exemption from registration under Section 4(2) of the Securities Act and Rule 506 of Regulation D promulgated thereunder. In determining that the issuance of such securities qualified for an exemption under Section 4(2) of the Securities Act and Rule 506 of Regulation D promulgated thereunder, ABS relied on the following facts: the securities were issued to recipients that represented they were "accredited investors" as defined in Rule 501 under the Securities Act acquiring the securities for investment purposes and without a view toward disposition thereof; ABS used no advertising or general solicitation in connection with the issuance of such securities; and the securities were issued as restricted securities.

During the three-year period preceding the date of this Current Report on Form 8-K, ABS issued an aggregate of 119,095 shares of its common stock to service providers as compensation for services rendered under ABS's former equity incentive plan. The issuance of those shares was not been registered under the Securities Act, and such securities were issued in reliance upon an exemption from registration under Rule 701 promulgated under the Securities Act. In determining that the issuance of such securities qualified for an exemption under Rule 701 promulgated under the Securities Act, ABS relied on the following facts: the securities were issued under ABS's written compensatory benefit plan; the recipients of the securities were bona fide service providers to ABS; and the securities were issued as restricted securities.

DESCRIPTION OF SECURITIES

Authorized Capital Stock; Issued and Outstanding Capital Stock

Effective May 24, 2013, we amended our Articles of Incorporation to increase our authorized common stock from 75,000,000 shares to 300,000,000 shares. Other than our common stock, we have no other class or series of authorized capital stock.

Also on May 24, 2013, we effected a forward stock split, by way of a stock dividend, of our issued and outstanding shares of common stock at a ratio of 11 shares to each one issued and outstanding share. As a result, our outstanding common stock increased from 3,960,000 shares to 43,560,000 shares immediately following the forward stock split. All share amounts of our common stock referenced in this Current Report on Form 8-K give effect to the 11-for-1 forward stock split described above, including those applicable to periods prior to the forward stock split.

As of June 26, 2013 after giving effect to the closing of the Merger, there were a total of 58,645,212 shares of our common stock issued and outstanding. Reference is made to the description of the current ownership of the Company set forth under the heading "The Merger and Related Transactions—Post-Merger Company Ownership" under Item 2.01 of this Current Report on Form 8-K.

Description of Common Stock

The holders of our common stock, par value \$0.001 per share, are entitled to one vote per share on all matters submitted to a vote of our stockholders, including the election of directors. Our articles of incorporation do not provide for cumulative voting in the election of directors, and our amended and restated bylaws provide that directors are elected by a plurality vote of the votes cast and entitled to vote on the election of directors at any meeting for the election of directors at which a quorum is present. Matters other than the election of directors to be voted on by stockholders are generally approved if, at a duly convened stockholder meeting, the number of votes cast in favor of the action exceeds the number of votes cast in opposition to the action, unless a different vote for the action is required by applicable law, our articles of incorporation or our amended and restated bylaws. Applicable Nevada law requires any amendment to our articles of incorporation to be approved by stockholders holding shares entitling them to exercise at least a majority of the voting power of the Company. The holders of our common stock will be entitled to cash dividends as may be declared, if any, by our Board of Directors from funds available. Upon liquidation, dissolution or winding up of our company, the holders of our common stock will be entitled to receive pro rata all assets available for distribution to the holders. All rights of our common stockholders described in this paragraph could be subject to any preferential voting, liquidation or other rights of any series of preferred stock that we may authorize and issue in the future. Our common stock is presently traded on the OTC Bulletin Board under the trading symbol "ARTH".

Description of Warrants

In connection with the Coldstream Financing, we have issued to one purchaser warrants to acquire up to 2,500,000 shares of our common stock, and we intend to issue additional warrants to those or other purchasers to acquire up to 1,500,000 shares of our common stock. The warrants have been issued together with shares of our common stock as units at a purchase price of \$0.50 per unit, with each unit consisting of one share of our common stock and a warrant to acquire one share of our common stock. The warrants have an exercise price of \$0.75 per share, are exercisable immediately, and have a term of exercise of 12 months following the date of issuance. The shares issuable upon exercise of the warrants are subject to adjustment for stock splits, stock dividends, reclassifications, reorganizations or other changes of the outstanding securities of the Company.

Transfer Agent

The transfer agent and registrar for our common stock is Empire Stock Transfer, 1859 Whitney Mesa Drive, Henderson, Nevada 89014.

Anti-Takeover Provisions of Nevada State Law

Some features of the Nevada Revised Statutes, which are further described below, may have the effect of deterring third parties from making takeover bids for control of us or may be used to hinder or delay a takeover bid. This would decrease the chance that our stockholders would realize a premium over market price for their shares of common stock as a result of a takeover bid.

Acquisition of Controlling Interest

The Nevada Revised Statutes contain provisions governing acquisition of a controlling interest of a Nevada corporation. These provisions provide generally that any person or entity that acquires a certain percentage of the outstanding voting shares of a Nevada corporation may be denied voting rights with respect to the acquired shares, unless certain criteria are satisfied. Our amended and restated bylaws provide that these provisions will not apply to us or to any existing or future stockholder or stockholders.

Combination with Interested Stockholder

The Nevada Revised Statutes contain provisions governing combination of a Nevada corporation that has 200 or more stockholders of record with an interested stockholder. These provisions may have the effect of delaying or making it more difficult to affect a change in control of the Company.

A corporation affected by these provisions may not engage in a combination within three years after the interested stockholder acquires his, her or its shares unless the combination or purchase is approved by the board of directors before the interested stockholder acquired such shares. Generally, if approval is not obtained, then after the expiration of the three-year period, the business combination may be consummated with the approval of the board of directors before the person became an interested stockholder or a majority of the voting power held by disinterested stockholders, or if the consideration to be received per share by disinterested stockholders is at least equal to the highest of:

- the highest price per share paid by the interested stockholder within the three years immediately preceding the date of the announcement of the combination or within three years immediately before, or in, the transaction in which he, she or it became an interested stockholder, whichever is higher;
- the market value per share on the date of announcement of the combination or the date the person became an interested stockholder, whichever is higher; or
- if higher for the holders of preferred stock, the highest liquidation value of the preferred stock, if any.

Generally, these provisions define an interested stockholder as a person who is the beneficial owner, directly or indirectly of 10% or more of the voting power of the outstanding voting shares of a corporation, and define combination to include any merger or consolidation with an interested stockholder, or any sale, lease, exchange, mortgage, pledge, transfer or other disposition, in one transaction or a series of transactions with an interested stockholder of assets of the corporation:

- having an aggregate market value equal to 5% or more of the aggregate market value of the assets of the corporation;
- having an aggregate market value equal to 5% or more of the aggregate market value of all outstanding shares of the corporation; or
- representing 10% or more of the earning power or net income of the corporation.

INDEMNIFICATION OF DIRECTORS AND OFFICERS

We have not entered into separated indemnification agreements with our directors and officers. Our amended and restated bylaws provide that we shall indemnify any director or officer to the fullest extent authorized by the laws of the State of Nevada. Our amended and restated bylaws further provide that we shall pay the expenses incurred by an officer or director (acting in his capacity as such) in defending any action, suit or proceeding in advance of the final disposition of such action, suit or proceeding, subject to the delivery to us by or on behalf of such director or officer of an undertaking to repay the amount of such expenses if it shall ultimately be determined that he or she is not entitled to be indemnified by us as authorized in our bylaws or otherwise.

The Nevada Revised Statutes provide us with the power to indemnify any of our directors, officers, employees and agents as follows:

- a corporation may indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action, suit or proceeding, whether civil, criminal, administrative or investigative, except an action by or in the right of the corporation, by reason of the fact that he is or was a director, officer, employee or agent of the corporation, or is or was serving at the request of the corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise, against expenses, including attorneys' fees, judgments, fines and amounts paid in settlement actually and reasonably incurred by him in connection with the action, suit or proceeding if he or she acted in good faith and in a manner which he or she reasonably believed to be in or not opposed to the best interests of the corporation, and, with respect to any criminal action or proceeding, had no reasonable cause to believe his or her conduct was unlawful;
- a corporation may indemnify any person who was or is a party or is threatened to be made a party to any threatened, pending or completed action or suit by or in the right of the corporation to procure a judgment in its favor by reason of the fact that he or she is or was a director, officer, employee or agent of the corporation, or is or was serving at the request of the corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise against expenses, including amounts paid in settlement and attorneys' fees actually and reasonably incurred by him or her in connection with the defense or settlement of the action or suit if he or she acted in good faith and in a manner which he or she reasonably believed to be in or not opposed to the best interests of the corporation. Indemnification may not be made for any claim, issue or matter as to which such a person has been adjudged by a court of competent jurisdiction, after exhaustion of all appeals therefrom, to be liable to the corporation or for amounts paid in settlement to the corporation, unless and only to the extent that the court in which the action or suit was brought or other court of competent jurisdiction determines upon application that in view of all the circumstances of the case, the person is fairly and reasonably entitled to indemnity for such expenses as the court deems proper; and
- to the extent that a director, officer, employee or agent of a corporation has been successful on the merits or otherwise in defense of any
 action, suit or proceeding, or in defense of any claim, issue or matter therein, the corporation must indemnify him or her against
 expenses, including attorneys' fees, actually and reasonably incurred by him or her in connection with the defense.

The Nevada Revised Statutes provide that a corporation may make any discretionary indemnification only as authorized in the specific case upon a determination that indemnification of the director, officer, employee or agent is proper in the circumstances. The determination must be made:

- by the stockholders of the corporation;
- by the board of directors of the corporation by majority vote of a quorum consisting of directors who were not parties to the action, suit
 or proceeding;
- if a majority vote of a quorum consisting of directors who were not parties to the action, suit or proceeding so orders, by independent legal counsel in a written opinion;
- if a quorum consisting of directors who were not parties to the action, suit or proceeding cannot be obtained, by independent legal counsel in a written opinion; or
- by court order.

The Nevada Revised Statutes further provide that a corporation may purchase and maintain insurance or make other financial arrangements on behalf of any person who is or was a director, officer, employee or agent of the corporation, or is or was serving at the request of the corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise for any liability asserted against him and liability and expenses incurred by him in his capacity as a director, officer, employee or agent, or arising out of his status as such, whether or not the corporation has the authority to indemnify him against such liability and expenses. We have secured a directors' and officers' liability insurance policy. We expect that we will continue to maintain such a policy.

FINANCIAL STATEMENTS

Reference is made to the financial statements and pro forma financial information relating to ABS contained in Item 9.01 of this Current Report on Form 8-K, which is incorporated herein by reference.

Our audited financial statements for the fiscal years ended September 30, 2012 and 2011 are available in our Annual Report on Form 10-K for the fiscal year ended September 30, 2012 filed with the SEC on December 31, 2012, and are incorporated herein by reference. Our unaudited financial statements for the three and six months ended Mach 31, 2013 and 2012 are available in our Quarterly Report on Form 10-Q for the quarter ended March 31, 2013 filed with the SEC on May 20, 2013, and are incorporated herein by reference.

CHANGES IN AND DISAGREEMENTS WITH ACCOUNTANTS ON ACCOUNTING AND FINANCIAL DISCLOSURE

Reference is made to the disclosure set forth in Item 4.01 of this Current Report on Form 8-K, which disclosure is incorporated herein by reference.

Item 3.02 Unregistered Sales of Equity Securities.

Reference is made to the disclosure set forth in Item 2.01 of this Current Report on Form 8-K, which disclosure is incorporated by reference into this Item 3.02.

Item 4.01 Changes in Registrant's Certifying Accountant.

(a) Effective on June 25, 2013 and with the approval of our Board of Directors, we dismissed Paritz & Co., P.A. ("Paritz") as our independent registered public accounting firm engaged to audit the Company's financial statements.

The reports issued by Paritz dated December 19, 2012 and December 29, 2011 relating to its audits of the balance sheets of the Company as of September 30, 2012 and 2011, and the related statements of operations, changes in shareholders' equity and cash flows for each of the fiscal years then ended and for the period from inception (September 16, 2009) through September 30, 2012, contained an explanatory paragraph stating that there was substantial doubt about the Company's ability to continue as a going concern. Other than as disclosed above, such reports did not contain an adverse opinion or disclaimer of opinion and were not qualified as to uncertainty, audit scope or accounting principles.

Our decision to dismiss Paritz is not the result of any disagreement between us and Paritz on matters of accounting principles or practices, financial statement disclosure or auditing scope or procedures. During the Company's two most recent fiscal years and the subsequent interim period through June 25, 2013, there were no disagreements with Paritz on any matter of accounting principles or practices, financial statement disclosure or auditing scope or procedures, which disagreements, if not resolved to the satisfaction of Paritz, would have caused Paritz to make a reference to the subject matter of the disagreement in connection with its reports. Pursuant to the rules of the SEC applicable to smaller reporting companies, Paritz was not required to provide an attestation as to the effectiveness of the Company's internal control over financial reporting for any period since the Company's inception. However, as disclosed in Item 9A of the Company's Annual Report on Form 10-K for the fiscal year ended September 30, 2012 and Part I, Item 4 of the Company's Form 10-Q for the quarterly period ended March 31, 2013, the Company's management determined that the Company's internal control over financial reporting was not effective as of the end of such periods due to the existence of the following material weaknesses:

- We have insufficient quantity of dedicated resources and experienced personnel involved in reviewing and designing internal controls. As a result, a material misstatement of the interim and annual financial statements could occur and not be prevented or detected on a timely basis.
- We do not have an audit committee. While not being legally obligated to have an audit committee, it is management's view that to have an audit committee, comprised of independent board members, is an important entity-level control over our financial statements.
- We did not perform an entity level risk assessment to evaluate the implication of relevant risks on financial reporting, including the
 impact of potential fraud-related risks and the risks related to non-routine transactions, if any, on our internal control over financial
 reporting. Lack of an entity-level risk assessment constituted an internal control design deficiency which resulted in more than a remote
 likelihood that a material error would not have been prevented or detected, and constituted a material weakness.
- We lack personnel with formal training to properly analyze and record complex transactions in accordance with U.S. GAAP.
- We have not achieved the optimal level of segregation of duties relative to key financial reporting functions.

Other than as disclosed above, there were no reportable events (as that term is defined in Item 304(a)(1)(v) of Regulation S-K) during the Company's two most recent fiscal years or during the subsequent interim period through June 25, 2013. The Company's Board of Directors discussed the subject matter referred to above with Paritz. The Company authorized Paritz to respond fully and without limitation to all requests of the successor accountant concerning all matters related to the annual and interim periods audited and reviewed by Paritz, including with respect to the subject matter of any reportable event.

We provided Paritz with a copy of the above disclosures we are making in response to Item 4.01 of this Current Report on Form 8-K and requested Paritz to furnish us with a letter addressed to the SEC stating whether or not it agrees with the above statements, and, if not, stating the respects in which it does not agree. A copy of the letter dated June 26, 2013, is filed as Exhibit 16.1 to this Current Report on Form 8-K.

(b) Effective on June 25, 2013 and with the approval of our Board of Directors, we engaged Moody, Famiglietti & Andronico, LLP ("MFA") as the Company's new independent registered public accounting firm. MFA was engaged by ABS before the closing of the Merger to audit its

financial statements for the years ended September 30, 2012 and 2011 and the related statements of operations, changes in shareholders' deficit and cash flows for each of the years then ended and for the period from inception (March 6, 2006) through September 30, 2012, which are included in this Current Report on Form 8-K.

During the Company's two most recent fiscal years and through the date of the Company's engagement of MFA on June 25, 2013, neither the Company nor anyone on its behalf consulted with MFA regarding either (i) the application of accounting principles to a specified transaction, either completed or proposed, or the type of audit opinion that might be rendered with respect to the Company's financial statements, and no written report or oral advice was provided to the Company by MFA that was an important factor considered by the Company in reaching a decision as to any accounting, auditing or financial reporting issue; or (ii) any matter that was the subject of a disagreement (as that term is defined in Item 304(a)(1)(iv) of Regulation S-K promulgated under the Securities Act and the related instructions) or a reportable event (as that term is defined in Item 304(a)(1)(v) of Regulation S-K) relating to the Company.

Item 5.01 Changes In Control of the Registrant.

Reference is made to the disclosure set forth in Item 2.01 of this Current Report on Form 8-K, which disclosure is incorporated by reference into this Item 5.01. Other than the transactions and agreements described in Item 2.01, our officers and directors know of no arrangements that may result in a change in control of the Company at a subsequent date.

Item 5.02 Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

(c) Effective on the closing of the Merger on June 26, 2013, we have appointed Alan T. Barber as our Chief Financial Officer. With his appointment, Mr. Barber will also serve as our principal financial officer and principal accounting officer. Concurrent with the effectiveness of Mr. Barber's appointment, Terrence W. Norchi will relinquish his role as Interim Chief Financial Officer and continue in his roles as President, Chief Executive Officer and a director of the Company.

In connection with his appointment as our Chief Financial Officer, we have entered into an executive employment agreement with Mr. Barber with an effective start date of June 26, 2013, pursuant to which Mr. Barber is obligated to perform his duties on a part-time basis and as compensation for such service receives an annual base salary of \$83,600. Mr. Barber's employment agreement continues until terminated by us or by Mr. Barber. Upon any termination of the employment agreement, whether by us, by Mr. Barber or as a result of Mr. Barber's death or disability, Mr. Barber is not entitled to any severance payments or benefits.

The foregoing description of the terms of Mr. Barber's employment agreement with us does not purport to be complete and is qualified in its entirety by reference to the full text of the agreement, which is attached hereto as Exhibit 10.9 to this Current Report on Form 8-K and is incorporated herein by reference.

We are not aware of any transaction relating to Mr. Barber that would require disclosure under Item 404(a) of Regulation S-K promulgated under the Securities Act and that is not disclosed herein. Reference is made to the disclosure set forth under the heading "Directors and Executive Officers—Business Experience—Alan T. Barber" in Item 2.01 of this Current Report on Form 8-K, which is incorporated into this Item 5.02(c) by reference.

- (d) Effective as of the closing of the Merger on June 26, 2013, Dr. Arthur Rosenthal was appointed as a director on our Board of Directors, to fill a vacancy created by an increase to the size of the Board of Directors. It is contemplated that Dr. Rosenthal may serve on certain committees of the Board of Directors in the future, but no such committees have been established and consequently no such appointment has been made as of the date of this Current Report on Form 8-K. We are not aware of any transaction relating to Dr. Rosenthal that would require disclosure under Item 404(a) of Regulation S-K promulgated under the Securities Act. Reference is made to the disclosure set forth under the heading "Directors and Executive Officers—Business Experience—Dr. Arthur Rosenthal" in Item 2.01 of this Current Report on Form 8-K, which is incorporated into this Item 5.02(d) by reference.
- (e) Effective as of June 26, 2013, we entered into an executive employment agreement with Dr. Terrence W. Norchi, our President and Chief Executive Officer. Dr. Norchi's employment agreement continues until terminated by us or Dr. Norchi, and provides for an initial annual base salary of \$275,000 and eligibility to receive an annual cash bonus in an amount up to 30% of Dr. Norchi's then-current annual base salary. Annual bonuses are awarded at the sole discretion of our Board of Directors.

If Dr. Norchi's employment agreement is terminated by us, unless it is terminated by us "For Cause" (as defined in the agreement), or is terminated by Dr. Norchi for "Good Reason" (as defined in the agreement), then Dr. Norchi, upon signing a release in favor of the Company, will be entitled to severance in an amount equal to 12 months of Dr. Norchi's then-current annual base salary, payable in the form of salary continuation, plus, if Dr. Norchi elects and subject to certain other conditions, payment of Dr. Norchi's premiums to continue his group health coverage under COBRA until the earlier of (i) 12 months following the date of such termination; or (ii) the date Dr. Norchi becomes covered under another employer's health plan. In addition, Dr. Norchi's employment agreement provides that, in the event of a change of control of the Company, termination by Dr. Norchi for Good Reason, termination by the Company for any reason other than For Cause, or termination as a result of Dr. Norchi's death, all unvested shares under outstanding equity grants to Dr. Norchi, if any, shall automatically accelerate and become fully vested.

Dr. Norchi's employment agreement provides the following definitions of "For Cause" and "Good Reason": (a) "For Cause" is the commission by the executive of a crime involving dishonesty, breach of trust, or physical harm to any person, executive's engagement by the executive in conduct that is in bad faith and materially injurious to the Company, commission by the executive of a material breach of the employment agreement, willful refusal by the executive to implement or follow a lawful policy or directive of the Company, or executive's engagement in misfeasance or malfeasance demonstrated by a pattern of failure to perform job duties diligently and professionally (other than any such failure resulting from Executive's incapacity due to physical or mental illness); and (b) "Good Reason" is a material reduction in executive's annual base salary, except for reductions that are comparable to reductions generally applicable to similarly-situated executives of the Company, the relocation of executive to a facility or location that is more than 50 miles from his primary place of employment and such relocation results in an increase in executive's one-way driving distance by more than 50 miles, or a material and adverse change in executive's authority, duties, or responsibilities with the Company or a material and adverse change in executive's reporting relationship within the Company.

The foregoing description of the terms of Dr. Norchi's employment agreement does not purport to be complete and is qualified in its entirety by reference to the full text of such agreement, which is attached hereto as Exhibit 10.8 and is incorporated herein by reference.

Item 5.06 Change in Shell Company Status.

We have determined that, as the result of the closing of the Merger as described above under Item 2.01 of this Current Report on Form 8-K, we have ceased to be a shell company as that term is defined in Rule 12b-2 promulgated under the Exchange Act. Reference is made to the disclosure set forth in Item 2.01 of this Current Report on Form 8-K, which disclosure is incorporated by reference into this Item 5.06.

Item 9.01 Financial Statements and Exhibits.

(a) Financial statements of business acquired.

The following are filed as Exhibit 99.1 to this Current Report on Form 8-K and are incorporated herein by reference:

The unaudited financial statements of ABS as of the six months ended March 31, 2013 and 2012.

The audited financial statements of ABS as of the years ended September 30, 2012 and 2011.

(b) Pro forma financial information.

The following is filed as Exhibit 99.2 to this Current Report on Form 8-K and are incorporated herein by reference:

The unaudited pro forma financial information of Arch Therapeutics, Inc. and its wholly owned subsidiary ABS as of the fiscal years ended September 30, 2012 and 2011 and the six months ended March 31, 2013 and 2012.

(c) Shell company transactions.

Reference is made to the disclosure set forth in Items 9.01(a) and 9.01(b), which disclosure is incorporated herein by reference.

(d) Exhibits.

xhibit	Description
2.1	Agreement and Plan of Merger dated May 10, 2013, by and among Almah, Inc., Arch Acquisition Corporation, and Arch
	Therapeutics, Inc. (incorporated by reference to Exhibit 2.1 to the Current Report on Form 8-K filed by the Company with
	the SEC on May 13, 2013)
3.1	Articles of Incorporation of Arch Therapeutics, Inc. (incorporated by reference to Exhibit 3.1 to the Registration Statement
	on Form S-1 filed by the Company with the SEC on January 5, 2012)
3.2	Certificate of Amendment to Articles of Incorporation of Arch Therapeutics, Inc. (incorporated by reference to Exhibit 3.2 to
	the Current Report on Form 8-K filed by the Company with the SEC on June 5, 2013)
3.3	Amended and Restated Bylaws of Arch Therapeutics, Inc. (incorporated by reference to Exhibit 3.1 to the Current Report of
	Form 8-K filed by the Company with the SEC on June 24, 2013)
10.1	Binding Letter of Intent by and between Almah, Inc. and Arch Therapeutics, Inc. dated April 19, 2013 (incorporated by
	reference to Exhibit 10.1 to the Current Report on Form 8-K filed by the Company with the SEC on April 25, 2013)
10.2	Promissory Note by and between Almah, Inc. and Arch Therapeutics, Inc. dated April 19, 2013 (incorporated by reference
10.2	to Exhibit 10.2 to the Current Report on Form 8-K filed by the Company with the SEC on April 25, 2013)
10.3	Financing Agreement by and between Almah, Inc. and Coldstream Summit Ltd. dated April 19, 2013 (incorporated by
10.4	reference to Exhibit 10.3 to the Current Report on Form 8-K filed by the Company with the SEC on April 25, 2013)
10.4	Form of Securities Purchase Agreement (incorporated by reference to Exhibit 10.4 to the Current Report on Form 8-K filed by the Company with the SEC on April 25, 2013)
10.5	Form of Warrant (incorporated by reference to Exhibit 10.5 to the Current Report on Form 8-K filed by the Company with
10.5	the SEC on April 25, 2013)
10.6	Amended and Restated Exclusive Patent License Agreement dated May 23, 2011 between ABS and the Massachusetts
10.0	Institute of Technology, as amended by the First Amendment to Amended and Restated Exclusive Patent License Agreemen
	dated May 15, 2012 between ABS and the Massachusetts Institute of Technology, and further amended by the Second
	Amendment to Amended and Restated Exclusive Patent License Agreement dated February 1, 2013 between ABS and the
	Massachusetts Institute of Technology, as further amended by the Third Amendment to Amended and Restated Exclusive
	Patent License Agreement dated April 30, 2013 between ABS and the Massachusetts Institute of Technology, and as further
	amended by the Letter Agreement dated June 10, 2013 between ABS and the Massachusetts Institute of Technology
10.7	Termination Agreement and Release dated June 25, 2013, between ABS and Terrence W. Norchi
10.8	Executive Employment Agreement dated June 26, 2013 between the Company and Terrence W. Norchi
10.9	Executive Employment Agreement dated June 26, 2013 between the Company and Alan T. Barber
16.1	Letter from Paritz & Co., L.P., dated June 26, 2013
21.1	List of Subsidiaries
99.1	Financial Statements of ABS
99.2	Pro Forma Financial Information of Arch Therapeutics, Inc. and its wholly owned subsidiary ABS

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

ARCH THERAPEUTICS, INC.

Dated: June 26, 2013 By: /s/ Terrence W. Norchi, M.D.

Name: Terrence W. Norchi, M.D.

Title: President, Chief Executive Officer

EXHIBIT INDEX

Exhibit	Description
2.1	Agreement and Plan of Merger dated May 10, 2013, by and among Almah, Inc., Arch Acquisition Corporation, and Arch
	Therapeutics, Inc. (incorporated by reference to Exhibit 2.1 to the Current Report on Form 8-K filed by the Company with the SEC on May 13, 2013)
3.1	Articles of Incorporation of Arch Therapeutics, Inc. (incorporated by reference to Exhibit 3.1 to the Registration Statement
	on Form S-1 filed by the Company with the SEC on January 5, 2012)
3.2	Certificate of Amendment to Articles of Incorporation of Arch Therapeutics, Inc. (incorporated by reference to Exhibit 3.2 t
2.2	the Current Report on Form 8-K filed by the Company with the SEC on June 5, 2013)
3.3	Amended and Restated Bylaws of Arch Therapeutics, Inc. (incorporated by reference to Exhibit 3.1 to the Current Report o Form 8-K filed by the Company with the SEC on June 24, 2013)
10.1	Binding Letter of Intent by and between Almah, Inc. and Arch Therapeutics, Inc. dated April 19, 2013 (incorporated by
10.1	reference to Exhibit 10.1 to the Current Report on Form 8-K filed by the Company with the SEC on April 25, 2013)
10.2	Promissory Note by and between Almah, Inc. and Arch Therapeutics, Inc. dated April 19, 2013 (incorporated by reference
10.2	to Exhibit 10.2 to the Current Report on Form 8-K filed by the Company with the SEC on April 25, 2013)
10.3	Financing Agreement by and between Almah, Inc. and Coldstream Summit Ltd. dated April 19, 2013 (incorporated by
	reference to Exhibit 10.3 to the Current Report on Form 8-K filed by the Company with the SEC on April 25, 2013)
10.4	Form of Securities Purchase Agreement (incorporated by reference to Exhibit 10.4 to the Current Report on Form 8-K filed
	by the Company with the SEC on April 25, 2013)
10.5	Form of Warrant (incorporated by reference to Exhibit 10.5 to the Current Report on Form 8-K filed by the Company with
10.6	the SEC on April 25, 2013)
10.6	Amended and Restated Exclusive Patent License Agreement dated May 23, 2011 between ABS and the Massachusetts Institute of Technology, as amended by the First Amendment to Amended and Restated Exclusive Patent License Agreement
	dated May 15, 2012 between ABS and the Massachusetts Institute of Technology, and further amended by the Second
	Amendment to Amended and Restated Exclusive Patent License Agreement dated February 1, 2013 between ABS and the
	Massachusetts Institute of Technology, as further amended by the Third Amendment to Amended and Restated Exclusive
	Patent License Agreement dated April 30, 2013 between ABS and the Massachusetts Institute of Technology, and as further
	amended by the Letter Agreement dated June 10, 2013 between ABS and the Massachusetts Institute of Technology
10.7	Termination Agreement and Release dated June 25, 2013, between ABS and Terrence W. Norchi
10.8	Executive Employment Agreement dated June 26, 2013 between the Company and Terrence W. Norchi
10.9	Executive Employment Agreement dated June 26, 2013 between the Company and Alan T. Barber
16.1	Letter from Paritz & Co., L.P., dated June 26, 2013
21.1	List of Subsidiaries
99.1	Financial Statements of ABS
99.2	Pro Forma Financial Information of Arch Therapeutics, Inc. and its wholly owned subsidiary ABS

MASSACHUSETTS INSTITUTE OF TECHNOLOGY

AND

ARCH THERAPEUTICS, INC.

AMENDED AND RESTATED EXCLUSIVE PATENT LICENSE AGREEMENT

TABLE OF CONTENTS

RECITALS	3
1. DEFINITIONS.	5
2. GRANT OF RIGHTS.	10
3. COMPANY DILIGENCE OBLIGATIONS.	12
4. ROYALTIES AND PAYMENT TERMS.	14
5. REPORTS AND RECORDS.	19
6. PATENT PROSECUTION.	21
7. INFRINGEMENT.	22
8. INDEMNIFICATION AND INSURANCE	24
9. NO REPRESENTATIONS OR WARRANTIES	26
10. ASSIGNMENT.	26
12. GENERAL COMPLIANCE WITH LAWS	28
13. TERMINATION	29
14. DISPUTE RESOLUTION.	31
15. MISCELLANEOUS.	32
APPENDIX A	36
APPENDIX B	38
EXHIBIT A	41
EXHIBIT B	41

MASSACHUSETTS INSTITUTE OF TECHNOLOGY AMENDED AND RESTATED EXCLUSIVE PATENT LICENSE AGREEMENT

This Agreement, effective as of the date set forth above the signatures of the parties below (the "EFFECTIVE DATE"), is between the Massachusetts Institute of Technology ("M.I.T."), a Massachusetts corporation, with a principal office at 77 Massachusetts Avenue, Cambridge, MA 02139-4307 and Arch Therapeutics, Inc. (f/k/a Clear Nano Solutions, Inc.) ("COMPANY"), a Massachusetts corporation having its principal office at 1 Chieftain Lane, Natick, MA 01760.

RECITALS

WHEREAS, M.I.T. is the owner of certain PATENT RIGHTS A (as later defined herein) relating to M.I.T. Case No. 6008, "Self-Assembly Of An Oligopeptide To Form A Robust Macroscopic Membranous Structure," by Todd Holmes, Curtis Lockshin, Alexander Rich and Shuguang Zhang, M.I.T. Case No. 6692, "Oligopeptide-Based Biomaterials Which Support Cell Attachment And Cell Growth And Neurite Outgrowth," by C. Michael Dipersio, Todd Holmes, Curtis Lockshin, Alexander Rich and Shuguang Zhang, M.I.T. Case No. 8813, "Encapsulation Of Chondrocytes In Self-Assembling Peptide Gel For Applications To Cartilage Tissue Engineering," by Alan J. Grodzinsky, John Kisiday and Shuguang Zhang, M.I.T. Case No. 9344, "Liver Cellular Reprogramming In Peptide Scaffold And Uses Thereof," by Carlos E. Semino, Colette Shen, James L. Sherley and Shuguang Zhang, M.I.T. Case No. 9786, "Self-assembling Peptide Scaffold Hydrogels Bridge Lesions in Central Nervous System," by Rutledge Ellis-Behnke, Gerald E. Schneider, Carlos E. Semino and Shuguang Zhang and M.I.T. Case No. 10154, "Designed Peptide Scaffold with Biological Activities and Uses Thereof," by Elsa Genove, Carlos E. Semino and Shuguang Zhang and has the right to grant licenses under said PATENT RIGHTS A;

WHEREAS, M.I.T. has the right to grant a license to Arch Therapeutics, Inc. in Field A under PATENT RIGHTS A in Appendix

WHEREAS, M.I.T. and Versitech Limited ("Versitech"), the technology transfer company of The University of Hong Kong, are the owners of certain PATENT RIGHTS B (as later defined herein) relating to **M.I.T. Case No. 11366**, "Instantaneous Hemostasis," by Rutledge Ellis-Behnke, Yu Xiang Liang, Gerald E. Schneider, Kwok-Fai So, David K C. Tay and Shuguang Zhang; and **M.I.T. Case No. 12061**, "Compositions and Methods for Promoting Hemostasis and other Physiological Activities." by Rutledge Ellis-Behnke, Yu Xiang Liang, Gerald E. Schneider, Kwok-Fai So and David K C. Tay;

<u>A</u>;

- WHEREAS, M.I.T. has the right to grant licenses under said PATENT RIGHTS B subject to the M.I.T. Versitech Joint Invention Agreement effective September 7, 2005 hereby incorporated in its entirety as <u>Appendix D</u>;
- WHEREAS, Rutledge Ellis-Behnke, an inventor of the PATENT RIGHTS and current employee of M.I.T., has acquired equity in COMPANY, the Conflict Avoidance Statement of Rutledge Ellis-Behnke is attached as Exhibit A hereto;
- WHEREAS, Rutledge Ellis-Behnke, an inventor of the PATENT RIGHTS, has acquired equity in COMPANY not resulting from this Agreement, the Inventor/Author Acknowledgment of No Equity Distribution in M.I.T.'s institutional equity share of Rutledge Ellis-Behnke is attached as Exhibit B hereto;
- WHEREAS, M.I.T.'s Vice President for Research has approved that Rutledge Ellis-Behnke, an inventor of the PATENT RIGHTS, now holds equity in COMPANY and that M.I.T. is accepting equity as partial consideration for the rights and licenses granted under this Agreement;
- WHEREAS, M.I.T. desires to have the PATENT RIGHTS developed and commercialized to benefit the public and is willing to grant a license thereunder;
- WHEREAS, COMPANY has represented to M.I.T., to induce M.I.T. to enter into this Agreement, that COMPANY shall commit itself to a thorough, vigorous and diligent program of exploiting the PATENT RIGHTS so that public utilization shall result therefrom; and
- WHEREAS, M.I.T. and COMPANY are party that certain Exclusive License Agreement dated December 5, 2007 (the "Prior Agreement"), pursuant to which the COMPANY obtained a license under the PATENT RIGHTS upon the terms and conditions set forth therein;
- WHEREAS, M.I.T. and COMPANY are party that certain First Amendment to the Prior Agreement dated October 28, 2010 (the "First Amendment"), pursuant to which the PATENT RIGHTS associated with M.I.T. Case No. 6008 were deleted from the PATENT RIGHTS A;

WHEREAS, M.I.T. subsequently granted to 3D Matrix Ltd. ("3DM") an exclusive, world-wide license to certain patent rights associated with M.I.T. Case No. 6008 in the field of medical and life sciences applications and cosmetics;

WHEREAS, M.I.T., the COMPANY and 3DM are party to that certain Non-Exclusive Patent Sublicense Agreement dated October 28, 2010, as amended by that certain First Letter Agreement dated [January 3, 2011], pursuant to which 3DM granted to COMPANY a free, fully-paid up, royalty-free, non-exclusive, world-wide sublicense under its in-licensed patent rights associated with M.I.T. Case No. 6008 in the STASIS FIELD OF USE:

WHEREAS, M.I.T. and the COMPANY now desire to amend and restate the Prior Agreement, in order to revise, clarify and restate the respective benefits and obligations of the parties as hereinafter set forth.

NOW, THEREFORE, M.I.T. and COMPANY, in consideration of the premises and for other good and valuable consideration the sufficiency of which is hereby acknowledged, hereby agree that the Prior Agreement shall be amended and restated as follows:

1. DEFINITIONS.

1.1 "AFFILIATE" shall mean any legal entity (such as a corporation, partnership, joint venture or limited liability company) that is directly or indirectly controlled by COMPANY. For the purposes of this definition, the term "control" means (i) beneficial ownership of at least fifty percent (50%) of the voting securities of a corporation or other business organization with voting securities or (ii) a fifty percent (50%) or greater interest in the net assets or profits of a partnership or other business organization without voting securities.

"CELLULAR DELIVERY" shall mean applications for the delivery of cells.

"CONFIDENTIAL INFORMATION" shall mean any confidential or proprietary information furnished by one party (the "Disclosing Party") to the other party (the "Receiving Party") in connection with this Agreement, provided that such information is specifically designated as confidential. Such CONFIDENTIAL INFORMATION shall include, without limitation, any diligence reports furnished to M.I.T. under Section 3.1, royalty reports furnished to M.I.T. under Section 5.2 and copies of sublicenses furnished to M.I.T. under Section 2.4.

"COMMERCIAL SALE" shall mean the final transfer or sale by COMPANY, an AFFILIATE or SUBLICENSEE, whether at retail, wholesale or otherwise, of any LICENSED PRODUCT to a party that is not an AFFILIATE or SUBLICENSEE hereunder. The following are not deemed to be COMMERCIAL SALE: a transfer to an AFFILIATE or SUBLICENSEE for purposes of clinical trials, research and development or other testing, or provision of samples of LICENSED PRODUCT for research, development, manufacturing or marketing purposes but not for resale.

- 1.5 "EXCLUSIVE PERIOD" shall mean the TERM as defined in Section 1.18
- 1.6 "FIELD A" shall mean all Products, Methods of Manufacture and Methods of Use Thereof for all STASIS, ADHESION AND BARRIER APPLICATIONS, but specifically excluding CELLULAR DELIVERY.
 - 1.7 "FIELD B" shall mean all fields.
 - 1.8 "LICENSED PRODUCT" shall mean any product that, in whole or in part:
 - (a) absent the license granted hereunder, would infringe one or more claims of the PATENT RIGHTS; or
 - (b) is manufactured by using a LICENSED PROCESS or that, when used, practices a LICENSED PROCESS.
- 1.9 "<u>LICENSED PROCESS</u>" shall mean any process that, absent the license granted hereunder, would infringe one or more claims of the PATENT RIGHTS or which uses a LICENSED PRODUCT.
- 1.10 "<u>NET SALES</u>" shall mean the gross amount billed by COMPANY and its AFFILIATES and SUBLICENSEES for LICENSED PRODUCTS and LICENSED PROCESSES intended for COMMERCIAL SALE, less the following:
 - (a) customary trade, quantity, or cash discounts, chargebacks and rebates to the extent actually allowed and taken;
 - (b) amounts repaid or credited by reason of rejection or return;
- (c) to the extent separately stated on purchase orders, invoices, or other documents of sale, any taxes or other governmental charges levied on the production, sale, transportation, delivery, or use of a LICENSED PRODUCT or LICENSED PROCESS which is paid by or on behalf of COMPANY; and
 - (d) outbound transportation costs prepaid or allowed and costs of insurance in transit.

If COMPANY or any of its AFFILIATES or SUBLICENSEES sell any LICENSED PRODUCT or LICENSED PROCESS in a COMMERCIAL SALE as a component of a combination of elements, then the ADJUSTED NET SALES shall be determined by multiplying NET SALES of such combination by the fraction A over A + B, in which "A" is the gross amount billed for the LICENSED PRODUCT and/or LICENSED PROCESS portion of the combination when sold separately during the REPORTING PERIOD in the country in which the sale was made, and "B" is the gross amount billed for the other elements of the combination sold separately during said REPORTING PERIOD in said country. In the event that no separate sale of either such LICENSED PRODUCT and/or LICENSED PROCESS or other elements of the combination is made during said REPORTING PERIOD in said country, the ADJUSTED NET SALES shall be determined by multiplying the NET SALES of such combination by the fraction C over C + D, in which "C" is the fully-absorbed cost of the LICENSED PRODUCT and/or LICENSED PROCESS portion of the combination, and "D" is the sum of the fully-absorbed costs of the other elements of the combination , such costs being arrived at using the standard accounting procedures of COMPANY which will be in accord with generally accepted accounting practices.

No deductions shall be made for commissions paid to individuals whether they be with independent sales agencies or regularly employed by COMPANY and on its payroll, or for cost of collections. NET SALES shall occur on the date of billing for a LICENSED PRODUCT or LICENSED PROCESS. If a LICENSED PRODUCT or a LICENSED PROCESS is distributed for COMMERCIAL SALE at a discounted price that is substantially lower than the customary price or market driven prices charged by COMPANY or distributed for COMMERCIAL SALE for non-cash consideration (whether or not at a discount), NET SALES shall be calculated based on the non-discounted amount of the LICENSED PRODUCT or LICENSED PROCESS charged to an independent third party during the same REPORTING PERIOD or, in the absence of such sales, on the fair market value of the LICENSED PRODUCT or LICENSED PROCESS. Sales or other transfers between or among COMPANY and any of its AFFILIATES for the purpose of subsequent resale to third parties shall not be included in the calculation of NET SALES.

Non-monetary consideration shall <u>not</u> be accepted by COMPANY, any AFFILIATE, or any SUBLICENSEE for any LICENSED PRODUCTS or LICENSED PROCESSES intended for COMMERCIAL SALE without the prior written consent of M.I.T.

NET SALES will be calculated only once with respect to each LICENSED PRODUCT or LICENSED PROCESS sold by COMPANY, any AFFILIATE and/or any SUBLICENSEE, even if such LICENSED PRODUCT or LICENSED PROCESS is sold more than once in the course of its transfer to the ultimate end-user. The foregoing notwithstanding, NET SALES will not include transfers among the COMPANY, any AFFILIATE and/or any SUBLICENSEE unless the recipient is the end-user.

- 1.11 "<u>PATENT CHALLENGE</u>" shall mean a challenge to the validity, patentability, enforceability and/or non-infringement of any of the PATENT RIGHTS (as defined below) or otherwise opposing any of the PATENT RIGHTS.
 - 1.12 "PATENT RIGHTS" shall mean PATENT RIGHTS A and PATENT RIGHTS B.
 - 1.13 "PATENT RIGHTS A" shall mean:
 - (a) the United States and international patents listed on Appendix A;
- (b) the United States and international patent applications and/or provisional applications listed on <u>Appendix A</u> and the resulting patents;
- (c) any patent applications resulting from the provisional applications listed on <u>Appendix A</u>, and any divisionals, continuations, continuation-in-part applications, and continued prosecution applications (and their relevant international equivalents) of the patent applications listed on <u>Appendix A</u> and of such patent applications that result from the provisional applications listed on <u>Appendix A</u>, to the extent the claims are directed to subject matter specifically described in the patent applications listed on <u>Appendix A</u>, and the resulting patents;
- (d) any patents resulting from reissues, reexaminations, or extensions (and their relevant international equivalents) of the patents described in (a), (b), and (c) above; and
- (e) international (non-United States) patent applications and provisional applications filed after the EFFECTIVE DATE and the relevant international equivalents to divisionals, continuations, continuation-in-part applications and continued prosecution applications of the patent applications to the extent the claims are directed to subject matter specifically described in the patents or patent applications referred to in (a), (b), (c), and (d) above, and the resulting patents.
 - 1.14 "PATENT RIGHTS B" shall mean:
 - (a) the United States and international patents listed on Appendix B;
- (b) the United States and international patent applications and/or provisional applications listed on $\underline{\text{Appendix B}}$ and the resulting patents;

- (c) any patent applications resulting from the provisional applications listed on $\underline{Appendix\ B}$, and any divisionals, continuations, continuation-in-part applications, and continued prosecution applications (and their relevant international equivalents) of the patent applications listed on $\underline{Appendix\ B}$ and of such patent applications that result from the provisional applications listed on $\underline{Appendix\ B}$, to the extent the claims are directed to subject matter specifically described in the patent applications listed on $\underline{Appendix\ B}$, and the resulting patents;
- (d) any patents resulting from reissues, reexaminations, or extensions (and their relevant international equivalents) of the patents described in (a), (b), and (c) above; and
- (e) international (non-United States) patent applications and provisional applications filed after the EFFECTIVE DATE and the relevant international equivalents to divisionals, continuations, continuation-in-part applications and continued prosecution applications of the patent applications to the extent the claims are directed to subject matter specifically described in the patents or patent applications referred to in (a), (b), (c), and (d) above, and the resulting patents.
- 1.15 "PRODUCT DEVELOPMENT PAYMENTS" shall mean payments to COMPANY or an AFFILIATE from a SUBLICENSEE for the purposes of funding *bona fide* research and development of LICENSED PRODUCTS or LICENSED PROCESSES and that are expressly intended only to fund or pay for:
 - (a) the purchase of equipment, supplies, products or services
- (b) the assignment of or hiring of employees and/or consultants to achieve a research or development goal for the commercialization of LICENSED PRODUCTS AND LICENSED PROCESSES, as indicated by their inclusion as specific line items in a written agreement between COMPANY or AFFILIATE and the SUBLICENSEE
 - (c) reasonable overhead charges related to the direct expense described in (a) and (b) above
- 1.16 <u>"RAISE CAPITAL"</u> shall mean (i) receive funds or property from the issuance of securities, (ii) acquire an interest in a joint venture to the extent of the proportionate share of such acquired joint venture interest in the funds or property, (iii) receive funds, services or property by way of a research or development grant from governmental, non-governmental or private sources, either as reimbursement of previously conducted research and development activities or for future research and development activities or (iv) receive funds, services or property (including, without limitation, as upfront, royalty, milestone, license maintenance, option or exclusivity fees or other payments or income) from consulting, feasibility studies, licenses, sublicenses (including from SUBLICENSEES), or contracts with partners of COMPANY.

- 1.17 "REPORTING PERIOD" shall begin on the first day of each calendar quarter and end on the last day of such calendar quarter.
- 1.18 <u>"STASIS, ADHESION AND BARRIER APPLICATIONS"</u> shall mean applications for the prevention or control of the movement or leakage of solid, fluid or gaseous substances in or on the body (including, but not limited to, bleeding, blood and blood components, cerebral spinal fluid, intestinal content or fluid, gas or air, bacteria), for hemostasis, for the prevention or control of surgical or other types of adhesions, or which function as a barrier (including, but not limited to, reduction of contamination).
- 1.19 "SUBLICENSE INCOME" shall mean any payments that COMPANY or an AFFILIATE receives from a SUBLICENSEE in consideration of the sublicense of the rights granted COMPANY and AFFILIATES under Sections 2.1 and 2.2, including without limitation license fees, milestone payments, and license maintenance fees, but specifically excluding funds received from the issuance of securities in COMPANY, funds received by way of a research or development grant from governmental, non-governmental or private sources, PRODUCT DEVELOPMENT PAYMENTS, reimbursement of manufacturing expenses and royalties on NET SALES.
- 1.20 "SUBLICENSEE" shall mean any non-AFFILIATE sublicensee of the rights granted COMPANY under-Sections 2.1 and 2.2, that has been retained for COMMERCIAL SALE.
- 1.21 "TERM" shall mean the term of this Agreement, which shall commence on the EFFECTIVE DATE and shall remain in effect until the expiration or abandonment of all issued patents and filed patent applications within the PATENT RIGHTS, unless earlier terminated in accordance with the provisions of this Agreement.
 - 1.22 "TERRITORY" shall mean worldwide.

2. GRANT OF RIGHTS.

2.1 <u>License Grants</u> under PATENT RIGHTS A. Subject to the terms of this Agreement, M.I.T. hereby grants to COMPANY and its AFFILIATES for the TERM a non-exclusive royalty-bearing license under the PATENT RIGHTS A to develop, make, have made, use, sell, offer to sell, lease, and import LICENSED PRODUCTS in the FIELD A in the TERRITORY and to develop and perform LICENSED PROCESSES in the FIELD A in the TERRITORY.

- 2.2 <u>License Grants</u> under PATENT RIGHTS B. Subject to the terms of this Agreement, M.I.T. hereby grants to COMPANY and its AFFILIATES for the TERM a royalty-bearing exclusive license under the PATENT RIGHTS B to develop, make, have made, use, sell, offer to sell, lease, and import LICENSED PRODUCTS in the FIELD B in the TERRITORY and to develop and perform LICENSED PROCESSES in the FIELD B in the TERRITORY.
- 2.3 Exclusivity under PATENT RIGHTS B. In order to establish an exclusive period for COMPANY, M.I.T. agrees that it shall not grant and has not granted any other license under the PATENT RIGHTS B to make, have made, use, sell, lease and import LICENSED PRODUCTS in the FIELD B in the TERRITORY or to perform LICENSED PROCESSES in the FIELD B in the TERRITORY until the expiration of the TERM.
- 2.4 <u>Sublicenses</u> COMPANY shall have the right to grant sublicenses of its rights under Sections 2.1 and 2.2 Any exclusivity of such sublicenses shall be limited to the PATENT RIGHTS B. The term of any sublicense may extend past the expiration date of the EXCLUSIVE PERIOD, but any exclusivity of such sublicense shall expire upon the expiration of the EXCLUSIVE PERIOD. COMPANY shall incorporate terms and conditions into its sublicense agreements sufficient to enable COMPANY to comply with this Agreement. COMPANY shall also include provisions in all sublicenses to provide that in the event that SUBLICENSEE brings a PATENT CHALLENGE against M.I.T. or assists another party in bringing a PATENT CHALLENGE against M.I.T. (except as required under a court order or subpoena) then COMPANY may terminate the sublicense. COMPANY shall promptly furnish M.I.T. with a fully signed photocopy of any sublicense agreement. Upon termination of this Agreement for any reason, any SUBLICENSEE not then in default shall have the right to seek a license from M.I.T. M.I.T. agrees to negotiate such licenses in good faith under reasonable terms and conditions.

2.5 Retained Rights.

- (a) M.I.T. M.I.T. retains the right to practice under the PATENT RIGHTS for research, teaching, and educational purposes.
- (b) <u>Federal Government</u>. COMPANY acknowledges that the U.S. federal government retains a royalty-free, non-exclusive, non-transferable license to practice any government-funded invention claimed in any PATENT RIGHTS as set forth in 35 U.S.C. §§ 201-211, and the regulations promulgated thereunder, as amended, or any successor statutes or regulations.

2.6 No Additional Rights. Nothing in this Agreement shall be construed to confer any rights upon COMPANY by implication, estoppel, or otherwise as to any technology or patent rights of M.I.T. or any other entity other than the PATENT RIGHTS, regardless of whether such technology or patent rights shall be dominant or subordinate to any PATENT RIGHTS.

3. COMPANY DILIGENCE OBLIGATIONS.

- 3.1 <u>Diligence Requirements</u>. COMPANY shall use diligent efforts, or shall cause its AFFILIATES and SUBLICENSEES to use diligent efforts, to develop LICENSED PRODUCTS or LICENSED PROCESSES and to introduce LICENSED PRODUCTS or LICENSED PROCESSES into the commercial market; thereafter, COMPANY or its AFFILIATES or SUBLICENSEES shall make LICENSED PRODUCTS or LICENSED PROCESSES reasonably available to the public. Specifically, COMPANY or AFFILIATE or SUBLICENSEE shall fulfill the following obligations:
- (a) M.I.T. acknowledges that COMPANY has previously furnished M.I.T. a written research and development plan describing the major tasks to be achieved in order to bring to market a LICENSED PRODUCT or a LICENSED PROCESS, specifying the number of staff and other resources to be devoted to such commercialization effort.
- (b) Within ninety (90) days after the end of each calendar year, COMPANY shall furnish M.I.T. with a written report on the progress of its efforts during the immediately preceding calendar year to develop and commercialize LICENSED PRODUCTS or LICENSED PROCESSES. The report shall also contain a discussion of intended efforts and sales projections for the year in which the report is submitted.
- (c) COMPANY shall permit an in-plant inspection by M.I.T. at regular intervals with at least six (6) months between each such inspection.
- (d) M.I.T. acknowledges that COMPANY successfully RAISED CAPITAL in the amount of at least One Million Dollars (\$1,000,000) prior to January 1, 2009.

- (e) In the aggregate, COMPANY shall RAISE CAPITAL of at least Four Million Dollars (\$4,000,000) by July 1, 2013.
- (f) M.I.T. acknowledges that COMPANY successfully funded the equivalent of no less than Three Hundred Thousand Dollars (\$300,000) (it being agreed that such funding may take the form of cash or cash equivalents, and/or equity grants and/or contributed/granted/donated services performed by third parties if such services result data or other results which can be accessed by COMPANY) toward the development or commercialization of LICENSED PRODUCTS and/or LICENSED PROCESSES in 2007 (prorated for partial year). M.I.T. acknowledges that COMPANY successfully funded the equivalent of no less than Six Hundred Thousand Dollars (\$600,000) (it being agreed that such funding may take the form of cash or cash equivalents, and/or equity grants and/or contributed/granted/donated services performed by third parties if such services result data or other results which can be accessed by COMPANY) toward the development or commercialization of LICENSED PRODUCTS and/or LICENSED PROCESSES in 2008, 2009 and 2010.
- (h) COMPANY shall fund the equivalent of no less than Six Hundred Thousand Dollars (\$600,000) (it being agreed that such funding may take the form of cash or cash equivalents, and/or equity grants and/or contributed/granted/donated services performed by third parties if such services result data or other results which can be accessed by COMPANY) toward the development or commercialization of LICENSED PRODUCTS and/or LICENSED PROCESSES in each calendar year (pro-rated for partial years) beginning in 2011 and ending with the first commercial sale of a LICENSED PRODUCT or a first commercial performance of a LICENSED PROCESS.
- (i) M.I.T. acknowledges that on or before July 1, 2009, COMPANY initiated an animal trial to collect data on the safety and effectiveness of a LICENSED PRODUCT.
- (j) M.I.T. acknowledges that COMPANY has entered into a commercially reasonable business arrangement with at least one manufacturing partner.
- (k) COMPANY will initiate a pre-investigational or comparable consultation with an appropriate center within a regulatory authority by 2014.
- (l) COMPANY shall submit an investigational new drug application, investigational device exemption or comparable application to an appropriate center within a regulatory agency for sale of a LICENSED PRODUCT by January 1, 2018 or achieve NET SALES of a LICENSED PRODUCT and/or a first commercial performance of a LICENSED PROCESS on or before January 1, 2019 in excess of \$500,000.

In the event that COMPANY (or an AFFILIATE or SUBLICENSEE) has failed to fulfill any of its obligations under this Section 3.1, then M.I.T. may treat such failure as a material breach in accordance with Section 12.3(b).

4. ROYALTIES AND PAYMENT TERMS.

4.1 Consideration for Grant of Rights.

(a) <u>License Issue Fee and Patent Cost Reimbursement</u>. M.I.T. acknowledges that COMPANY paid to M.I.T. on the effective date of the Prior Agreement a license issue fee of Twenty-five Thousand dollars (\$25,000), and, in accordance with Section 6.3, reimbursed M.I.T. for its actual expenses incurred as of the effective date of the Prior Agreement in connection with obtaining the PATENT RIGHTS. These payments are nonrefundable and are pursuant to the schedule outlined in Section 6.3(c).

(b) <u>License Maintenance Fees</u>. M.I.T. acknowledges that COMPANY previously paid M.I.T a license maintenance fee in the amount of \$10,000 for the calendar year commencing on January 1, 2009 and in the amount of \$10,000 for the calendar year commencing on January 1, 2010. There shall be no license maintenance fee for the calendar year commencing on January 1, 2011. COMPANY shall pay to M.I.T. the following license maintenance fees within 30 (thirty) days of the dates set forth below:

January 1, 2012 \$25,000 and each January 1 of every year thereafter \$25,000

This annual license maintenance fee is nonrefundable; however, the license maintenance fee shall be credited to running royalties subsequently due on NET SALES earned during the same calendar year, if any. License maintenance fees paid in excess of running royalties due in such calendar year shall not be creditable to amounts due for future years.

- (c) <u>Running Royalties</u>. COMPANY shall pay to M.I.T. a running royalty of three percent (3%) of NET SALES or ADJUSTED NET SALES, as appropriate, by COMPANY, AFFILIATES and SUBLICENSEES. Running royalties shall be payable for each REPORTING PERIOD and shall be due to M.I.T. within ninety (90) days of the end of each REPORTING PERIOD.
- (d) Sharing of SUBLICENSE INCOME. COMPANY shall pay M.I.T. a total of fifteen percent (15%) of all SUBLICENSE INCOME received by COMPANY or AFFILIATES, excluding running royalties on NET SALES of SUBLICENSEES (which are payable pursuant to clause (c) above).

Such amount shall be payable for each REPORTING PERIOD and shall be due to M.I.T. within ninety (90) days of the end of each REPORTING PERIOD.

- (e) <u>Milestone Payment</u>. COMPANY shall pay to M.I.T. a one-time milestone payment of Fifty Thousand Dollars (\$50,000) upon the first COMMERCIAL SALE of a LICENSED PRODUCT by COMPANY, its AFFILIATE or SUBLICENSEE. This payment is nonrefundable and amounts due pursuant to this Section will be paid to M.I.T. over the subsequent 12 months in approximately equal quarterly installments within thirty (30) days of invoicing.
- (f) Third Party Royalty Offset. If COMPANY or an AFFILIATE or SUBLICENSEE is required to pay royalties to one or more third parties to obtain a patent license in the absence of which it could not legally make, import, use, sell, or offer for sale LICENSED PRODUCTS or LICENSED PROCESSES in any country, and COMPANY provides M.I.T. with reasonably satisfactory evidence of such third-party royalty payment requirements, including a signed copy of any such license agreements, then COMPANY may offset a total of fifty percent (50%) of such third-party payments against running royalties owed to M.I.T. as defined in Section 4.1 (c) above, hereunder, provided that COMPANY requires such third parties to offset its royalties as a result of royalties paid to M.I.T. by at least the same amount and provided that in no one year shall the aggregate of all such expenses be credited against more than fifty percent (50%) of royalty payments that would otherwise be due to M.I.T.
- (g) <u>Consequences of a PATENT CHALLENGE</u>. In the event that (i) COMPANY or any of its AFFILIATES brings a PATENT CHALLENGE against M.I.T., or (ii) COMPANY or any of its AFFILIATES assists another party in bringing a PATENT CHALLENGE against M.I.T. (except as required under a court order or subpoena), and (iii) M.I.T. does not choose to exercise its rights to terminate this Agreement pursuant to Section 12.4, then in the event that such a PATENT CHALLENGE is successful, COMPANY will have no right to recoup any royalties paid during the period of challenge. In the event that a PATENT CHALLENGE is unsuccessful, COMPANY shall reimburse M.I.T. for all reasonable legal fees and expenses incurred in its defense against the PATENT CHALLENGE.

(h) No Multiple Royalties. If the manufacture, use, lease, or sale of any LICENSED PRODUCT or the performance of any LICENSED PROCESS is covered by more than one of the PATENT RIGHTS, multiple royalties shall <u>not</u> be due.

(i) Equity.

(i) <u>Initial Grant.</u> COMPANY shall issue a total of <u>FORTY SIX THOUSAND SEVEN HUNDRED</u> (46,700) shares of Common Stock of COMPANY, no par value per share, (the "Shares") in the name of M.I.T. and of such persons as M.I.T. shall direct ("M.I.T. Holder"), and each M.I.T. Holder shall receive such number of shares as M.I.T. shall direct. Such issuance shall be recorded on the Stock Transfer Ledger of COMPANY on the EFFECTIVE DATE and the Shares shall be delivered to M.I.T. and M.I.T. Holders, if any, within thirty (30) days of the EFFECTIVE DATE.

COMPANY represents to M.I.T. that, as of the Effective Date, the aggregate number of Shares equals Four Percent (4%) of the COMPANY's issued and outstanding Common Stock calculated on a "Fully Diluted Basis." For purposes of this Section 4.1(i), "Fully Diluted Basis" shall mean that the total number of issued and outstanding shares of the COMPANY's Common Stock shall be calculated to include conversion of all issued and outstanding securities then convertible into common stock, the exercise of all then outstanding options and warrants to purchase shares of common stock, whether or not then exercisable, and shall assume the issuance or grant of all securities reserved for issuance pursuant to any COMPANY stock or stock option plan in effect on the date of the calculation.

(ii) Anti-Dilution Protection. COMPANY shall issue additional shares of Common Stock to M.I.T. as it pertains to the equity grant in Section 4.1(i) of this License Agreement signed by the Technology Licensing Office on behalf of M.I.T. and each M.I.T. Holder pro rata, such that M.I.T. 's and each M.I.T. Holders' aggregate ownership of the outstanding Common Stock shall not fall below Four Percent (4%) on a Fully Diluted Basis, as calculated after giving effect to the anti-dilutive issuance. Such issuances shall continue until a total of Four Million Dollars (\$4,000,000) in cash in exchange for COMPANY's capital stock (the "Funding Threshold") shall be received by COMPANY. Thereafter, no additional shares shall be due to M.I.T. or any M.I.T. Holder pursuant to this section. For purposes of clarity, this anti-dilution provision does not provide anti-dilution protection to the Deshpande Center for Technological Innovation at M.I.T., which holds shares in the COMPANY under a separate Agreement.

(iii) <u>Participation in Future Private Equity Offerings</u>. After the date of the Funding Threshold, M.I.T. (specifically not including M.I.T. Holders) shall have the right to purchase additional shares of the COMPANY's Common Stock in any private offering by the COMPANY of its equity securities in exchange for cash, to maintain its pro rata ownership as calculated immediately prior to such offering on a Fully Diluted Basis, pursuant to the terms and conditions at least as favorable as those granted to the other offerees. All rights granted to M.I.T. pursuant to this Section 4.1(i) (iii) shall terminate immediately after the COMPANY RAISES CAPITAL in excess of \$6.0 million.

(iv) Adjustments for Punitive Round Financings. After the date of the Funding Threshold (the "Funding Threshold Date"), if COMPANY issues Common Stock, or any equity security exercisable for or convertible into Common Stock, such that the price per share of COMPANY's Common Stock is less than the M.I.T. Share Price (as defined below) (a "Dilutive Issuance"), then immediately following such Dilutive Issuance, COMPANY shall issue to M.I.T. shares of Common Stock such that the M.I.T. Share Number (as defined below) equals the product obtained by multiplying the M.I.T. Share Number in effect immediately before the Dilutive Issuance by the Adjustment Fraction defined below. The M.I.T. Share Price in effect immediately after the Dilutive Issuance by the Adjustment Fraction defined below.

The Adjustment Fraction equals: $\frac{(A + C)}{(A + B)}$

where:

A = the number of shares of Common Stock issued and outstanding on a Fully Diluted Basis immediately prior to

the Dilutive Issuance.

B = the number of shares of Common Stock that could be purchased at the M.I.T. Share Price immediately prior to the Dilutive Issuance using the net aggregate consideration received by COMPANY in connection with the Dilutive Issuance.

C = the number of shares of Common Stock or of a security exercisable for or convertible into Common Stock issued, on a Fully Diluted Basis, pursuant to the Dilutive Issuance.

In addition, the following definitions shall apply to this Section 4.1(i) (iv):

"Fair Market Value" of a share of Common Stock shall be the

highest price per share that the COMPANY could obtain from a willing buyer (not a current employee or director) for shares of Common Stock sold by the COMPANY, from authorized but unissued shares, as determined in good faith by the Board of Directors of the COMPANY, unless the COMPANY shall become subject to a merger, acquisition or other consolidation pursuant to which the COMPANY is not the surviving party, in which case the current fair market value of a share of Common Stock shall be deemed to be the value received by the holders of the COMPANY's Common Stock for each share of Common Stock pursuant to the COMPANY's acquisition.

"M.I.T. Share Number" shall mean the number of shares of

COMPANY's Common Stock that M.I.T. owns on the date of the Dilutive Issuance, as adjusted from time to time pursuant to this section. Notwithstanding the foregoing, any shares of Common Stock acquired by M.I.T. pursuant to Section 4.1(i) (iii) shall not be included in the M.I.T. Share Number.

"M.I.T. Share Price" shall mean the value per share of the shares of Common Stock included in the M.I.T. Share Number, as adjusted from time to time pursuant to this section. For purposes of this section, the initial M.I.T. Share Price to be used in an adjustment resulting from the first Dilutive Issuance to occur after the Funding Threshold Date shall be the Fair Market Value per share of the Common Stock of the COMPANY effective on the Funding Threshold Date.

All rights granted to M.I.T. pursuant to this Section 4.1(i) (iv) shall terminate immediately after the COMPANY RAISES CAPITAL in excess of \$6.0 million.

4.2 <u>Royalty Buy-out</u>. At any time prior to thirty (30) days following the earlier of (i) all or substantially all of the COMPANY being acquired by a third party, or (ii) COMPANY's initial public offering of securities, COMPANY or its successor entity may eliminate its future obligation to pay License Maintenance Fees as set forth in Section 4.1 (b), Running Royalties as set forth in Section 4.1(c) and sharing of SUBLICENSE INCOME as set forth in Section 4.1(d) for a one-time and non-refundable payment to M.I.T. of Seven and One Half Million Dollars (\$7,500,000). All other responsibilities under this Agreement will remain in force for the life of the agreement.

4.3 Payments.

- (a) <u>Method of Payment</u>. All payments under this Agreement should be made payable to "Massachusetts Institute of Technology" and sent to the address identified in Section 14.1. Each payment should reference this Agreement and identify the obligation under this Agreement that the payment satisfies.
- (b) <u>Payments in U.S. Dollars</u>. All payments due under this Agreement shall be drawn on a United States bank and shall be payable in United States dollars. Conversion of foreign currency to U.S. dollars shall be made at the conversion rate existing in the United States (as reported in the <u>Wall Street Journal</u>) on the last working day of the calendar quarter of the applicable REPORTING PERIOD. Such payments shall be without deduction of exchange, collection, or other charges, and, specifically, without deduction of withholding or similar taxes or other government imposed fees or taxes, except as permitted in the definition of NET SALES.
- (c) <u>Late Payments</u>. Any payments by COMPANY that are not paid on or before the date such payments are due under this Agreement shall bear interest, to the extent permitted by law, at two percentage points above the Prime Rate of interest as reported in the <u>Wall Street Journal</u> on the date payment is due.

5. REPORTS AND RECORDS.

- 5.1 Frequency of Reports.
- (a) <u>Upon First COMMERCIAL SALE of a LICENSED PRODUCT or Commercial Performance of a LICENSED PROCESS</u>. COMPANY shall report to M.I.T. the date of first COMMERCIAL SALE of a LICENSED PRODUCT and the date of first commercial performance of a LICENSED PROCESS within ninety (90) days of occurrence in each country.
- (b) <u>After First COMMERCIAL SALE</u>. After the first COMMERCIAL SALE of a LICENSED PRODUCT or first commercial performance of a LICENSED PROCESS, COMPANY shall deliver reports to M.I.T. within ninety (90) days of the end of each REPORTING PERIOD, containing information concerning the immediately preceding REPORTING PERIOD, as further described in Section 5.2.

- 5.2 <u>Content of Reports and Payments</u>. Each report delivered by COMPANY to M.I.T. shall contain at least the following information for the immediately preceding REPORTING PERIOD:
- (i) the number of LICENSED PRODUCTS sold, leased or distributed by COMPANY, its AFFILIATES and SUBLICENSEES to independent third parties in each country, and, if applicable, the number of LICENSED PRODUCTS used by COMPANY, its AFFILIATES and SUBLICENSEES in the provision of services in each country;
- (ii) a description of LICENSED PROCESSES performed by COMPANY, its AFFILIATES and SUBLICENSEES in each country as may be pertinent to a royalty accounting hereunder;
- (iii) the gross price charged by COMPANY, its AFFILIATES and SUBLICENSEES for each LICENSED PRODUCT and, if applicable, the gross price charged for each LICENSED PRODUCT used to provide services in each country; and the gross price charged for each LICENSED PROCESS performed by COMPANY, its AFFILIATES and SUBLICENSEES in each country;
- (iv) calculation of NET SALES for the applicable REPORTING PERIOD in each country, including a listing of applicable deductions;
- (v) total royalty payable on NET SALES, or ADJUSTED NET SALES, as appropriate, in U.S. dollars, together with the exchange rates used for conversion;
- (vi) the amount of SUBLICENSE INCOME received by COMPANY from each SUBLICENSEE and the amount due to M.I.T. from such SUBLICENSE INCOME, including an itemized breakdown of the sources of income comprising the SUBLICENSE INCOME; and
- (vii) the number of sublicenses entered into for the PATENT RIGHTS, LICENSED PRODUCTS and/or LICENSED PROCESSES.

If no amounts are due to M.I.T. for any REPORTING PERIOD, the report shall so state.

5.3 <u>Financial Statements</u>. On or before the ninetieth (90th) day following the close of COMPANY's fiscal year, COMPANY shall provide M.I.T. with COMPANY's financial statements for the preceding fiscal year including, at a minimum, a balance sheet and an income statement, reviewed by COMPANY's treasurer or chief financial officer or by an independent auditor.

5.4 <u>Records</u> . COMPANY shall maintain, and shall cause its AFFILIATES and SUBLICENSEES to maintain, complete and accurate
records relating to the rights and obligations under this Agreement and any amounts payable to M.I.T. in relation to this Agreement, which
records shall contain sufficient information to permit M.I.T. to confirm the accuracy of any reports delivered to M.I.T. and compliance in other
respects with this Agreement. The relevant party shall retain such records for at least five (5) years following the end of the calendar year to
which they pertain, during which time M.I.T., or M.I.T.'s appointed agents, shall have the right, at M.I.T.'s expense, to inspect such records
during normal business hours to verify any reports and payments made or compliance in other respects under this Agreement. In the event that
any audit performed under this Section reveals an underpayment in excess of five percent (5%), COMPANY shall bear the full cost of such
audit and shall remit any amounts due to M.I.T. within thirty (30) days of receiving notice thereof from M.I.T.

6. PATENT PROSECUTION.

- 6.1 <u>Responsibility for PATENT RIGHTS</u>. M.I.T. shall prepare, file, prosecute, and maintain all of the PATENT RIGHTS. COMPANY shall have reasonable opportunities to advise M.I.T. and shall cooperate with M.I.T. in such filing, prosecution and maintenance.
- $6.2 \underline{\text{International (non-United States) Filings.}}$ Appendix C is a list of countries in which patent applications corresponding to the United States patent applications listed in $\underline{\text{Appendix B}}$ shall be filed, prosecuted, and maintained. $\underline{\text{Appendix C}}$ may be amended by mutual agreement of COMPANY and M.I.T.
 - 6.3 Payment of Expenses.
- (a) <u>PATENT RIGHTS A</u>. COMPANY shall be responsible for payment of twenty percent (20%) of all fees and costs, including attorneys' fees, relating to the filing, prosecution and maintenance of the PATENT RIGHTS A incurred after the EFFECTIVE DATE. Notwithstanding the foregoing, COMPANY shall not be required to reimburse M.I.T. for any such expenses associated with <u>PATENT RIGHTS A</u> accrued through February 28, 2011 (approximately \$[_____]) until March 1, 2012.

- (b) <u>PATENT RIGHTS B.</u> Payment of all fees and costs, including attorneys' fees, relating to the filing, prosecution and maintenance of the PATENT RIGHTS B shall be the responsibility of COMPANY, whether such amounts were incurred before or after the EFFECTIVE DATE. Notwithstanding the foregoing, COMPANY shall not be required to reimburse M.I.T. for any such expenses associated with <u>PATENT RIGHTS B</u> accrued through February 28, 2011 (approximately \$[45,000]) until March 1, 2012. For clarity, COMPANY shall be responsible for payment of ongoing patent expenses as they are incurred for the period beginning March 1, 2011.
- (c) M.I.T shall provide COMPANY with a non-binding good faith estimate of the above-described expenses for the subsequent twelve (12) month period every twelve (12) months, on July 1 of each year.
- (d) In all instances, M.I.T. shall pay the fees prescribed for large entities to the United States Patent and Trademark Office provided that COMPANY submits appropriate certification to M.I.T.

7. INFRINGEMENT.

- 7.1 <u>Notification of Infringement</u>. Each party agrees to provide written notice to the other party promptly after becoming aware of any infringement of the PATENT RIGHTS.
 - 7.2 Right to Prosecute Infringements.
- (a) <u>COMPANY</u> Right to <u>Prosecute PATENT RIGHTS B</u>. So long as COMPANY remains the exclusive licensee of the PATENT RIGHTS B in the FIELD B in the TERRITORY, COMPANY, to the extent permitted by law, shall have the right, under its own control and at its own expense, to prosecute any third party infringement of the PATENT RIGHTS B in the FIELD B in the TERRITORY, subject to Sections 7.4 and 7.5. If required by law, M.I.T. shall permit any action under this Section to be brought in its name, including being joined as a party-plaintiff, provided that COMPANY shall hold M.I.T. harmless from, and indemnify M.I.T. against, any costs, expenses, or liability that M.I.T. incurs in connection with such action.

Prior to commencing any such action, COMPANY shall consult with M.I.T. and shall consider the views of M.I.T. regarding the advisability of the proposed action and its effect on the public interest. COMPANY shall not enter into any settlement, consent judgment, or other voluntary final disposition of any infringement action under this Section without the prior written consent of M.I.T.

- (b) M.I.T. Right to Prosecute PATENT RIGHTS B. In the event that COMPANY is unsuccessful in persuading the alleged infringer to desist or fails to have initiated an infringement action within a reasonable time after COMPANY first becomes aware of the basis for such action, M.I.T. shall have the right, at its sole discretion, to prosecute such infringement under its sole control and at its sole expense, and any recovery obtained shall belong to M.I.T.
- (c) M.I.T. Right to Prosecute PATENT RIGHTS A: M.I.T. shall have the right, but shall not be obligated, to prosecute at its own expense all infringements of the PATENT RIGHTS A and, in furtherance of such right, COMPANY hereby agrees that M.I.T. may include COMPANY as a party plaintiff in any such suit, without expense to COMPANY. M.I.T. shall consult with COMPANY and shall consider its views. The total cost of any such infringement action commenced or defended solely by M.I.T. shall be borne by M.I.T., and M.I.T. shall keep any recovery or damages derived therefrom, whether compensatory for past infringement or punitive.
- M.I.T. shall not enter into any settlement, consent judgment, or other voluntary final disposition of any infringement action under this Section without the prior notification of COMPANY. Should M.I.T. decide not to pursue its right to prosecute, M.I.T will notify COMPANY, and will request from 3DM-Japan that COMPANY be granted standing to enforce the PATENT RIGHTS A.
- 7.3 <u>Declaratory Judgment Actions</u>. In the event that a PATENT CHALLENGE TO PATENT RIGHTS B is brought against M.I.T. or COMPANY by a third party, COMPANY, at its option, shall have the right within twenty (20) days after commencement of such action to take over the sole defense of the action at its own expense, subject to Sections 7.4 and 7.5. If COMPANY does not exercise this right, M.I.T. may take over the sole defense of the action at M.I.T.'s sole expense.
- 7.4 Offsets. COMPANY may offset a total of fifty percent (50%) of any expenses incurred under Sections 7.2 and 7.3 against any payments due to M.I.T. under Article 4, provided that in no event shall such payments under Article 4, when aggregated with any other offsets and credits allowed under this Agreement, be reduced by more than fifty percent (50%) in any REPORTING PERIOD.

- 7.5 Recovery. Any recovery obtained in an action brought by COMPANY under Sections 7.2 or 7.3 shall be distributed as follows: (i) each party shall be reimbursed for any expenses incurred in the action (including the amount of any royalty or other payments withheld from M.I.T. as described in Section 7.4), (ii) as to ordinary damages, COMPANY shall receive an amount equal to its lost profits or a reasonable royalty on the infringing sales, or whichever measure of damages the court shall have applied, and COMPANY shall pay to M.I.T. based upon such amount a reasonable approximation of the royalties and other amounts that COMPANY would have paid to M.I.T. if COMPANY had sold the infringing products, processes and services rather than the infringer, and (iii) as to special or punitive damages, the parties shall share equally in any award.
- 7.6 <u>Cooperation</u>. Each party agrees to cooperate in any action under this Article which is controlled by the other party, provided that the controlling party reimburses the cooperating party promptly for any costs and expenses incurred by the cooperating party in connection with providing such assistance.
- 7.7 Right to Sublicense. So long as COMPANY remains the exclusive licensee of the PATENT RIGHTS B in the FIELD B in the TERRITORY, COMPANY shall have the sole right to sublicense any alleged infringer in the FIELD B in the TERRITORY for future use of the PATENT RIGHTS B in accordance with the terms and conditions of this Agreement relating to sublicenses. Any upfront fees as part of such sublicense shall be shared equally between COMPANY and M.I.T.; other revenues to COMPANY pursuant to such sublicense shall be treated as set forth in Article 4.

8. INDEMNIFICATION AND INSURANCE

8.1 Indemnification.

(a) Indemnity of M.I.T. COMPANY shall indemnify, defend, and hold harmless M.I.T., Versitech and their trustees, officers, faculty, students, employees, and agents and their respective successors, heirs and assigns (the "Indemnitees"), against any liability, damage, loss, or expense (including reasonable attorneys' fees and expenses) incurred by or imposed upon any of the Indemnitees in connection with any claims, suits, investigations, actions, demands or judgments arising out of or related to the exercise of any rights granted to COMPANY under this Agreement. For purposes of clarity, COMPANY shall not be required to indemnify M.I.T. for any claims by third party licensees or sublicensees of PATENT RIGHTS A claiming that M.I.T. improperly granted COMPANY any rights under this Agreement.

- (b) Indemnity of COMPANY. M.I.T. warrants that it has the right to enter into this license agreement.
- (c) <u>Procedures</u>. The Indemnitees agree to provide COMPANY with prompt written notice of any claim, suit, action, demand, or judgment for which indemnification is sought under this Agreement. COMPANY agrees, at its own expense, to provide attorneys reasonably acceptable to M.I.T. to defend against any such claim. The Indemnitees shall cooperate fully with COMPANY in such defense and will permit COMPANY to conduct and control such defense and the disposition of such claim, suit, or action (including all decisions relative to litigation, appeal, and settlement); provided, however, that any Indemnitee shall have the right to retain its own counsel, at the expense of COMPANY, if representation of such Indemnitee by the counsel retained by COMPANY would be inappropriate because of actual or potential differences in the interests of such Indemnitee and any other party represented by such counsel. COMPANY agrees to keep M.I.T. informed of the progress in the defense and disposition of such claim and to consult with M.I.T. with regard to any proposed settlement.
- 8.2 <u>Insurance</u>. COMPANY shall obtain and carry in full force and effect commercial general liability insurance including product liability, and errors and omissions insurance which shall protect COMPANY and Indemnitees with respect to events covered by Section 8.1(a) above. Such insurance (i) shall be issued by an insurer licensed to practice in the Commonwealth of Massachusetts or an insurer preapproved by M.I.T., such approval not to be unreasonably withheld, (ii) shall list M.I.T. as an additional insured thereunder, (iii) shall be endorsed to include product liability coverage, and (iv) shall endeavor to require thirty (30) days written notice to be given to M.I.T. prior to any cancellation or material change thereof. Product liability insurance shall be in place thirty (30) days prior to the initiation of a human clinical trial. Errors and Omissions insurance will be in place thirty (30) days prior to initiation of any professional Consulting services. The limits of such insurance shall not be less than One Million Dollars (\$1,000,000) per occurrence with an aggregate of Two Million Dollars (\$2,000,000) for general liability; and One Million Dollars (\$1,000,000) per occurrence with an aggregate of Two Million Dollars (\$2,000,000) for errors and omissions. In the alternate, COMPANY may self-insure subject to prior approval of M.I.T. COMPANY shall provide M.I.T. with Certificates of Insurance evidencing compliance with this Section. COMPANY shall continue to maintain such insurance or self-insurance after the expiration or termination of this Agreement during any period in which COMPANY or any AFFILIATE or SUBLICENSEE continues (i) to make, use, or sell a product that was a LICENSED PRODUCT under this Agreement or (ii) to perform a service that was a LICENSED PROCESS under this Agreement, and thereafter for a period of five (5) years.

9. NO REPRESENTATIONS OR WARRANTIES

EXCEPT AS MAY OTHERWISE BE EXPRESSLY SET FORTH IN THIS AGREEMENT, M.I.T. AND VERSITECH MAKE NO REPRESENTATIONS OR WARRANTIES OF ANY KIND CONCERNING THE PATENT RIGHTS, EXPRESS OR IMPLIED, INCLUDING WITHOUT LIMITATION WARRANTIES OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, NONINFRINGEMENT, VALIDITY OF PATENT RIGHTS CLAIMS, WHETHER ISSUED OR PENDING, AND THE ABSENCE OF LATENT OR OTHER DEFECTS, WHETHER OR NOT DISCOVERABLE. Specifically, and not to limit the foregoing, M.I.T. and Versitech make no warranty or representation (i) regarding the validity or scope of the PATENT RIGHTS, and (ii) that the exploitation of the PATENT RIGHTS or any LICENSED PRODUCT or LICENSED PROCESS will not infringe any patents or other intellectual property rights of M.I.T. or Versitech or of a third party.

IN NO EVENT SHALL M.I.T., VERSITECH, THEIR TRUSTEES, DIRECTORS, OFFICERS, EMPLOYEES AND AFFILIATES BE LIABLE FOR INCIDENTAL OR CONSEQUENTIAL DAMAGES OF ANY KIND, INCLUDING ECONOMIC DAMAGES OR INJURY TO PROPERTY AND LOST PROFITS, REGARDLESS OF WHETHER M.I.T. OR VERSITECH SHALL BE ADVISED, SHALL HAVE OTHER REASON TO KNOW, OR IN FACT SHALL KNOW OF THE POSSIBILITY OF THE FOREGOING.

M.I.T. WARRANTS THAT, TO THE BEST OF ITS KNOWLEDGE, IT HAS THE RIGHT TO ENTER INTO THIS AGREEMENT WITH COMPANY.

10. ASSIGNMENT.

This Agreement is personal to COMPANY and no rights or obligations may be assigned by COMPANY without the prior written consent of M.I.T. except as otherwise provided in this Agreement. Notwithstanding the foregoing, COMPANY may assign this Agreement in connection with the sale or transfer of all or substantially all of COMPANY's equity or assets, by merger, consolidation or otherwise provided that (a) the assignee shall agree in writing to be bound by the terms and conditions hereof prior to such assignment, and (b) the total valuation of the COMPANY is at least seven million five hundred thousand dollars (\$7,500,000). In the event that the total valuation of the COMPANY is less than \$7.5 million, then the assignment may be made only with M.I.T.'s written permission, such permission not to be unreasonably withheld. Failure of such assignee to agree to be bound by the terms and conditions of this Agreement shall be grounds for termination by M.I.T. under Section 13.3.

11. CONFIDENTIAL INFORMATION

- (a) <u>Designation</u>. CONFIDENTIAL INFORMATION that is disclosed in writing shall be marked with a legend indicating its confidential status (such as "Confidential" or "Proprietary"). CONFIDENTIAL INFORMATION that is disclosed orally or visually shall be considered confidential if designated as such prior to, during or immediately after disclosure.
- (b) Obligations. The Receiving Party shall (i) maintain such CONFIDENTIAL INFORMATION in strict confidence and shall not disclose, without the consent of the Disclosing Party, CONFIDENTIAL INFORMATION to third parties, except that the Receiving Party may disclose or permit the disclosure of any CONFIDENTIAL INFORMATION to its directors, officers, employees, consultants, and advisors who are obligated to maintain the confidential nature of such CONFIDENTIAL INFORMATION and who need to know such Confidential Information for the purposes of this Agreement; (ii) use such CONFIDENTIAL INFORMATION solely for the purposes of this Agreement; and (iii) allow its trustees or directors, officers, employees, consultants, and advisors to reproduce the CONFIDENTIAL INFORMATION only to the extent necessary for the purposes of this Agreement, with all such reproductions being considered CONFIDENTIAL INFORMATION. The Receiving Party shall be responsible for any unauthorized disclosure or use of CONFIDENTIAL INFORMATION by its trustees or directors, officers, employees, consultants, and advisors. The obligations set forth in this Article 11 shall survive for a period of five (5) years after disclosure of any portion of CONFIDENTIAL INFORMATION, provided that, with respect to any portion of such CONFIDENTIAL INFORMATION remains a trade secret.
- (c) Exceptions. The obligations of the Receiving Party under Subsection 11(b) above shall not apply to the extent that the Receiving Party can demonstrate that certain CONFIDENTIAL INFORMATION (i) was in the public domain prior to the time of its disclosure under this Agreement; (ii) entered the public domain after the time of its disclosure under this Agreement through means other than an unauthorized disclosure resulting from an act or omission by the Receiving Party; (iii) was independently developed or discovered by the Receiving Party without use of the CONFIDENTIAL INFORMATION; (iv) is or was disclosed to the Receiving Party at any time, whether prior to or after the time of its disclosure under this Agreement, by a third party having no fiduciary relationship with the Disclosing Party and having no obligation of confidentiality with respect to such CONFIDENTIAL INFORMATION; or (v) is required to be disclosed to comply with applicable laws or regulations, or with a court or administrative order, provided that the Disclosing Party receives reasonable prior written notice of such disclosure.

(d) Ownership and Return. The Receiving Party acknowledges that the Disclosing Party (or any third party entrusting its own information to the Disclosing Party) claims ownership of its CONFIDENTIAL INFORMATION in the possession of the Receiving Party. Upon the expiration or termination of this Agreement, and at the request of the Disclosing Party, the Receiving Party shall return to the Disclosing Party all originals, copies, and summaries of documents, materials, and other tangible manifestations of CONFIDENTIAL INFORMATION in the possession or control of the Receiving Party, except that the Receiving Party may retain one copy of the CONFIDENTIAL INFORMATION in the possession of its legal counsel solely for the purpose of monitoring its obligations under this Agreement.

12. GENERAL COMPLIANCE WITH LAWS

- 12.1 <u>Compliance with Laws</u>. COMPANY shall use reasonable commercial efforts to comply with all commercially material local, state, federal, and international laws and regulations relating to the development, manufacture, use, and sale of LICENSED PRODUCTS and LICENSED PROCESSES.
- 12.2 Export Control. COMPANY and its AFFILIATES and SUBLICENSEES shall comply with all United States laws and regulations controlling the export of certain commodities and technical data, including without limitation all Export Administration Regulations of the United States Department of Commerce. Among other things, these laws and regulations prohibit or require a license for the export of certain types of commodities and technical data to specified countries. COMPANY hereby gives written assurance that it will comply with, and will cause its AFFILIATES and SUBLICENSEES to comply with, all United States export control laws and regulations, that it bears sole responsibility for any violation of such laws and regulations by itself or its AFFILIATES or SUBLICENSEES, and that it will indemnify, defend, and hold M.I.T and Versitech harmless (in accordance with Section 8.1) for the consequences of any such violation.

- 12.3 Non-Use of M.I.T. Name. COMPANY and its AFFILIATES and SUBLICENSEES shall not use the name of "Massachusetts Institute of Technology," "Versitech," "The University of Hong Kong" or any variation, adaptation, or abbreviation thereof, or of any of their trustees, officers, faculty, students, employees, or agents, or any trademark or logo owned by M.I.T., "Versitech," or "The University of Hong Kong" or any terms of this Agreement in any promotional material or other public announcement or disclosure without the prior written consent of M.I.T. The foregoing notwithstanding, without the consent of M.I.T., COMPANY may make factual statements during the term of this Agreement that COMPANY has a license from M.I.T. under one or more of the patents and/or patent applications comprising the PATENT RIGHTS, , and may make disclosures or statements required by law or regulation.
- 12.4 <u>Marking of LICENSED PRODUCTS</u>. To the extent commercially feasible and consistent with prevailing business practices, COMPANY shall mark, and shall cause its AFFILIATES and SUBLICENSEES to mark, all LICENSED PRODUCTS that are manufactured or sold under this Agreement with the number of each issued patent under the PATENT RIGHTS that applies to such LICENSED PRODUCT.

13. TERMINATION

- 13.1 <u>Voluntary Termination by COMPANY</u>. COMPANY shall have the right to terminate this Agreement, for any reason, (i) upon at least six (6) months prior written notice to M.I.T., such notice to state the date at least six (6) months in the future upon which termination is to be effective, and (ii) upon payment of all amounts due to M.I.T. through such termination effective date.
- 13.2 <u>Cessation of Business</u>. If COMPANY ceases to carry on its business related to this Agreement M.I.T. shall have the right to terminate this Agreement immediately upon written notice to COMPANY.
 - 13.3 Termination for Default.
- (a) Nonpayment. In the event COMPANY fails to pay any amounts due and payable to M.I.T. hereunder, and fails to make such payments within thirty (30) days after receiving written notice of such failure, M.I.T. may terminate this Agreement immediately upon written notice to COMPANY.

- (b) <u>Material Breach</u>. In the event COMPANY commits a material breach of its obligations under this Agreement, except for breach as described in Section 12.3(a), and fails to cure that breach within sixty (60) days after receiving written notice thereof, M.I.T. may terminate this Agreement immediately upon written notice to COMPANY.
 - 13.4 Termination as a Consequence of Patent Challenge.
- (a) <u>By COMPANY</u>. If COMPANY or any of its AFFILIATES brings a PATENT CHALLENGE against M.I.T., or assists others in bringing a PATENT CHALLENGE against M.I.T. (except as required under a court order or subpoena), then M.I.T. may immediately terminate this Agreement and/or the license granted hereunder.
- (b) <u>By SUBLICENSEE</u>. If a SUBLICENSEE brings a PATENT CHALLENGE or assists another party in bringing a PATENT CHALLENGE (except as required under a court order or subpoena), then M.I.T. may send a written demand to COMPANY to terminate such sublicense. If COMPANY fails to so terminate such sublicense within thirty (30) days after M.I.T.'s demand, M.I.T. may immediately terminate this Agreement and/or the license granted hereunder.
 - 13.5 Effect of Termination.
- (a) <u>Survival</u>. The following provisions shall survive the expiration or termination of this Agreement: Articles 1, 8, 9, 11, 14 and 15, and Sections 4.1(i), 5.2 (obligation to provide final report and payment), 5.4, 12.1, 12.2 and 13.5.
- (b) <u>Inventory</u>. Upon the early termination of this Agreement, COMPANY and its AFFILIATES and SUBLICENSEES may complete and sell any work-in-progress and inventory of LICENSED PRODUCTS that exist as of the effective date of termination, provided that (i) COMPANY pays M.I.T. the applicable running royalty or other amounts due on such sales of LICENSED PRODUCTS in accordance with the terms and conditions of this Agreement, and (ii) COMPANY and its AFFILIATES and SUBLICENSEES shall complete and sell all work-in-progress and inventory of LICENSED PRODUCTS within six (6) months after the effective date of termination.
- (c) <u>Pre-termination Obligations</u>. In no event shall termination of this Agreement release COMPANY, AFFILIATES, or SUBLICENSEES from the obligation to pay any amounts that became due on or before the effective date of termination.

14. DISPUTE RESOLUTION.

- 14.1 <u>Mandatory Procedures</u>. The parties agree that any dispute arising out of or relating to this Agreement shall be resolved solely by means of the procedures set forth in this Article, and that such procedures constitute legally binding obligations that are an essential provision of this Agreement. If either party fails to observe the procedures of this Article, as may be modified by their written agreement, the other party may bring an action for specific performance of these procedures in any court of competent jurisdiction.
- 14.2 <u>Equitable Remedies</u>. Although the procedures specified in this Article are the sole and exclusive procedures for the resolution of disputes arising out of or relating to this Agreement, either party may seek a preliminary injunction or other provisional equitable relief if, in its reasonable judgment, such action is necessary to avoid irreparable harm to itself or to preserve its rights under this Agreement.
 - 14.3 Dispute Resolution Procedures.
- (a) <u>Mediation</u>. In the event any dispute arising out of or relating to this Agreement remains unresolved within sixty (60) days from the date the affected party informed the other party of such dispute, either party may initiate mediation upon written notice to the other party ("Notice Date"), whereupon both parties shall be obligated to engage in a mediation proceeding under the then current Center for Public Resources ("CPR") Model Procedure for Mediation of Business Disputes (http://www.cpradr.org), except that specific provisions of this Article shall override inconsistent provisions of the CPR Model Procedure. The mediator will be selected from the CPR Panels of Neutrals. If the parties cannot agree upon the selection of a mediator within fifteen (15) business days after the Notice Date, then upon the request of either party, the CPR shall appoint the mediator. The parties shall attempt to resolve the dispute through mediation until the first of the following occurs: (i) the parties reach a written settlement; (ii) the mediator notifies the parties in writing that they have reached an impasse; (iii) the parties agree in writing that they have reached an impasse; or (iv) the parties have not reached a settlement within sixty (60) days after the Notice Date.
- (b) <u>Trial Without Jury</u>. If the parties fail to resolve the dispute through mediation, or if neither party elects to initiate mediation, each party shall have the right to pursue any other remedies legally available to resolve the dispute, provided, however, that the parties expressly waive any right to a jury trial in any legal proceeding under this Article.

- 14.4 <u>Performance to Continue</u>. Each party shall continue to perform its undisputed obligations under this Agreement pending final resolution of any dispute arising out of or relating to this Agreement; provided, however, that a party may suspend performance of its undisputed obligations during any period in which the other party fails or refuses to perform its undisputed obligations. Nothing in this Article is intended to relieve COMPANY from its obligation to make undisputed payments pursuant to Articles 4 and 6 of this Agreement.
- 14.5 <u>Statute of Limitations</u>. The parties agree that all applicable statutes of limitation and time-based defenses (such as estoppel and laches) shall be tolled while the procedures set forth in Sections 14.3(a) are pending. The parties shall cooperate in taking any actions necessary to achieve this result.

15. MISCELLANEOUS.

15.1 <u>Notice</u>. Any notices required or permitted under this Agreement shall be in writing, shall specifically refer to this Agreement, and shall be sent by hand, recognized national overnight courier, confirmed facsimile transmission, confirmed electronic mail, or registered or certified mail, postage prepaid, return receipt requested, to the following addresses or facsimile numbers of the parties:

If to M.I.T., all matters relating to the license:

Massachusetts Institute of Technology Technology Licensing Office, Rms NE25-230 and NE18-501 One Cambridge Center, Kendall Square Cambridge, MA 02142-1601 Attention: Director

Tel: 617-253-6966 Fax: 617-258-6790

If to M.I.T., relating to any equity action after the initial issuance of shares:

Massachusetts Institute of Technology Treasurer's Office 238 Main Street Cambridge, MA 02142 Attention: Phillips B. Moore

Tel: 617-253-5422 Fax: 617-258-6676

If to COMPANY:	Arch Therapeutics, Inc
	1 Chieftain Lane
	Natick, MA 01760
	Attention: Terrence W. Norchi, President and CEC
	Tel:
	Fax:

All notices under this Agreement shall be deemed effective upon receipt. A party may change its contact information immediately upon written notice to the other party in the manner provided in this Section.

- 15.2 Governing Law. This Agreement and all disputes arising out of or related to this Agreement, or the performance, enforcement, breach or termination hereof, and any remedies relating thereto, shall be construed, governed, interpreted and applied in accordance with the laws of the Commonwealth of Massachusetts, U.S.A., without regard to conflict of laws principles, except that questions affecting the construction and effect of any patent shall be determined by the law of the country in which the patent shall have been granted. The state and federal courts having jurisdiction over Cambridge, MA, USA, provide the exclusive forum for any PATENT CHALLENGE and/or any court action between the parties relating to this Agreement. COMPANY submits to the jurisdiction of such courts and waives any claim that such court lacks jurisdiction over COMPANY or its AFFILIATES or constitutes an inconvenient or improper forum.
- 15.3 <u>Force Majeure</u>. Neither party will be responsible for delays resulting from causes beyond the reasonable control of such party, including without limitation fire, explosion, flood, war, strike, or riot, provided that the nonperforming party uses commercially reasonable efforts to avoid or remove such causes of nonperformance and continues performance under this Agreement with reasonable dispatch whenever such causes are removed.
- 15.4 <u>Amendment and Waiver</u>. This Agreement may be amended, supplemented, or otherwise modified only by means of a written instrument signed by both parties. Any waiver of any rights or failure to act in a specific instance shall relate only to such instance and shall not be construed as an agreement to waive any rights or fail to act in any other instance, whether or not similar.

- 15.5 Severability. In the event that any provision of this Agreement shall be held invalid or unenforceable for any reason, such invalidity or unenforceability shall not affect any other provision of this Agreement, and the parties shall negotiate in good faith to modify the Agreement to preserve (to the extent possible) their original intent. If the parties fail to reach a modified agreement within thirty (30) days after the relevant provision is held invalid or unenforceable, then the dispute shall be resolved in accordance with the procedures set forth in Article 14. While the dispute is pending resolution, this Agreement shall be construed as if such provision were deleted by agreement of the parties.
- 15.6 <u>Binding Effect</u>. This Agreement shall be binding upon and inure to the benefit of the parties and their respective permitted successors and assigns.
 - 15.7 Headings. All headings are for convenience only and shall not affect the meaning of any provision of this Agreement.
- 15.8 Entire Agreement. This Agreement constitutes the entire agreement between the parties with respect to its subject matter and supersedes all prior agreements or understandings between the parties relating to its subject matter.

IN WITNESS WHEREOF, the parties have caused this Agreement to be executed by their duly authorized representatives.

The EFFECTIVE DATE of this Agreement is May 23, 2011.

MASSACHUSETTS INSTITUTE OF TECHNOLOGY

ARCH THERAPEUTICS, INC.

By: /s/ Lita Nelson

Name: Lita Nelson

Title: Director, Technology Licensing Office

By: /s/ Terrence W. Norchi, M.D

Name: Terrence_W. Norchi, MD

Title: President and CEO

APPENDIX A List of Patent Applications and Patents

I. United States Patents and Applications

M.I.T. Case No. 6692

United States of America Patent No. 5955343, Issued September 21, 1999

United States of America Patent No. 6800481, Issued October 5, 2004

United States of America Patent No. 7098028, Issued August 29, 2006

United States of America Serial No. 11/512753, Filed August 29, 2006

"Stable Macroscopic Membranes Formed By Self-Assembly Of Amphiphilic Peptides And Uses Therefor"

by C. Michael Dipersio, Todd Holmes, Curtis Lockshin, Alexander Rich and Shuguang Zhang

M.I.T. Case No. 8813

United States of America Serial No. 09/778200, Filed February 6, 2001 "Peptide Scaffold Encapsulation Of Tissue Cells And Uses Therof" by Alan J. Grodzinsky, John Kisiday and Shuguang Zhang

M.I.T. Case No. 9344

United States of America Serial No. 10/193942, Filed July 15, 2002
"Cellular Reprogramming In Peptide Hydrogel And Uses Thereof"
By Carlos E. Semino, Colette Shen, James L. Sherley and Shuguang Zhang

M.I.T. Case No. 9786

United States of America Serial No. 10/968790, Filed October 18, 2004
"Self-Assembling Peptides For Regeneration And Repair Of Neural Tissue"
by Rutledge Ellis-Behnke, Gerald E. Schneider, Carlos E. Semino and Shuguang Zhang

M.I.T. Case No. 10154

United States of America Serial No. 10/877068, Filed June 25, 2004 "Self-Assembling Peptides Incorporating Modifications And Methods Of Use Thereof" by Elsa Genove, Carlos E. Semino and Shuguang Zhang

II. International (non-U.S.) Patents and Applications

M.I.T. Case No. 8813

Australia Serial No. 2002-248406, Filed February 6, 2002 Canada Serial No. 2344954, Filed April 25, 2001 European Patent Convention Serial No. 02717398.8, Filed February 6, 2002 Japan Serial No. 2002-563298, Filed February 6, 2002 Patent Cooperation Treaty Serial No. US02/03482, Filed February 6, 2002 "Peptide Scaffold Encapsulation Of Tissue Cells And Uses Thereof" by Alan J. Grodzinsky, John Kisiday and Shuguang Zhang

M.I.T. Case No. 9344

Patent Cooperation Treaty Serial No. US02/03607, Filed February 6, 2002 "Liver Cellular Reprogramming In Peptide Hydrogel And Uses Thereof" by Carlos E. Semino, Colette Shen, James L. Sherley and Shuguang Zhang

M.I.T. Case No. 10154

Canada Serial No. 2530482, Filed June 25, 2004 European Patent Convention Serial No. 04785965.7, Filed June 25, 2004 Japan Serial No. 2006-517694, Filed June 25, 2004 Patent Cooperation Treaty Serial No. US04/020549, Filed June 25, 2004 "Self-Assembling Peptides Incorporating Modifications And Methods Of Use Thereof" by Elsa Genove, Carlos E. Semino and Shuguang Zhang

APPENDIX B List of Patent Applications and Patents

I. United States Patents and Applications

M.I.T. Case No. 11366

United States of America Serial No. 60/758827, Filed January 13, 2006

"Compositions And Methods For Inhibiting Movement Or Leakage Of A Body Substance And Promoting Tissue Repair"

United States of America Serial No. 11/411745, Filed April 25, 2006

"Compositions And Methods For Promoting Hemostasis And Other Physiological Activities"

by Rutledge Ellis-Behnke, Yu Xiang Liang, Gerald E. Schneider, Kwok-Fai So, David K C. Tay and Shuguang Zhang

M.I.T. Case No. 12061

United States of America Serial No. 11/796132, Filed April 25, 2007

"Compositions And Methods For Affecting Movement Of Contaminants, Bodily Fluids Or Other Entities, And/Or Affecting Other Physiological Conditions"

United States of America Serial No. 60/745601, Filed April 25, 2006

"Compositions And Methods For Promoting Hemostasis And Other Physiological Activities"

by Rutledge Ellis-Behnke, Yu Xiang Liang, Gerald E. Schneider, Kwok-Fai So and David K C. Tay

II. International (non-U.S.) Patents and Applications

M.I.T. Case No. 11366

Patent Cooperation Treaty Serial No. US06/015850, Filed April 25, 2006

"Compositions And Methods For Promoting Hemostasis And Other Physiological Activities"

by Rutledge Ellis-Behnke, Yu Xiang Liang, Gerald E. Schneider, Kwok-Fai So, David K C. Tay and Shuguang Zhang

M.I.T. Case No. 12061

Patent Cooperation Treaty Serial No. US07/010041, Filed April 25, 2007

"Compositions And Methods For Promoting Hemostasis And And Other Physiological Activites"

by Rutledge Ellis-Behnke, Yu Xiang Liang, Gerald E. Schneider, Kwok-Fai So and David K C. Tay

APPENDIX C

<u>List of Countries (excluding United States) for which</u> PATENT RIGHTS B . Applications Will Be Filed, Prosecuted and Maintained

MIT case No. 11366

Australia Canada China Europe

India Israel

Japan

Korea

APPENDIX D

M.I.T.-Versitech Joint Invention Agreement

EXHIBIT A

CONFLICT AVOIDANCE STATEMENT

Name: Rutledge Ellis-Behnke, PhD

Dept. or Lab.: <u>Brain and Cognitive Sciences</u>

Company: <u>Clear Nano Solutions</u>

Address: <u>355 Summer Street</u>

Manchester, MA 01944

Licensed Technology: Cases: See Appendix A and B

Because of the M.I.T. license granted to the above company and my equity* position and continuing relationship with this company, I acknowledge the potential for a possible conflict of interest between the performance of research at M.I.T. and my contractual or other obligations to this company. Therefore, I will not:

- 1) use students at M.I.T. for research and development projects for the company;
- 2) restrict or delay access to information from my M.I.T. research;
- 3) take direct or indirect research support from the company in order to support my activities at M.I.T.; or
- 4) employ students at the company, except in accordance with Section 4.5.2, "Faculty and Students," in the Policies and Procedures Guide.

In addition, in order to avoid the appearance of a conflict, I will attempt to differentiate clearly between the intellectual directions of my M.I.T. research and my contributions to the company. To that end, I will expressly inform my department head/laboratory director annually of the general nature of my activities on behalf of the company.

Signed: /s/ Rutledge Ellis-Behnke, PhD

Date: 1/22/08

Approved by: /s/ Mriganka Sur

Name (print): Mriganka Sur (Dept. Head or Lab Dir)

* "Equity" includes stock, options, warrants or other financial instruments convertible into stock, which are directly or indirectly controlled by the inventor.

EXHIBIT B

INVENTOR/AUTHOR ACKNOWLEDGMENT OF NO EQUITY DISTRIBUTION

Form Version 8/22/01

In partial reliance on the undersigned's execution of this Acknowledgment, M.I.T. has entered into the license agreement to which this Acknowledgment is attached (the "LICENSE") in which COMPANY received certain licenses to the technology listed below, on some or all of which the undersigned is a listed inventor or author. The undersigned, independently of the LICENSE, has received or will soon acquire equity in ("COMPANY"), and, in accordance with M.I.T.'s licensing policies contained in M.I.T.'s *Guide to the Ownership, Distribution and Commercial Development of M.I.T. Technology*, as that policy may be amended from time to time (specifically §4.2.5 as of this Form Version date), the undersigned, on his/her own behalf and on behalf of his/her heirs and assigns, acknowledges and agrees that he/she has no right to receive any share of equity income received by M.I.T. in consideration for the LICENSE.

Technology Licensed as of the EFFECTIVE DATE of	of the LICENSE:	
		-
ppendix A & B		
ss:Juliet Ellis-Behnke:		Signed Rutledge Ellis-Behnke
ss. Juliet Ellis-Dellike.		Print Name: /s/ Rutledge Ellis-Behnke
		Date: 12/12/07
		12/12/07
	42	

First Amendment to the Amended and Restated Exclusive Patent License Agreement

This First Amendment, effective as of the date set forth above the signatures of the parties below, amends the Amended and Restated Exclusive Patent License Agreement dated May 23, 2011(the "LICENSE AGREEMENT") by and between the Massachusetts Institute of Technology, a Massachusetts corporation having its principal office at 77 Massachusetts Avenue, Cambridge, Massachusetts 02139 ("M.I.T.") and Arch Therapeutics, Inc. ("COMPANY"), a Massachusetts corporation having its principal office at 1 Chieftain Lane, Natick, MA 01760.

WHEREAS, M.I.T. and COMPANY wish to modify the provisions of the LICENSE AGREEMENT to account for certain unmet financial obligations by COMPANY;

NOW, THEREFORE, in consideration of the promises and mutual covenants contained herein, the parties hereto agree as follows:

- 1. Sections 3.1(e) and 3.1(k) of the LICENSE AGREEMENT are hereby deleted in their entirety and replaced with the following:
 - 3.1 Diligence Requirements.
- (e) (i) In addition to Section 3.1(d), COMPANY shall RAISE CAPITAL of at least Two Million dollars (\$2,000,000) solely from the sale of the COMPANY's equity securities for its own account by December 31, 2012.
- (ii) In addition to Section 3.1(e)(i), COMPANY shall RAISE CAPITAL of at least Three Million dollars (\$3,000,000) by July 1, 2014.
- (k) COMPANY shall conduct a pre-investigational meeting or comparable consultation with the U.S. Food and Drug Administration (or equivalent European regulatory authority) to review a clinical testing strategy for a LICENSED PRODUCT by January 1, 2014.
- 2. Sections 4.1(b) and 4.1(e) of the LICENSE AGREEMENT are hereby deleted in their entirety and replaced with the following:

4.1 Consideration for Grant of Rights.

(b) <u>License Maintenance Fees</u>. M.I.T. acknowledges that COMPANY previously paid M.I.T. a license maintenance fee in the amount of \$10,000 for the calendar year commencing on January 1, 2009 and in the amount of \$10,000 for the calendar year commencing on January 1, 2010. There shall be no license maintenance fee for the calendar year commencing on January 1, 2011 and January 1, 2012. COMPANY shall pay to M.I.T. the following license maintenance fees within 30 (thirty) days of the dates set forth below:

January 1, 2013	\$25,000
January 1, 2014	\$35,000
January 1, 2015	\$45,000
January 1, 2016 and each January 1 of every year thereafter	\$50,000

This annual license maintenance fee is nonrefundable; however, the license maintenance fee may be credited to running royalties subsequently due on NET SALES earned during the same calendar year, if any. License maintenance fees paid in excess of running royalties due in such calendar year shall not be creditable to amounts due for future years.

(e) <u>Milestone Payments</u>. COMPANY shall pay to M.I.T. the amounts set forth below upon the first achievement by COMPANY or any of its AFFILIATES or SUBLICENSEES of certain milestone events as described below.

Milestone Event	<u>Payment</u>
Initiation of human clinical studies, including evaluation of proof of concept, proof of mechanism and/or efficacy, of a LICENSED PRODUCT	Ten Thousand dollars (\$10,000)
Filing of a submission/application for regulatory/marketing approval of a LICENSED PRODUCT	Twenty Five Thousand dollars (\$25,000)
First COMMERCIAL SALE of a LICENSED PRODUCT	Fifty Thousand dollars (\$50,000)
Cumulative NET SALES of Ten Million dollars (\$10,000,000) in either or both FIELD A and FIELD B in the TERRITORY	One Hundred Thousand dollars (\$100,000)

COMPANY shall notify M.I.T. within ten (10) days of the achievement of any of the above milestones by COMPANY or any of its AFFILIATES or SUBLICENSEES. COMPANY shall make such non-refundable, non-creditable milestone payments within sixty (60) days after achievement of each of the milestones.

3. Section 6.3 of the LICENSE AGREEMENT is hereby deleted in its entirety and replaced with the following:

6.3 Payment of Expenses.

Payment of all reasonable out-of-pocket fees and costs, including attorneys' fees, relating to the filing, prosecution and maintenance of the PATENT RIGHTS shall be the responsibility of COMPANY and other commercial licensees of the PATENT RIGHTS as they exist from time to time (as used herein a "commercial licensee" shall mean a for-profit entity that has been granted a license under the PATENT RIGHTS to develop and sell products).

- (a) <u>PATENT RIGHTS A</u>. COMPANY shall be responsible for a pro-rata share of all patent related fees and costs relating to the filing, prosecution and maintenance of the PATENT RIGHTS A incurred after the EFFECTIVE DATE. As of the EFFECTIVE DATE, COMPANY's pro rata share for the PATENT RIGHTS A is twenty five percent (25%). Notwithstanding the foregoing, COMPANY shall not be required to reimburse M.I.T. for any such expenses associated with PATENT RIGHTS A incurred through February 28, 2011 (approximately \$9,456.28) until December 31, 2012.
- (b) <u>PATENT RIGHTS B.</u> Payment of all patent related fees and costs relating to the filing, prosecution and maintenance of the PATENT RIGHTS B shall be the responsibility of COMPANY, whether such amounts were incurred before or after the EFFECTIVE DATE. Notwithstanding the foregoing, COMPANY shall not be required to reimburse M.I.T. for any such expenses associated with PATENT RIGHTS B incurred through February 28, 2011 (approximately \$71,586.67) until December 31, 2012.
- (c) For clarity, COMPANY shall be responsible for ongoing payment of all ongoing patent expenses in accordance with this Section 6.3 as they are incurred for the period beginning March 1, 2011.
- (d) COMPANY shall reimburse all amounts due pursuant to this Section 6.3 within thirty (30) days of invoicing; late payments shall accrue interest pursuant to Section 4.2(c). In all instances, M.I.T. shall pay the fees prescribed for large entities to the United States Patent and Trademark Office.

4. For the period beginning April 1, 2012 and ending December 31, 2012, M.I.T. and COMPANY agree that COMPANY shall reimburse M.I.T. for all patent expenses associated with PATENT RIGHTS A and PATENT RIGHTS B incurred after February 28, 2011, pursuant to Section 6.3(c) of the LICENSE AGREEMENT (as amended herein), according to the following schedule:

April 1, 2012	\$1,000
May 1, 2012	\$1,000
June 1, 2012	\$2,000
July 1, 2012	\$2,000
August 1, 2012	\$3,000
September 1, 2012	\$3,000
October 1, 2012	\$4,000
November 1, 2012	\$4,000
December 1, 2012	\$5,000

This reimbursement schedule shall be effective only until December 31, 2012. Notwithstanding the foregoing: (i) at any time that COMPANY has RAISED CAPITAL from the sale of the COMPANY's equity securities for its own account of at least Two Hundred Fifty Thousand dollars (\$250,000), from that date forward, COMPANY shall reimburse M.I.T. at a rate of Five Thousand dollars (\$5,000) a month until December 31, 2012; and (ii) all outstanding patent expenses due in accordance with Section 6.3 shall be reimbursed in full by January 1, 2013.

- 5. Notwithstanding anything to the contrary in the LICENSE AGREEMENT, M.I.T. and COMPANY agree that in the event COMPANY fails to fulfill any of its obligations under either of (1) Sections 3.1(e)(i) and 6.3 of the LICENSE AGREEMENT, as amended by this First Amendment, or (2) Section 4 of this First Amendment, then M.I.T. may treat such failure as a material breach and shall have the right to immediately terminate the LICENSE AGREEMENT upon written notice to COMPANY, without an opportunity to cure.
- 6. Except as specifically modified or amended hereby, all other terms and conditions of the LICENSE AGREEMENT shall remain unchanged and in full force and effect. Capitalized terms used herein and not defined shall have the meanings set forth in the LICENSE AGREEMENT.

IN WITNESS WHEREOF, the parties have caused this First Amendment to be executed under seal by their duly authorized representatives.

The Effective Date of this First Amendment is May 15, 2012.	
MASSACHUSETTS INSTITUTE OF TECHNOLOGY	ARCH THERAPEUTICS, INC.
By: /s/ Lita Nelson	By: /s/ Terrence W. Norchi, M.D
Name:Lita Nelson	Name: Terrence W. Norchi, MD
Title: Director, Technology Licensing Office	Title: President and CEO
47	

SECOND AMENDMENT

This Second Amendment, effective as of the date set forth above the signatures of the parties below, amends the Amended and Restated Exclusive Patent License Agreement dated May 23, 2011, as amended by the First Amendment on May 15, 2012 (the "LICENSE AGREEMENT") between the Massachusetts Institute of Technology, a Massachusetts corporation having its principal office at 77 Massachusetts Avenue, Cambridge, Massachusetts, 02139 ("M.I.T.") and Arch Therapeutics, Inc., a Massachusetts corporation, with a principal place of business at 1 Chieftain Lane, Natick, MA 01760 ("COMPANY").

WHEREAS, M.I.T. and COMPANY wish to modify the provisions of the LICENSE AGREEMENT to account for certain unmet financial obligations and diligence requirements by COMPANY;

NOW, THEREFORE, in consideration of the promises and mutual covenants contained herein, the parties hereby agree as follows:

- 1. COMPANY currently owes M.I.T. approximately \$120,003 in patent costs incurred prior to December 31, 2012. For the period beginning February 28, 2013 and ending April 30, 2013, M.I.T. and COMPANY agree that COMPANY shall reimburse M.I.T. for these expenses according to the following schedule:
 - a. Payment of \$12,000, which is approximately equal to ten percent (10%) of the outstanding patent costs, by February 28, 2013.
 - b. Payment of \$24,000, which is approximately equal to twenty percent (20%) of the outstanding patent costs, by March 31, 2013.
- c. Payment of the remaining outstanding patent costs incurred prior to December 31, 2012, which approximately equals \$84,003 by April 30, 2013.
- 2. COMPANY currently owes M.I.T. the \$25,000 license maintenance fee due January 1, 2013, as set forth in Section 4.1(b) of the LICENSE AGREEMENT. M.I.T. and COMPANY agree that COMPANY shall pay such fee by April 30, 2013.

- 3. Effective January 1, 2013, COMPANY shall provide M.I.T. with advance payment of all newly incurred patent costs, as evidenced by M.I.T.'s receipt of such amounts either (a) at least ten (10) business days after COMPANY has requested that M.I.T. proceed with the patent action, or (b) at least fifteen (15) business days prior to the expected date of filing of the relevant material with the applicable patent office, as shall be determined at M.I.T.'s sole discretion. Such practice shall continue until (i) all outstanding costs due under the LICENSE AGREEMENT have been fully paid, as set forth in Sections 1 and 2 above, and (ii) COMPANY has been fully compliant with the advance payment of patent expenses in accordance with this Section 2 for a period of six (6) months after the effective date of this Second Amendment.
 - 4. Section 6.3(d) of the LICENSE AGREEMENT is hereby deleted in its entirety and replaced with the following:
- (d) COMPANY shall reimburse all amounts due pursuant to this Section 6.3 within thirty (30) days of invoicing; late payments shall accrue interest pursuant to Section 4.3(c). In all instances, M.I.T. shall pay the fees prescribed for large entities to the United States Patent and Trademark Office.
 - 5. Section 3.1(e) of the LICENSE AGREEMENT is hereby deleted in its entirety and replaced with the following:
- (e) (i) Notwithstanding Section 3.1(d) and any funds raised prior to May 15, 2012, COMPANY shall raise additional capital of at least Two Million Dollars (\$2,000,000) of cash proceeds by April 30, 2013, provided that at least forty percent (40%) of such cash proceeds are able to be used for general and administrative costs.
- (e) (ii) Notwithstanding Section 3.1(d) and any amounts raised prior to May 15, 2012, COMPANY shall raise additional capital of at least Five Million Dollars (\$5,000,000) of cash proceeds by September 30, 2014, provided that at least forty percent (40%) of such cash proceeds are able to be used for general and administrative costs.

For clarity, the parties acknowledge and agree that the funding amount indicated in the diligence requirement set forth in Section 3.1(e)(i) is not intended to reflect aggregate capital raised; for example, the Two Million Dollars that shall be raised by April 30, 2013 pursuant to Section 3.1(e)(i) shall not include any amounts that were raised prior to May 15, 2012.

- 6. During the month of February 2013, COMPANY shall introduce M.I.T. to potential investors in COMPANY in connection with the fundraising described in Section 3.1(e)(i) of the LICENSE AGREEMENT, as amended herein, and shall facilitate scheduling meetings between M.I.T. and such investors, as requested by M.I.T.
- 7. Pursuant to Section 4.1(i), 46,700 shares of Common Stock of COMPANY were due to M.I.T. on the Effective Date of the LICENSE AGREEMENT. COMPANY shall issue these shares, as well as any additional equity due to M.I.T., by March 31, 2013.
- 8. Failure of COMPANY to meet any of the obligations set forth in this Second Amendment will result in termination of the LICENSE AGREEMENT, effective immediately, with no opportunity to cure.
- 9. Except as specifically modified or amended hereby, all other terms and conditions of the LICENSE AGREEMENT shall remain unchanged and in full force and effect. Capitalized terms used herein and not defined shall have the meanings set forth in the LICENSE AGREEMENT.

IN WITNESS WHEREOF, the parties have caused this Second Amendment to be executed under seal by their duly authorized representatives.

The Effective Date of this Second Amendment is February 1, 2013.

MASSACHUSETTS INSTITUTE OF TECHNOLOGY	ARCH THERAPEUTICS, INC.
By: /s/ Lita Nelson	By: /s/ Terrence W. Norchi, M.D
Name:Lita Nelson	Name: Terrence W. Norchi, MD
Title: Director, Technology Licensing Office	Title: President and CEO
5	50

THIRD AMENDMENT

This Third Amendment, effective as of the date set forth above the signatures of the parties below, amends the Amended and Restated Exclusive Patent License Agreement dated May 23, 2011, as amended by the First Amendment dated May 15, 2012 and the Second Amendment dated February 1, 2013 (the "LICENSE AGREEMENT") between the Massachusetts Institute of Technology, a Massachusetts corporation having its principal office at 77 Massachusetts Avenue, Cambridge, Massachusetts, 02139 ("M.I.T.") and Arch Therapeutics, Inc., a Massachusetts corporation, with a principal place of business at 1 Chieftain Lane, Natick MA 01760 ("COMPANY").

WHEREAS, COMPANY has represented to M.I.T. that COMPANY has made substantial progress towards raising capital in order to meet certain obligations set forth in the LICENSE AGREEMENT, including the financial obligations and diligence requirements set forth in the Second Amendment to the LICENSE AGREEMENT;

NOW, THEREFORE, in consideration of the promises and mutual covenants contained herein, the parties hereby agree as follows:

- 1. COMPANY represents and warrants to M.I.T. that:
 - a. In December 2012, COMPANY was awarded a loan from the Massachusetts Life Sciences Center (the "Center") Accelerator Loan Program in the amount of One Million dollars (\$1,000,000) (the "MLSC FUNDS"), receivable by COMPANY, subject to the following conditions:
 - i. COMPANY must raise an additional One Million Five Hundred Thousand dollars (\$1,500,000) of new debt or equity by April 30, 2013, not including any debt or equity existing prior to December 1, 2012; and
 - ii. An additional round of diligence satisfactory to the Center in its sole discretion.
 - b. COMPANY has Raised One Million Five Hundred Thousand dollars (\$1,500,000) of cash proceeds from debt or equity, inclusive of convertible debt, (the "INVESTOR FUNDS") between December 1, 2012 and April 30, 2013 which funds are in COMPANY's own accounts.

- c. At least forty percent (40%) of the total of the MLSC FUNDS and the INVESTOR FUNDS are able to be used for general and administrative costs.
- d. COMPANY has signed a binding letter of intent with an investor ("COLDSTREAM"), who shall invest One Million Two Hundred Thousand dollars (\$1,200,000) by April 30, 2013 followed by an additional Eight Hundred Thousand dollars (\$800,000) by approximately April 30, 2014. COLDSTREAM has funded \$1,200,000 to COMPANY as of April 29, 2013 which funds are included in the INVESTOR FUNDS described in Section 1.b. above.
- 2. In view of COMPANY's representations set forth in Section 1 above, M.I.T. hereby agrees to extend the April 30, 2013 deadline set forth in Section 3.1(e)(i) of the LICENSE AGREEMENT until June 30, 2013 to allow COMPANY to execute the Massachusetts Life Sciences Center Life Sciences Accelerator Funding Agreement.
- 3. Concurrent with the signing of this Third Amendment, on April 30, 2013 COMPANY shall pay to M.I.T. the following amount:
 - a. The \$25,000 license maintenance fee (and associated interest of \$109.38) due January 1, 2013, as set forth in Section 4.1(b) of the LICENSE AGREEMENT.
- 4. On or before May 6, 2013, COMPANY shall pay to M.I.T. the following amounts:
 - a. Outstanding patent costs (and associated interest) incurred prior to December 31, 2012 in the amount of \$84,303.22.
 - b. Outstanding patent costs incurred as of December 31, 2012 (and associated interest) in the amount of \$29,879.71.
 - c. Pre-payment of patent expenses associated with South Korean patent application number 10-2010-7019627 in the amount of \$12,000.
- 5. COMPANY acknowledges and agrees that it shall continue to comply with the advanced payment of patent costs as described in Section 3 of the Second Amendment to the LICENSE AGREEMENT.
- 6. Pursuant to Section 4.1(i), COMPANY shall issue any additional shares of COMPANY's Common Stock owed to M.I.T. by June 30, 2013.

- 7. Failure of COMPANY to meet any of the obligations set forth in this Third Amendment will result in termination of the LICENSE AGREEMENT, effective immediately, with no opportunity to cure.
- 8. Except as specifically modified or amended hereby, all other terms and conditions of the LICENESE AGREEMENT shall remain unchanged and in full force and effect. Capitalized terms used herein and not defined shall have the meanings set forth in the LICENSE AGREEMENT.

IN WITNESS WHEREOF, the parties have caused this Third Amendment to be executed under seal by their duly authorized representatives.

The Effective Date of this Third Amendment is April 30, 2013.

MASSACHUSETTS INSTITUTE OF TECHNOLOGY	ARCH THERAPEUTICS, INC.
By: /s/ John H. Turner, Jr.	By: /s/ Terrence W. Norchi, M.D
Name: John H. Turner, Jr.	Name: Terrence W. Norchi, MD
Title: Associate Director, Technology Licensing Office	Title: President and CEO

53

LETTER AGREEMENT

This Letter Agreement, effective as of the date set forth above the signatures of the parties below, confirms the understanding between the Massachusetts Institute of Technology, a Massachusetts corporation having its principal office at 77 Massachusetts Avenue, Cambridge, Massachusetts, 02139 ("M.I.T.") and Arch Therapeutics, Inc., a Massachusetts corporation, with a principal place of business at 1 Chieftain Lane, Natick MA 01760 ("COMPANY"), with respect to the Amended and Restated Exclusive Patent License Agreement between M.I.T. and COMPANY dated May 23, 2011, as amended by the First Amendment dated May 15, 2012, the Second Amendment dated February 1, 2013 and the Third Amendment dated April 30, 2013 (the "LICENSE AGREEMENT"). Capitalized terms used herein and not defined shall have the meanings set forth in the LICENSE AGREEMENT.

WHEREAS, COMPANY has represented to M.I.T. that COMPANY plans to undergo a reverse triangular merger with Arch Therapeutics, Inc., a Nevada corporation (formerly known as Almah, Inc.) ("Parent"), which is a publicly traded "shell company," as that term is defined in Rule 12b-2 under the Securities Exchange Act of 1934, as amended (the "Merger");

NOW, THEREFORE, in consideration of the promises and mutual covenants contained herein, the parties hereby agree as follows:

- 1. COMPANY represents and warrants to M.I.T. that the current and pro forma capitalization table attached as Exhibit 1 hereto accurately reflects issuances included in the Funding Threshold through the date hereof and is true and correct on a Fully Diluted Basis as of the date hereof (with respect to the current capitalization table) and as of immediately following the closing of the Merger (with respect to the pro forma capitalization table).
- 2. COMPANY and M.I.T. hereby agree that, subject to and upon the closing of the Merger, M.I.T. shall be entitled to receive, and COMPANY shall cause to be issued to M.I.T., an aggregate number of shares equal to Four Percent (4%), or Two Million Seven Hundred and Twenty Thousand (2,720,000) shares, of the common stock of Parent on a Fully Diluted Basis as of immediately following the closing of the Merger.
- 3. COMPANY shall deliver the shares due under Section 2 above to M.I.T. within fifteen (15) days of the closing of the Merger.
- 4. Upon the delivery of the shares to M.I.T. pursuant to Section 3 above, M.I.T. agrees that COMPANY has fulfilled in entirety the obligations under Section 4.1(i) of the LICENSE AGREEMENT, and that COMPANY has no further obligations to M.I.T. thereunder.
- 5. As of the Effective Date of this Letter Agreement, M.I.T. acknowledges that COMPANY has complied with Sections 1a, 1b, 3, 6 and 7 of the Second Amendment to the LICENSE AGREEMENT, and Sections 3, 4 and 5 of the Third Amendment to the LICENSE AGREEMENT.

- 6. As of the Effective Date of this Letter Agreement, solely with respect to immediate termination with no opportunity to cure in the event of COMPANY failing to meet the obligations set forth in the Second Amendment to the LICENSE AGREEMENT and the Third Amendment to the LICENSE AGREEMENT, respectively, M.I.T. agrees that Section 8 of the Second Amendment to the LICENSE AGREEMENT and Section 7 of the Third Amendment to the LICENSE AGREEMENT, respectively, shall no longer apply. The Parties acknowledge and agree that any issues regarding termination, including for default, shall be governed as set forth in the relevant provisions of the LICENSE AGREEMENT.
- 7. M.I.T. hereby agrees (a) to attend, in person or by proxy, the special meeting of the shareholders of Arch to be held on or about June 14, 2013 for the purpose of voting on the Merger (the "Meeting"), and (b) to vote in favor of the Merger at the Meeting.
- 8. No amendment or modification of or supplement to the terms of this Letter Agreement shall be binding on a party unless reduced to writing and signed by both parties. Any waiver of any rights or failure to act in a specific instance shall not operate or be construed as an agreement to waive any rights or fail to act in any other instance, whether or not similar.
- 9. This Letter Agreement, together with the LICENSE AGREEMENT, sets forth the entire agreement among the Parties as to the subject matter hereof and supersedes all prior and contemporaneous agreements, understandings, negotiations and discussions, whether oral or written, between the Parties as to the subject matter hereof. This Letter Agreement and all disputes arising out of or related to this Letter Agreement, or the performance, enforcement, breach or termination hereof, and any remedies relating thereto, shall be construed, governed, interpreted and applied in accordance with the laws of the Commonwealth of Massachusetts, U.S.A., without regard to conflict of laws principles. Except as expressly set forth in this Letter Agreement, all of the terms and provisions of the LICENSE AGREEMENT shall remain unchanged and unmodified, and the LICENSE AGREEMENT as amended hereby shall remain in full force and effect and be read together and construed with this Letter Agreement.

{Signature Page Follows}

IN WITNESS WHEREOF, the parties have caused this Letter Agreement to t	be executed under seal by their duly authorized representatives.
The Effective Date of this Letter Agreement is June 10, 2013.	
MASSACHUSETTS INSTITUTE OF TECHNOLOGY	ARCH THERAPEUTICS, INC.
By: /s/ Lita Nelson	By: /s/ Terrence W. Norchi, M.D
Name:Lita Nelson	Name: Terrence W. Norchi, MD
Title: Director, Technology Licensing Office	Title: President and CEO

TERMINATION AGREEMENT AND RELEASE

This **TERMINATION AGREEMENT AND RELEASE** (this "**Agreement**"), dated June 25, 2013, is entered into by and between Arch Therapeutics, Inc., a Massachusetts corporation (the "**Company**"), and Dr. Terrence W. Norchi ("**T. Norchi**"), with respect to the following facts:

- A. The Company and T. Norchi are parties to that certain letter agreement dated November 1, 2010 setting forth the terms of T. Norchi's compensation for service as an employee of the Company (the "**Employment Agreement**").
- B. The Company has issued to T. Norchi that certain promissory note dated January 21, 2010 (the "**Promissory Note**"), pursuant to which the Company is obligated to pay to T. Norchi an aggregate principal amount of \$275,200 plus accrued interest thereon at the agreed rate of 6% per annum through December 31, 2009 and 10% per annum thereafter, as set forth in the Promissory Note.
- C. The Company is a party to that certain Agreement and Plan of Merger dated May 10, 2013 by and among Arch Therapeutics, Inc., a Nevada corporation ("Parent"), Arch Acquisition Corporation, a wholly owned subsidiary of Parent, and the Company, pursuant to which Arch Acquisition Corporation will merge with and into the Company and the Company will become a wholly owned subsidiary of Parent (the "Merger").
- D. The Company and T. Norchi now desire to terminate in full the Employment Agreement, effective as of the closing of the Merger (the "Closing"), and the Promissory Note effective as of the date of the repayment of amounts owed thereunder, all pursuant to the terms and conditions set forth herein.

NOW THEREFORE, in consideration of the foregoing premises and the mutual covenants set forth below, the parties hereby agree as follows:

1. TERMINATION.

- 1.1 Employment Agreement. Effective as of and conditioned upon the Closing, each of the Company and T. Norchi terminate the Employment Agreement by mutual agreement. Effective as of such termination, all rights and obligations of each of the Company and T. Norchi under the Employment Agreement shall be deemed satisfied in full and each such party shall have no further obligations thereunder. Without limiting the generality of the foregoing, T. Norchi hereby waives in full any rights he may have, have had or be deemed to have under the Employment Agreement with respect to cash, equity or other compensation owed by the Company. Notwithstanding anything to the contrary in the Employment Agreement, no terms or conditions of the Employment Agreement shall survive the termination thereof set forth in this Section 1.1.
- 1.2 Promissory Note. The Company and T. Norchi hereby acknowledge and agree that, as of June 24, 2013, the Company has paid in full all amounts owed under the Promissory Note and, effective as of such repayment, the Promissory Note is terminated in all respects, and all of the Company's obligations under the Promissory Note have been satisfied in full and the Company has no further obligations thereunder. Without limiting the generality of the foregoing, T. Norchi hereby waives in full any rights he may have, have had or be deemed to have under the Promissory Note with respect to any equity or rights to receive equity thereunder. Notwithstanding anything to the contrary in the Promissory, no terms or conditions of the Promissory Note shall survive the termination thereof set forth in this Section 1.2.

2. RELEASE.

- Release. T. Norchi, on behalf of himself and his successors and assigns, hereby forever and fully, generally and specifically, separately and collectively, releases, absolves and discharges the Company and its past and present agents, employees, officers, directors, shareholders, insurers, executors, attorneys, administrators, predecessors, successors, assigns, divisions, subsidiaries and affiliated companies and partnerships (collectively, the "Company Affiliates") of and from any and all claims, liabilities, demands, causes of action, costs, expenses, attorneys' fees, damages, indemnities and obligations of every kind and nature, at law, in equity or otherwise, known and unknown, discoverable and undiscoverable, suspected and unsuspected, disclosed and undisclosed, fixed or contingent, which T. Norchi or his successors and assigns ever had, now has, or hereafter can, shall or may claim to have, existing as of the date of this Agreement and arising out of, resulting from or related to the Employment Agreement or the Promissory Note (collectively, the "Released Claims").
- Acknowledgement. T. Norchi, on behalf of himself and his successors and assigns, acknowledges that the release set forth in this Section 2 is intended to include a release of presently unknown and unsuspected claims, and expressly waives any and all rights which it may have under any state or federal statute or common law principle to the contrary. T. Norchi acknowledges that there is a risk that subsequent to the execution of this Agreement he may discover additional facts related to the Released Claims, or may discover, incur, suffer or sustain injury, loss, damage, costs, attorneys' fees, expenses, or any of these, which are in some way caused by or connected with the Released Claims, or which are unknown and unanticipated as of the date hereof, or which are not presently capable of being ascertained. Nevertheless, T. Norchi acknowledges that this Agreement has been negotiated and agreed in light of that realization, and intends the release set forth in this Section 2 to include unknown and unsuspected claims within the scope of the subject matter of the release.

3. MISCELLANEOUS.

- 3.1 <u>Counterparts</u>. This Agreement may be executed in counterparts, each of which shall be deemed to be an original and together shall be deemed to be one and the same document.
- 3.2 <u>Governing Law.</u> This Agreement shall be governed by and construed in accordance with the laws of the State of Nevada, without regard to the conflicts of law principles thereof.

[The remainder of this page left intentionally blank]

	IN WITNESS WHEREOF , the parties have executed this Agreement as of the date first written above to be effective as of the
Closing.	
	ARCH THERAPEUTICS, INC.
	By: /s/ Avtar Dhillon
	Name: Avtar Dhillon
	Title: Director
	/s/ Terrence W. Norchi
	Terrence W. Norchi



ARCH THERAPEUTICS, INC. EXECUTIVE EMPLOYMENT AGREEMENT

This EXECUTIVE EMPLOYMENT AGREEMENT (the "Agreement") is entered into as of June 25, 2013 (the "Effective Date"), by and between Arch Therapeutics, Inc., a Nevada corporation (the "Company"), and Terrence W. Norchi ("Executive"). The parties hereby agree as follows:

1. Duties.

- **Position.** Executive shall serve as the Company's President and Chief Executive Officer and shall have the duties and responsibilities incident to such position and such other duties as may be determined by the Company's Board of Directors. Executive shall perform faithfully, cooperatively and diligently all of his job duties and responsibilities and agrees to and shall devote his full time, attention and effort to the business of the Company, its affiliates as directed, and other assignments as directed by the Board of Directors. Executive will report to the Board of Directors.
- **(b) Best Efforts.** Executive will expend his best efforts on behalf of the Company in connection with his employment and will abide by all of the Company's applicable employment policies and decisions made by Board of Directors, as well as all applicable federal, state and local laws, regulations or ordinances.
- (c) Start Date. The effective date of this Agreement shall be June 26, 2013 (the "Start Date"). Executive shall comply with and be bound by the terms of this Agreement commencing on the Start Date.
- (d) Other Activities. Except upon the prior written consent of the Company, Executive will not, during the term of this Agreement, (i) accept any other employment, or (ii) engage, directly or indirectly, in any other business activity (whether or not pursued for pecuniary advantage) that might interfere with Executive's duties and responsibilities hereunder or create a conflict of interest with the Company.
- **(e) No Conflict.** Executive represents and warrants that Executive's execution of this Agreement, Executive's employment with the Company, and the performance of Executive's proposed duties under this Agreement shall not violate any obligations Executive may have to any other employer, person or entity, including any obligations with respect to proprietary or confidential information of any other person or entity.

2. Compensation.

(a) Annual Base Salary. As compensation for Executive's performance of his duties hereunder, the Company shall pay to Executive an initial base annual salary of two-hundred seventy-five thousand dollars (\$275,000), starting on the Start Date (the "Annual Base Salary"), payable in accordance with the normal payroll practices of Company, less required deductions for state and federal withholding tax, social security and all other employment taxes and payroll deductions. If, due to the Company's capital constraints, the Company is unable to pay to Executive any portion of the Annual Base Salary when due, such unpaid portion of the Annual Base Salary shall be deferred, without bearing interest, and shall remain payable until such time as the Company's capital constraints are resolved and the outstanding Annual Base Salary is fully paid.



- (b) Annual Bonus. Executive shall be eligible at the sole discretion of the Board of Directors to receive an annual cash bonus in an amount up to 30% of his then-current Annual Base Salary (the "Annual Bonus"). The actual amount of the Annual Bonus will be determined by the Board of Directors based on Executive's achieving Company and personal goals established and mutually agreed upon between the Executive and the Company. If awarded, the Annual Bonus will be paid on or before March 15 of the year following the year in which the Annual Bonus was earned. If, due to the Company's capital constraints, the Company is unable to pay to Executive any portion of the Annual Bonus when due, such unpaid portion of the Annual Bonus shall be deferred, without bearing interest, and shall remain payable until such time as the Company's capital constraints are resolved and the outstanding Annual Bonus is fully paid.
- (c) Annual Review of Base Salary. Executive's Annual Base Salary will be reviewed from time to time in accordance with the established procedures of the Company for adjusting salaries for similarly situated employees.
- **Equity Grants.** Executive will be, from time to time, eligible, at the sole discretion of the Board of Directors, to receive equity incentive grants in accordance with any applicable equity incentive plan of the Company. In the event (i) of a Corporate Transaction or Change of Control (as such terms are defined in the Company's applicable equity incentive plan, including without limitation the Company's 2013 Stock Incentive Plan, as it may be amended from time to time), (ii) Executive's employment is terminated by Executive for Good Reason (as defined in Section 4 below), (iii) Executive's employment is terminated by the Company other than For Cause (as defined in Section 4 below), or (iv) Executive's employment is terminated as a result of death, then in each case 100% of the number of any unvested shares subject to any outstanding equity incentive grant made to Executive shall accelerate automatically and become vested shares as of the date of such event.

3. Benefits.

- (a) Health and Welfare Benefit Plans. The Company shall provide to Executive health, dental and vision and other benefits on the same or substantially similar terms as those provided to the other executive officers of the Company.
- **(b) Customary Benefits.** Executive shall be eligible to participate in the benefits made generally available by the Company to similarly-situated executives, in accordance with the benefit plans established by the Company, and as may be amended from time to time in the Company's sole discretion. Executive shall be eligible to participate in any equity compensation or incentive plans that the Company has adopted or may adopt in its sole discretion that are applicable to similarly-situated executives, subject in all cases to approval by the Board of Directors of any grant thereunder.



- (c) **Business Expenses.** The Company shall reimburse Executive for reasonable business expenses incurred in the performance of Executive's duties hereunder in accordance with the Company's expense reimbursement guidelines.
- **(d) Vacation.** Executive shall be entitled to paid vacation, personal and sick days each calendar year, in accordance with the Company's plans, policies and programs then in effect. Initially Executive will be granted four (4) weeks of paid vacation per annum and ten (10) additional days of paid time-off per annum.

4. At-Will Employment; Termination of Employment.

- (a) At-Will Termination by Company. Executive's employment with the Company shall be "at-will" at all times. The Company may terminate Executive's employment with the Company at any time, without any advance notice, for any reason or no reason at all, notwithstanding anything to the contrary contained in or arising from any statements, policies or practices of the Company relating to the employment, discipline or termination of its employees. Upon and after such termination, all obligations of the Company under this Agreement shall cease, except as otherwise provided herein.
- (b) Severance Upon Termination by the Company Other Than For Cause. Except in situations where the employment of Executive is terminated by the Company For Cause (as defined below), in the event that the Company terminates Executive's employment at any time, Executive shall be eligible to receive an amount equal to twelve (12) months of the Executive's then-current Annual Base Salary, payable in the form of salary continuation ("Severance"). In addition, if Executive elects to continue his group health coverage under COBRA, the Company will pay Executive's COBRA premiums for coverage until the earlier of (i) the end of the twelve (12) month period following the date of such termination; or (ii) the date Executive becomes covered under another employer's health plan; provided, however, that, in the event that the Company determines, in its sole discretion, that such payments are no longer exempt from the application of Section 409A of the Internal Revenue Code of 1986, as amended (the "Code") or may be subject to tax or penalty pursuant to Section 4980D of the Code, then the Company shall pay Executive an amount equal to each remaining COBRA premium as taxable compensation in monthly installments. Executive shall not be entitled to any Severance if Executive's employment is terminated by Executive (except as expressly provided in Section 4(f) below).
- (c) Termination For Cause. For purposes of this Agreement, "For Cause" shall mean: (i) Executive commits a crime involving dishonesty, breach of trust, or physical harm to any person; (ii) Executive willfully engages in conduct that is in bad faith and materially injurious to the Company, including without limitation misappropriation of trade secrets, fraud or embezzlement; (iii) Executive commits a material breach of this Agreement, which breach is not cured within twenty calendar days after written notice to Executive from the Company; (iv) Executive willfully refuses to implement or follow a lawful policy or directive of the Company, which breach is not cured within twenty calendar days after written notice to Executive from the Company; or (v) Executive engages in misfeasance or malfeasance demonstrated by a pattern of failure to perform job duties diligently and professionally (other than any such failure resulting from Executive's incapacity due to physical or mental illness). The Company may terminate Executive's employment For Cause at any time, without any advance notice. The Company shall pay Executive all compensation to which Executive is entitled up through the date of termination, subject to any other rights or remedies of the Company under law; and thereafter all obligations of the Company under this Agreement shall cease.



- (d) Termination By Death. Executive's employment shall terminate automatically upon Executive's death. The Company shall pay to Executive's beneficiaries or estate, as appropriate, any compensation to which Executive is entitled up through the date of termination. Thereafter all obligations of the Company under this Agreement shall cease. Nothing in this Section 4(d) shall affect any entitlement of Executive's heirs or devisees to the benefits of any life insurance plan or other applicable benefits.
- (e) At-Will Termination by Executive. Executive may terminate employment with the Company at any time for any reason or no reason at all, upon four (4) weeks' advance written notice. During such notice period Executive shall continue to diligently perform all of Executive's duties hereunder. The Company shall have the option, in its sole discretion, to make Executive's termination effective at any time prior to the end of such notice period, in which case Executive would receive compensation only up through the effective date of termination of his employment. Thereafter all obligations of the Company shall cease.
- **(f)** Severance Upon Termination By Executive for Good Reason. For purposes of this Agreement, "Good Reason" shall mean the occurrence of any of the following without Executive's prior written consent: (i) a material reduction in Executive's Annual Base Salary, except for reductions that are comparable to reductions generally applicable to similarly-situated executives of the Company; (ii) the relocation of Executive to a facility or location that is more than fifty (50) miles from his primary place of employment and such relocation results in an increase in Executive's one-way driving distance by more than fifty (50) miles; or (iii) a material and adverse change in Executive's authority, duties, or responsibilities with the Company or a material and adverse change in Executive's reporting relationship; in each case other than any isolated, insubstantial and inadvertent failure by the Company that is not in bad faith and is cured within thirty (30) business days after Executive gives the Company notice of such event, which must be given within ninety (90) days after the event giving rise to the claim of Good Reason occurs. Executive's continued employment shall not constitute consent to, or a waiver of rights with respect to, any act or failure to act constituting Good Reason hereunder; provided, however, that no such event described above shall constitute Good Reason unless: (A) Executive gives notice of termination to the Company specifying the condition or event relied upon for such termination within ninety (90) days of the initial existence of such event; and (B) the Company fails to cure the condition or event constituting Good Reason within thirty (30) days following receipt of Executive's notice of termination (the "Cure Period"). If the Company fails to remedy the condition constituting Good Reason during the applicable Cure Period, Executive's "separation from service" (within the meaning of Section 409A) must occur, if at all, within ninety (90) days following such Cure Period in order for such termination as a result of such condition to constitute a termination for Good Reason. Upon Executive's termination of his employment for Good Reason, Executive will be entitled to receive Severance. In addition, if Executive elects to continue his group health coverage under COBRA, the Company will pay Executive's COBRA premiums for coverage until the earlier of (1) the end of the twelve (12) month period following the date of such termination; or (2) the date Executive becomes covered under another employer's health plan; provided, however, that, in the event that the Company determines, in its sole discretion, that such payments are no longer exempt from the application of Section 409A of the Code or may be subject to tax or penalty pursuant to Section 4980D of the Code, then the Company shall pay Executive an amount equal to each remaining COBRA premium as taxable compensation in monthly installments.



(g) Termination Obligations

- (i) Return of Property. Executive agrees that all property (including without limitation all equipment, tangible proprietary information, documents, records, notes, contracts and computer-generated materials) furnished to or created or prepared by Executive incident to Executive's employment belongs to the Company and shall be promptly returned to the Company upon termination of Executive's employment.
- (ii) Resignation and Cooperation. Upon termination of Executive's employment, Executive shall be deemed to have resigned from all offices and directorships then held with the Company. Following any termination of employment, Executive shall cooperate with the Company in the winding up of pending work on behalf of the Company and the orderly transfer of work to other employees. Executive shall also cooperate with the Company in the defense of any action brought by any third party against the Company that relates to Executive's employment by the Company.
- (h) Release. The receipt of any payment pursuant to this Section 4 shall be subject to Executive timely signing and not revoking a standard release of all claims in a form reasonably satisfactory to the Company (the "Severance Release"). To be timely, the Severance Release must become effective and irrevocable no later than sixty (60) days following the Severance Date (the "Severance Release Deadline"). If the Severance Release does not become effective and irrevocable by the Severance Release Deadline, Executive hereby forfeits any rights to the severance benefits described in this Section 4. In no event will any severance benefits be paid under this Section 4 until the Severance Release becomes effective and irrevocable. Subject to Annex A attached hereto, severance benefits shall commence once the Severance Release becomes effective and irrevocable.
- (i) Exclusive Remedy. Executive agrees that the payments and benefits contemplated by this Section 4 (and any applicable acceleration of vesting of an equity-based award in accordance with the terms of such award in connection with the termination of Executive's employment) shall constitute the exclusive and sole remedy for any termination of his employment, and Executive covenants not to assert or pursue any other remedies, at law or in equity, with respect to any termination of employment.

5. Inventions and Proprietary Information; Prohibition on Third Party Information

(a) **Proprietary Information Agreement.** Executive shall sign and be bound by the terms of the Company's Employee Proprietary Information and Inventions Assignment Agreement (the "Proprietary Information Agreement").



(b) Non-Disclosure of Third Party Information. Executive represents, warrants and covenants that Executive shall not disclose to the Company, or use, or induce the Company to use, any proprietary information or trade secrets of others at any time, including without limitation any proprietary information or trade secrets of any former employer, if any; and Executive acknowledges and agrees that any violation of this provision shall be grounds for Executive's immediate termination and could subject Executive to substantial civil liabilities and criminal penalties. Executive further specifically and expressly acknowledges that no officer or other employee or representative of the Company has requested or instructed Executive to disclose or use any such third party proprietary information or trade secrets.

6. General Provisions.

- (a) Successors and Assigns. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their heirs, personal representatives and successors, including any successor of the Company by reason of any dissolution, merger, consolidation, sale of assets or other reorganization of the Company.
- Waiver. The rights and remedies of the parties to this Agreement are cumulative and not alternative. Neither the failure nor any delay by any party in exercising any right, power or privilege under this Agreement or the documents referred to in this Agreement will operate as a waiver of such right, power or privilege; and no single or partial exercise of any such right, power or privilege will preclude any other or further exercise of such right, power or privilege or the exercise of any other right, power or privilege. To the maximum extent permitted by applicable law, (i) no claim or right arising out of this Agreement or the documents referred to in this Agreement can be discharged by one party, in whole or in part, by a waiver or renunciation of the claim or right unless in writing signed by the other party; (ii) no waiver that may be given by a party will be applicable except in the specific instance for which it is given; and (iii) no notice to or demand on one party will be deemed to be a waiver of any obligation of such party or of the right of the party giving such notice or demand to take further action without notice or demand as provided in this Agreement or the documents referred to in this Agreement.
- (c) Validity. The invalidity or unenforceability of any provision or provisions of this Agreement shall not affect the validity or enforceability of any other provision of this Agreement, which shall remain in full force and effect.
- (d) **Headings.** The headings set forth in this Agreement are for convenience only and shall not be used in interpreting this Agreement.
- **Governing Law; Venue.** This Agreement will be governed by and construed in accordance with the laws of the United States and the State of New York, without giving effect to its conflict of law rules. Except for actions for injunctive or other equitable relief, which may be brought in any court of competent jurisdiction, any legal suit, action or proceeding arising out of or relating to this Agreement shall be commenced in a federal court in the Commonwealth of Massachusetts or in state court in the Commonwealth of Massachusetts, and each party hereto irrevocably submits to the exclusive jurisdiction and venue of any such court in any such suit, action or proceeding.



- **Counterparts.** This Agreement may be executed in one or more counterparts, all of which when fully executed and delivered by all parties hereto and taken together shall constitute a single agreement, binding against each of the parties.
- **Survival.** Sections 4, 5 and 6 of this Agreement (including the terms and provisions of the Proprietary Information Agreement as set forth therein) shall survive Executive's employment by the Company.
- (h) Notices. All notices, consents, waivers and other communications under this Agreement shall be in writing and will be deemed to have been duly given when (i) delivered by hand (with written confirmation of receipt); (ii) sent by facsimile (with written confirmation of receipt); or (iii) when received by the addressee, if sent by a nationally recognized overnight delivery service or by United States first class registered or certified mail, return receipt requested, to the principal address of the other party set forth below, or to such other address as either party shall have furnished to the other in writing in accordance herewith.

If to Executive:

Terrence W. Norchi 1 Chieftain Lane Natick, MA 01760

If to the Company:

Arch Therapeutics, Inc. Attn: Avtar Dhillon One Broadway, 14th Floor Cambridge, MA 02412

Executive's employment by the Company or any of the Company's affiliates and may not be contradicted by evidence of any prior or contemporaneous statements or agreements, except for agreements specifically referenced herein (including the Proprietary Information Agreement and any agreement relating to any stock option or other equity award that may be granted to Executive). Without limiting the generality of the foregoing, this Agreement shall supersede and replace in its entirety any agreements or other relationships relating to Executive's former employment by any subsidiary or other affiliate of the Company. To the extent that the practices, policies or procedures of the Company, now or in the future, apply to Executive and are inconsistent with the terms of this Agreement, the provisions of this Agreement shall control. Except as otherwise expressly provided herein, any subsequent change in Executive's duties, position, or compensation will not affect the validity or scope of this Agreement.

EXECUTIVE ACKNOWLEDGES THAT EXECUTIVE HAS HAD THE OPPORTUNITY TO CONSULT LEGAL COUNSEL CONCERNING THIS AGREEMENT, THAT EXECUTIVE HAS READ AND UNDERSTANDS THIS AGREEMENT IN FULL, THAT EXECUTIVE IS FULLY AWARE OF ITS LEGAL EFFECT, AND THAT EXECUTIVE HAS ENTERED INTO IT FREELY BASED ON EXECUTIVE'S OWN JUDGMENT AND NOT ON ANY REPRESENTATIONS OR PROMISES OTHER THAN THOSE CONTAINED IN THIS AGREEMENT.

[Remainder of Page Intentionally Left Blank]



IN WITNESS WHEREOF, the parties have duly executed this Agreement as of the date first written above.

EXECUTIVE	
/s/ Terrence W. Norchi Name: Terrence W. Norchi	
ARCH THERAPEUTICSS, INC.	
By: /s/ Avtar Dhillon Name: Avtar Dhillon Title: Chairman of the Board	
	[Signature Page to Executive Employment Agreement]



ANNEX A

SECTION 409A ADDENDUM

Notwithstanding anything to the contrary in the Agreement, no severance pay or benefits to be paid or provided to Executive, if any, pursuant to the Agreement that, when considered together with any other severance payments or separation benefits, are considered deferred compensation under Section 409A of the Internal Revenue Code of 1986, as amended, and the final regulations and any guidance promulgated thereunder ("Section 409A") (together, the "Deferred Payments") will be paid or otherwise provided until Executive has had a "separation from service" within the meaning of Section 409A. Similarly, no severance payable to Executive, if any, that otherwise would be exempt from Section 409A pursuant to Treasury Regulation Section 1.409A-1(b)(9) will be payable until Executive has had a "separation from service" within the meaning of Section 409A. Each payment and benefit payable under the Agreement is intended to constitute a separate payment for purposes of Section 1.409A-2(b)(2) of the Treasury Regulations.

Any severance payments or benefits under the Agreement that would be considered Deferred Payments will be paid or will commence on the sixtieth (60^{th}) day following Executive's separation from service, or, if later, such time as required by the next paragraph.

Notwithstanding anything to the contrary in the Agreement, if Executive is a "specified Executive" within the meaning of Section 409A at the time of Executive's termination (other than due to death), then the Deferred Payments that would otherwise have been payable within the first six (6) months following Executive's separation from service, will be paid on the first payroll date that occurs on or after the date six (6) months and one (1) day following the date of Executive's separation from service, but in no event later than seven (7) months after the date of such separation from service. All subsequent Deferred Payments, if any, will be payable in accordance with the payment schedule applicable to each payment or benefit. Notwithstanding anything herein to the contrary, if Executive dies following Executive's separation from service, but prior to the six (6) month anniversary of the separation from service, then any payments delayed in accordance with this paragraph will be payable in a lump sum as soon as administratively practicable after the date of Executive's death and all other Deferred Payments will be payable in accordance with the payment schedule applicable to each payment or benefit.

Any amount paid under the Agreement that satisfies the requirements of the "short-term deferral" rule set forth in Section 1.409A-1(b)(4) of the Treasury Regulations will not constitute Deferred Payments. Any amount paid under the Agreement that qualifies as a payment made as a result of an involuntary separation from service pursuant to Section 1.409A-1(b)(9)(iii) of the Treasury Regulations that does not exceed the Section 409A Limit (as defined below) will not constituted Deferred Payments. For this purpose, the "Section 409A Limit" will mean two (2) times the lesser of: (i) Executive's annualized compensation based upon the annual rate of pay paid to him during Executive's taxable year preceding his taxable year of his separation from service as determined under Treasury Regulation Section 1.409A-1(b)(9)(iii)(A)(1) and any Internal Revenue Service guidance issued with respect thereto; or (ii) the maximum amount that may be taken into account under a qualified plan pursuant to Section 401(a)(17) of the Internal Revenue Code for the year in which Executive's separation from service occurred.



The foregoing provisions are intended to comply with the requirements of Section 409A so that none of the severance payments and benefits to be provided hereunder will be subject to the additional tax imposed under Section 409A, and any ambiguities herein will be interpreted to so comply. The Company and Executive agree to work together in good faith to consider amendments to the Agreement and to take such reasonable actions which are necessary, appropriate or desirable to avoid imposition of any additional tax or income recognition prior to actual payment to Executive under Section 409A.

ARCH THERAPEUTICS, INC. EXECUTIVE EMPLOYMENT AGREEMENT

This EXECUTIVE EMPLOYMENT AGREEMENT (the "Agreement") is entered into as of June 26, 2013 (the "Effective Date"), by and between Arch Therapeutics, Inc., a Nevada corporation (the "Company"), and Alan T. Barber ("Executive"). The parties hereby agree as follows:

1. Duties.

- **Position.** Executive shall serve as the Company's Chief Financial Officer and shall have the duties and responsibilities incident to such position and such other duties as may be determined by the Company's Chief Executive Officer, including without limitation overseeing activities of the Company's controller, if any, and other members of the finance team; interacting with the Company's third-party accountants, as designated by the Company's Board of Directors or any committee thereof; and assisting the Chief Executive Officer and others with respect to financial transactions as needed. Executive shall perform faithfully, cooperatively and diligently all of his job duties and responsibilities to the business of the Company, its subsidiaries as directed, and other assignments as directed by the Chief Executive Officer. Executive will report to the Chief Executive Officer. The Company and Executive hereby acknowledge and agree that Executive shall serve as a part-time employee. Executive recognizes, understands and agrees that he may be required during certain periods to devote more time, attention and effort to his position than other periods. The Company and Executive will review from time to time as needed the number of Executive's projected weekly workdays for potential adjustment in order to meet the Company's and Executive's needs.
- **(b) Best Efforts.** Executive will expend his best efforts on behalf of the Company in connection with his employment and will abide by all of the Company's applicable employment policies and decisions made by Board of Directors, as well as all applicable federal, state and local laws, regulations or ordinances.
- (c) Start Date. Executive agrees that he will commence employment with the Company on June 26, 2013 (the "Start Date").
- (d) Other Activities. Executive will not, during the term of this Agreement, engage, directly or indirectly, in any other business activity (whether or not pursued for pecuniary advantage) that interferes with Executive's duties and responsibilities hereunder or create a conflict of interest with the Company.
- (e) No Conflict. Executive represents and warrants that Executive's execution of this Agreement, Executive's employment with the Company, and the performance of Executive's proposed duties under this Agreement shall not violate any obligations Executive may have to any other employer, person or entity, including any obligations with respect to proprietary or confidential information of any other person or entity.

2. Compensation.

- (a) Annual Base Salary. As compensation for Executive's performance of his duties hereunder, the Company shall pay to Executive an initial base annual salary of eighty-three thousand six hundred dollars (\$83,600), starting on the Start Date (the "Annual Base Salary"), payable in accordance with the normal payroll practices of Company, less required deductions for state and federal withholding tax, social security and all other employment taxes and payroll deductions.
 - **(b) Annual Bonus.** Executive shall not be eligible for an annual cash bonus.
- (c) Periodic Review of Base Salary. Executive's Annual Base Salary will be reviewed by the Company from time to time as appropriate, depending on, among other things at the Company's and the Board of Director's discretion, the amount of time Executive is required to devote to the performance of his duties hereunder.

3. Benefits.

- (a) **Health and Welfare Benefit Plans.** Executive shall be eligible for the same health, dental and vision and other benefits provided by the Company on the same or substantially similar terms as those provided to other similarly situated employees of the Company.
- **Customary Benefits.** Executive shall be eligible to participate in the benefits made generally available by the Company to similarly situated employees, in accordance with the benefit plans established by the Company, and as may be amended from time to time in the Company's sole discretion.
- **Business Expenses.** The Company shall reimburse Executive for reasonable business expenses incurred in the performance of Executive's duties hereunder in accordance with the Company's expense reimbursement guidelines.

4. At-Will Employment; Termination of Employment.

- (a) At-Will Termination by Company. Executive's employment with the Company shall be "at-will" at all times. The Company may terminate Executive's employment with the Company at any time, without any advance notice, for any reason or no reason at all, notwithstanding anything to the contrary contained in or arising from any statements, policies or practices of the Company relating to the employment, discipline or termination of its employees. Upon and after such termination, all obligations of the Company under this Agreement shall cease, except as otherwise provided herein.
- **Severance.** Executive is not eligible for severance or other payments upon any termination of his employment. In the event of any termination of Executive's employment by the Company, the Company shall pay to Executive all compensation to which Executive is entitled up through the date of termination, and thereafter all obligations of the Company under this Agreement shall cease.

- (c) By Death. Executive's employment shall terminate automatically upon Executive's death. The Company shall pay to Executive's beneficiaries or estate, as appropriate, any compensation then due and owing. Thereafter all obligations of the Company under this Agreement shall cease. Nothing in this Section shall affect any entitlement of Executive's heirs or devisees to the benefits of any life insurance plan or other applicable benefits.
- (d) By Disability. If Executive becomes eligible for the Company's long-term disability benefits, if any, or if Executive is unable to carry out the responsibilities and functions of the position held by Executive by reason of any physical or mental impairment for more than ninety consecutive days or more than one hundred and twenty days in any twelve-month period, then, to the extent permitted by law, the Company may terminate Executive's employment. The Company shall pay to Executive all compensation to which Executive is entitled up through the date of termination, and thereafter all obligations of the Company under this Agreement shall cease. Nothing in this Section shall affect Executive's rights under any disability plan in which Executive is a participant.
- (e) At-Will Termination by Executive. Executive may terminate employment with the Company at any time for any reason or no reason at all. In the event of any termination of Executive's employment by the Executive hereunder, the Company shall pay to Executive all compensation to which Executive is entitled up through the date of termination, and thereafter all obligations of the Company under this Agreement shall cease.

(f) Termination Obligations

- (i) Return of Property. Executive agrees that all property (including without limitation all equipment, tangible proprietary information, documents, records, notes, contracts and computer-generated materials) furnished to or created or prepared by Executive incident to Executive's employment belongs to the Company and shall be promptly returned to the Company upon termination of Executive's employment.
- (ii) Resignation and Cooperation. Upon termination of Executive's employment, Executive shall be deemed to have resigned from all offices and directorships then held with the Company. Following any termination of employment, Executive shall cooperate with the Company in the winding up of pending work on behalf of the Company and the orderly transfer of work to other employees. Executive shall also cooperate with the Company in the defense of any action brought by any third party against the Company that relates to Executive's employment by the Company.
- **Exclusive Remedy.** Executive covenants not to assert or pursue any remedies, at law or in equity, with respect to any termination of employment.

5. Inventions and Proprietary Information; Prohibition on Third Party Information

(a) **Proprietary Information Agreement.** Executive shall sign and be bound by the terms of the Company's Employee Proprietary Information and Inventions Assignment Agreement (the "Proprietary Information Agreement").

(b) Non-Disclosure of Third Party Information. Executive represents, warrants and covenants that Executive shall not disclose to the Company, or use, or induce the Company to use, any proprietary information or trade secrets of others at any time, including without limitation any proprietary information or trade secrets of any former employer, if any; and Executive acknowledges and agrees that any violation of this provision shall be grounds for Executive's immediate termination and could subject Executive to substantial civil liabilities and criminal penalties. Executive further specifically and expressly acknowledges that no officer or other employee or representative of the Company has requested or instructed Executive to disclose or use any such third party proprietary information or trade secrets.

6. General Provisions.

- (a) Successors and Assigns. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their heirs, personal representatives and successors, including any successor of the Company by reason of any dissolution, merger, consolidation, sale of assets or other reorganization of the Company.
- **(b) Waiver.** The rights and remedies of the parties to this Agreement are cumulative and not alternative. Neither the failure nor any delay by any party in exercising any right, power or privilege under this Agreement or the documents referred to in this Agreement will operate as a waiver of such right, power or privilege; and no single or partial exercise of any such right, power or privilege will preclude any other or further exercise of such right, power or privilege or the exercise of any other right, power or privilege. To the maximum extent permitted by applicable law, (i) no claim or right arising out of this Agreement or the documents referred to in this Agreement can be discharged by one party, in whole or in part, by a waiver or renunciation of the claim or right unless in writing signed by the other party; (ii) no waiver that may be given by a party will be applicable except in the specific instance for which it is given; and (iii) no notice to or demand on one party will be deemed to be a waiver of any obligation of such party or of the right of the party giving such notice or demand to take further action without notice or demand as provided in this Agreement or the documents referred to in this Agreement.
- (c) Validity. The invalidity or unenforceability of any provision or provisions of this Agreement shall not affect the validity or enforceability of any other provision of this Agreement, which shall remain in full force and effect.
- (d) **Headings.** The headings set forth in this Agreement are for convenience only and shall not be used in interpreting this Agreement.
- (e) Governing Law; Venue. This Agreement will be governed by and construed in accordance with the laws of the United States and the State of New York, without giving effect to its conflict of law rules. Except for actions for injunctive or other equitable relief, which may be brought in any court of competent jurisdiction, any legal suit, action or proceeding arising out of or relating to this Agreement shall be commenced in a federal court in the Commonwealth of Massachusetts or in state court in the Commonwealth of Massachusetts, and each party hereto irrevocably submits to the exclusive jurisdiction and venue of any such court in any such suit, action or proceeding.

- **Counterparts.** This Agreement may be executed in one or more counterparts, all of which when fully executed and delivered by all parties hereto and taken together shall constitute a single agreement, binding against each of the parties.
- **Survival.** Sections 4, 5 and 6 of this Agreement (including the terms and provisions of the Proprietary Information Agreement as set forth therein) shall survive Executive's employment by Company.
- (h) Notices. All notices, consents, waivers and other communications under this Agreement shall be in writing and will be deemed to have been duly given when (i) delivered by hand (with written confirmation of receipt); (ii) sent by facsimile (with written confirmation of receipt); or (iii) when received by the addressee, if sent by a nationally recognized overnight delivery service or by United States first class registered or certified mail, return receipt requested, to the principal address of the other party set forth below, or to such other address as either party shall have furnished to the other in writing in accordance herewith.

If to Executive:

Alan T. Barber 134 Alcott Road Concord, MA 01742

If to the Company:

Arch Therapeutics, Inc. Attn: Terrence Norchi PO Box 748 Natick, MA 01760

Executive's employment by the Company or any of the Company's affiliates and may not be contradicted by evidence of any prior or contemporaneous statements or agreements, except for agreements specifically referenced herein (including the Proprietary Information Agreement and any agreement relating to any stock option or other equity award that may be granted to Executive). Without limiting the generality of the foregoing, this Agreement shall supersede and replace in its entirety any agreements or other relationships relating to Executive's former employment or consulting relationship with any subsidiary or other affiliate of the Company. To the extent that the practices, policies or procedures of the Company, now or in the future, apply to Executive and are inconsistent with the terms of this Agreement, the provisions of this Agreement shall control. Except as otherwise expressly provided herein, any subsequent change in Executive's duties, position, or compensation will not affect the validity or scope of this Agreement.

EXECUTIVE ACKNOWLEDGES THAT EXECUTIVE HAS HAD THE OPPORTUNITY TO CONSULT LEGAL COUNSEL CONCERNING THIS AGREEMENT, THAT EXECUTIVE HAS READ AND UNDERSTANDS THIS AGREEMENT IN FULL, THAT EXECUTIVE IS FULLY AWARE OF ITS LEGAL EFFECT, AND THAT EXECUTIVE HAS ENTERED INTO IT FREELY BASED ON EXECUTIVE'S OWN JUDGMENT AND NOT ON ANY REPRESENTATIONS OR PROMISES OTHER THAN THOSE CONTAINED IN THIS AGREEMENT.

[Remainder of Page Intentionally Left Blank]

IN WITNESS WHEREOF, the parties have duly executed this Agreement as of the date first written above.

EXECUTIVE

/s/ Alan T. Barber

Alan T. Barber

ARCH THERAPEUTICSS,

INC.

By:/s/ Terrence W. Norchi Name: <u>Terrence W. Norchi</u>

Title: President and CEO

[Signature Page to Executive Employment Agreement]

June 26, 2013

Securities and Exchange Commission Washington, DC 20549

Ladies and Gentlemen:

We have read Item 4.01 of Form 8-K of Arch Therapeutics, Inc. dated June 26, 2013. We agree with the statements made concerning our firm contained therein.

Yours very truly,

/s/ Paritz & Company, P.A.

Paritz & Company, P.A.

SUBSIDIARIES OF ARCH THERAPEUTICS, INC.

Name	Jurisdiction
Arch Biosurgery, Inc.	Massachusetts



REPORT OF INDEPENDENT REGISTERED PUBLIC ACCOUNTING FIRM

To the Board of Directors of Arch Therapeutics Inc. Boston, Massachusetts

We have audited the accompanying balance sheets of Arch Therapeutics Inc. (the "Company") as of September 30, 2012 and 2011 and the related statements of operations, changes in stockholders' deficit and cash flows for the years then ended and for the period from inception (March 6, 2006) through September 30, 2012. These financial statements are the responsibility of the Company's management. Our responsibility is to express an opinion on these financial statements based on our audits.

We conducted our audits in accordance with the standards of the Public Company Accounting Oversight Board (United States). Those standards require that we plan and perform the audits to obtain reasonable assurance about whether the financial statements are free of material misstatement. The Company is not required to have, nor were we engaged to perform, an audit of its internal control over financial reporting. Our audits included consideration of internal control over financial reporting as a basis for designing audit procedures that are appropriate in the circumstances, but not for the purpose of expressing an opinion on the effectiveness of the Company's internal control over financial reporting. Accordingly, we express no such opinion. An audit also includes examining, on a test basis, evidence supporting the amounts and disclosures in the financial statements, assessing the accounting principles used and significant estimates made by management, as well as evaluating the overall financial statement presentation. We believe that our audits provide a reasonable basis for our opinion.

In our opinion, the financial statements referred to above present fairly, in all material respects, the financial position of Arch Therapeutics Inc. as of September 30, 2012 and 2011, and the results of their operations and their cash flows for the years then ended and for the period from inception (March 6, 2006) through September 30, 2012, in conformity with accounting principles generally accepted in the United States of America.

The accompanying financial statements have been prepared assuming that Arch Therapeutics Inc. will continue as a going concern. As discussed in Note 1 to the financial statements, the Company has suffered recurring losses and negative cash flows from operations and has an accumulated deficit that raises substantial doubt about its ability to continue as a going concern. Management's plans in regard to these matters are also described in Note 1. The financial statements do not include any adjustments that might result from the outcome of this uncertainty.

Moody, Famiglietti & Andronico, LLP

Moody, Lamiglittle & andronico, ZZP

Tewksbury, Massachusetts

June 7, 2013

(A Development Stage Company) Balance Sheets

September 30, 2012 and 2011

		2012		2011
ASSETS				
Current assets:				
Cash and cash equivalents	\$	17,139	\$	36,775
Prepaid expenses and other current assets		3,308		3,956
Total current assets		20,447		40,731
Long term assets:				
Property and equipment, net	_	908	_	4,280
Total assets	\$	21,355	\$	45,011
LIABILITIES AND STOCKHOLDERS' DEFICIT				
Current liabilities:				
Current maturities of convertible notes payable	\$	1,395,000	\$	1,245,000
Current maturities of convertible notes payable, related parties	Ψ	105,000	Ψ	55,000
Notes payable, related party		275,200		275,200
Accounts payable		258,426		118,548
Accrued expenses and other liabilities		49,510		27,002
Current portion of accrued interest		352,755		232,909
Current portion of accrued interest to related parties		116,548		67,578
Total current liabilities		2,552,439		2,021,237
Long-term liabilities:				
Convertible notes payable, net of current maturities		235,000		150,000
Convertible notes payable, net of current maturities, related parties		-		50,000
Accrued interest to related party, net of current portion		-		11,759
Accrued interest, net of current portion		6,351		7,540
Total long-term liabilities	_	241,351	_	219,299
Total liabilities		2,793,790		2,240,536
Commitments and contingencies				
Stockholders' deficit:				
Common stock, no par value, 1,275,000 shares authorized,		502		582
1,165,015 and 1,163,015 shares issued and outstanding at September 30, 2012 and 2011, respectively Deficit accumulated during the development stage		583		
Total stockholders' deficit		(2,773,018)		(2,196,107) (2,105,525)
Total Stockholders deficit		(2,772,435)		(2,195,525)
Total liabilities and stockholders' deficit	\$	21,355	\$	45,011

Statements of Operations
Years Ended September 30, 2012 and 2011 and the
Period from Inception (March 6, 2006) through September 30, 2012

		2012		2011	Inc 6,	Period from eption (March 2006) through eptember 30, 2012
Other Revenues	\$	-	\$	_	\$	431,461
Operating expenses: General and administrative expenses		333,503		362,096		2,135,965
Research and development expenses		87,021		122,738		647,772
Total operating expenses	_	420,524	_	484,834	_	
Total operating expenses	_	420,324		484,834		2,783,737
Operating loss		(420,524)		(484,834)		(2,352,276)
Other (expense) income:						
Interest expense		(156,865)		(133,462)		(479,717)
Other income		478		45,100		58,975
Total other expense		(156,387)		(88,362)		(420,742)
Net loss	\$	(576,911)	\$	(573,196)	\$	(2,773,018)
Net loss per common share basic and diluted	\$	(0.50)	\$	(0.49)	\$	(3.34)
Weighted average number of shares outstanding		1,165,015		1,163,015		831,155

(A Development Stage Company) Statements of Changes in Stockholders' Deficit Period from Inception (March 6, 2006) through September 30, 2012

	Commo Shares	on Stock Amount	Deficit Accumulated During the Development Stage	Total Stockholders' Deficit
Balance at inception (March 6, 2006)	-	\$ -	\$ -	\$ -
Net loss			(18,153)	(18,153)
Balance at September 30, 2006	-	-	(18,153)	(18,153)
Issuance of common stock to founders	660,000	330	-	330
Net loss			(447,170)	(447,170)
Balance at September 30, 2007	660,000	330	(465,323)	(464,993)
Net loss			(233,040)	(233,040)
Balance at September 30, 2008	660,000	330	(698,363)	(698,033)
Issuances of common stock	385,920	193	-	193
Net loss			(504,687)	(504,687)
Balance at September 30, 2009	1,045,920	\$ 523	\$ (1,203,050)	\$ (1,202,527)

(A Development Stage Company)
Statements of Changes in Stockholders' Deficit (continued)
Period from Inception (March 6, 2006) through September 30, 2012

	Commo	on Stock	Deficit Accumulated During the Development	Total Stockholders'
	Shares	Amount	Stage	Deficit
Balance at September 30, 2009	1,045,920	\$ 523	\$ (1,203,050)	\$ 1,202,527
Issuances of common stock	53,500	27	-	27
Net loss	_		(419,861)	(419,861)
Balance at September 30, 2010	1,099,420	550	(1,622,911)	(1,622,361)
Issuances of common stock	63,595	32	-	32
Net loss			(573,196)	(573,196)
Balance at September 30, 2011	1,163,015	582	(2,196,107)	(2,195,525)
Issuances of common stock	2,000	1	-	1
Net loss			(576,911)	(576,911)
Balance at September 30, 2012	1,165,015	\$ 583	\$ (2,773,018)	\$ (2,772,435)

(A Development Stage Company)

Statements of Cash Flows

Years Ended September 30, 2012 and 2011 and the

Period from Inception (March 6, 2006) through September 30, 2012

Period from Inception (March 6, 2006) through September 30, 2012 2011 2012 Cash flows from operating activities: (576,911) \$ (573,196) \$ Net loss \$ (2,773,018)Adjustments to reconcile net loss to cash used in operating activities: 3,372 3,624 Depreciation expense 18,145 Noncash interest expense on convertible notes payable 118,657 95,771 359,106 37,070 Noncash interest expense on notes payable to related party 37,211 116,548 Issuance of common stock for services 253 Changes in operating assets and liabilities: 648 (912)Prepaid expenses and other current assets (3,308)139,878 51,010 258,426 Accounts payable Accrued expenses and other liabilities 22,508 11,536 49,510 Net cash used in operating activities (254,636)(375,065)(1,974,338)Cash flows from investing activities: Purchases of property and equipment (19,053)Net cash used in investing activities (19,053)Cash flows from financing activities: Proceeds from issuance of common stock 330 Proceeds from issuance of notes payable to related party 1,200 275,200 105,000 Proceeds from issuance of convertible notes payable to related parties Proceeds from issuance of convertible notes payable 235,000 400,000 1,630,000 Net cash provided by financing activities 235,000 401,200 2,010,530 Net (decrease) increase in cash and cash equivalents (19,636)26,135 17,139 Cash and cash equivalents, beginning of period 36,775 10,640 Cash and cash equivalents, end of period 17,139 36,775 17,139

There has been no cash paid for interest or income taxes from inception (March 6, 2006) through September 30, 2012.

(A Development Stage Company)
Notes to Financial Statements
Years Ended September 30, 2012 and September 30, 2011 and the
Period from Inception (March 6, 2006) to September 30, 2012

1. DESCRIPTION OF BUSINESS

Arch Therapeutics, Inc. (the "Company") was incorporated March 6, 2006 in Massachusetts. The Company is a medical device company developing polymers comprising synthetic peptides intended to form gel-like barriers over wounds to stop or control bleeding and seal wounds.

The Company is in the development stage and is devoting substantially all of its efforts toward product research and development. The Company has incurred losses of \$2,748,018 since inception. To date, the Company has principally raised capital through the issuance of debt and convertible debt.

The Company expects to incur substantial expenditures for the foreseeable future for the research, development and commercialization of its potential products. The Company does not have sufficient cash and cash equivalents to support its current operating plan. The Company will be required to raise additional capital, obtain alternative means of financial support, or both, in order to continue to fund operations. However, there can be no assurance that the Company will be successful in securing additional resources on terms acceptable to the Company, if at all. Therefore, there exists substantial doubt about the Company's ability to continue as a going concern. The financial statements do not include any adjustments related to the recoverability of assets that might be necessary despite this uncertainty.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Basis of Accounting

The Company is in the development stage and is devoting substantially all of its efforts to raising capital, developing technologies, establishing customer and vendor relationships, and recruiting new employees. Accordingly, the accompanying financial statements are presented under the development stage accounting provisions of the Financial Accounting Standards Board (FASB).

Use of Estimates

Management is required to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the dates of the financial statements and the reported amounts of revenue and expenses during the reporting periods. Actual results could differ from those estimates.

Cash and Cash Equivalents

The Company considers all highly liquid investments with an original maturity of three months or less to be cash equivalents.

Concentration of Credit Risk

Financial instruments that potentially subject the Company to concentration of credit risk consist primarily of cash and cash equivalents. The Company maintains its cash in bank deposits accounts, which, at times, may exceed federally insured limits. The Company has not experienced any losses in such accounts. The Company believes it is not exposed to any significant credit risk on cash and cash equivalents.

Arch Therapeutics, Inc.
(A Development Stage Company)
Notes to Financial Statements
Years Ended September 30, 2012 and September 30, 2011 and the
Period from Inception (March 6, 2006) to September 30, 2012

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (Continued)

Property and Equipment

Property and equipment are recorded at cost and depreciated using the straight-line method over the estimated useful life of the related asset. Upon sale or retirement, the cost and accumulated depreciation are eliminated from their respective accounts, and the resulting gain or loss is included in income or loss for the period. Repair and maintenance expenditures are charged to expense as incurred.

Impairment of Long-Lived Assets

Long-lived assets are reviewed for impairment when circumstances indicate the carrying value of an asset may not be recoverable in accordance with ASC 360, *Property, Plant and Equipment*. For assets that are to be held and used, impairment is recognized when the estimated undiscounted cash flows associated with the asset or group of assets is less than their carrying value. If impairment exists, an adjustment is made to write the asset down to its fair value, and a loss is recorded as the difference between the carrying value and fair value. Fair values are determined based on quoted market values, discounted cash flows or internal and external appraisals, as applicable. Assets to be disposed of are carried at the lower of carrying value or estimated net realizable value.

Convertible Debt

The Company records a discount to convertible notes for the intrinsic value of conversion options embedded in debt instruments based upon the differences between the fair value of the underlying preferred stock at the commitment date of the note transaction and the effective conversion price embedded in the note. Debt discounts under these arrangements are amortized to noncash interest expense using the effective interest rate method over the term of the related debt to their date of maturity. If a security or instrument becomes convertible only upon the occurrence of a future event outside the control of the Company, or, is convertible from inception, but contains conversion terms that change upon the occurrence of a future event, then any contingent beneficial conversion feature is measured and recognized when the triggering event occurs and contingency has been resolved.

Income Taxes

In accordance with ASC 740, *Income Taxes*, the Company recognizes deferred tax assets and liabilities for the expected future tax consequences or events that have been included in the Company's financial statements and/or tax returns. Deferred tax assets and liabilities are based upon the differences between the financial statement carrying amounts and the tax bases of existing assets and liabilities and for loss and credit carryforwards using enacted tax rates expected to be in effect in the years in which the differences are expected to reverse. Deferred tax assets are reduced by a valuation allowance if it is more likely than not that some portion or all of the deferred tax asset will not be realized.

The Company provides reserves for potential payments of tax to various tax authorities related to uncertain tax positions when management determines that it is probable that a loss will be incurred related to these matters and the amount of the loss is reasonably determinable. The Company has no reserves related to uncertain tax positions as of September 30, 2012 and 2011.

Arch Therapeutics, Inc.
(A Development Stage Company)
Notes to Financial Statements
Years Ended September 30, 2012 and September 30, 2011 and the
Period from Inception (March 6, 2006) to September 30, 2012

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (Continued)

Revenue

In March 2010, the FASB issued ASU 2010-17, *Revenue Recognition* – Milestone Method (Topic 605). The update is effective on a prospective basis for milestones achieved in fiscal years beginning on or after June 15, 2010. The Company adopted this accounting policy effective for the fiscal year beginning October 1, 2010. The Company recognizes revenue in accordance with the milestone method of revenue recognition for arrangements involving research or development or other performance obligations whereby a portion or all of the consideration is contingent upon achievement of milestone events. Under these provisions, arrangement consideration contingent upon achievement of a milestone is recognized by the Company in the period the milestone is met when the Company concludes that the milestone is substantive. Upon inception of each applicable arrangement, the Company assesses each milestone and the consideration payable upon achievement of each milestone and concludes that the milestone is substantive if all of the following criteria are met: (i) the consideration is commensurate with the Company's performance or the enhanced value of a delivered item which is a direct result of the Company's performance to achieve the milestone, (ii) the consideration relates to past performance and there are no refund rights or other penalties related to the consideration based on completion of future performance and (iii) the consideration is reasonable relative to all the deliverables and payment terms within the arrangement. The related consideration for milestones that are considered substantive is recognized in its entirety in the period which the milestone is met. As of September 30, 2012, the Company has not recorded any revenue for these types of activities.

Other Revenue

During the period from inception (March 6, 2006) to September 30, 2012, the Company had a contract with a pharmaceutical company to allow that pharmaceutical company access to materials for a fixed period of time. Other revenue from this contract was recognized based upon the proportional performance method, over the period of access. The Company does not consider this revenue to be significant and does not consider this event to be a regular practice.

Research and Development

The Company expenses internal and external research and development costs, including costs of funded research and development arrangements, in the period incurred. Research and development related income is recognized over the term of the related project under the proportional performance method based on costs incurred.

Page 9

Arch Therapeutics, Inc.
(A Development Stage Company)
Notes to Financial Statements
Years Ended September 30, 2012 and September 30, 2011 and the
Period from Inception (March 6, 2006) to September 30, 2012

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (Continued)

Accounting for Stock-Based Compensation

The Company accounts for employee stock-based compensation in accordance with the guidance of FASB ASC Topic 718, *Compensation-Stock Compensation*, which requires all share-based payments to employees, including grants of employee stock options, to be recognized in the financial statements based on their fair values. The Company accounts for non-employee stock-based compensation in accordance with the guidance of FASB ASC Topic 505, *Equity* ("FASB ASC Topic 505"), which requires that companies recognize compensation expense based on the estimated fair value of options granted to non-employees over their vesting period, which is generally the period during which services are rendered by such non-employees. FASB ASC Topic 505 requires the Company to remeasure the fair value of stock options issued to non-employee at each reporting period during the vesting period or until services are complete.

In accordance with FASB ASC Topic 718, *Compensation-Stock Compensation*, the Company has elected to use the Black-Scholes option pricing model to determine the fair value of options granted and recognizes the compensation cost of share-based awards on a straight-line basis over the vesting period of the award.

The determination of the fair value of share-based payment awards utilizing the Black-Scholes model is affected by the fair value of the common stock and a number of other assumptions, including expected volatility, expected life, risk-free interest rate and expected dividends. The Company does not have a history of market prices of the common stock as it is not a public company, and as such volatility is estimated in accordance with Staff Accounting Bulletin ("SAB") No. 107, Share-Based Payment ("SAB No. 107"), using historical volatilities of similar public entities. The expected life of awards is estimated based on the simplified method, as defined in SAB No. 107, for employees, or the term of the award for consultants. The risk-free interest rate assumption is based on observed interest rates appropriate for the terms of our awards. The dividend yield assumption is based on history and the expectation of paying no dividends. Forfeitures are estimated at the time of grant and revised, if necessary, in subsequent periods if actual forfeitures differ from those estimates. Stock-based compensation expense, when recognized in the financial statements, is based on awards that are ultimately expected to vest.

Fair Value Measurements

The Company measures both financial and nonfinancial assets and liabilities in accordance with FASB ASC Topic 820, *Fair Value Measurements and Disclosures*, except those that are recognized or disclosed in the financial statements at fair value on a recurring basis. The standard created a fair value hierarchy which prioritizes the inputs to valuation techniques used to measure fair value into three broad levels as follows: Level 1 inputs are quoted prices (unadjusted) in active markets for identical assets or liabilities; Level 2 inputs are inputs other than quoted prices included within Level 1 that are observable for the asset or liability, either directly or indirectly; and Level 3 inputs are unobservable inputs that reflect the Company's own assumptions about the assumptions market participants would use in pricing the asset or liability.

The Company's financial instruments include cash and cash equivalents. Because of their short maturity, the carrying amount of cash and cash equivalents are considered to approximate fair value.

Subsequent Events

The Company evaluated all events or transactions that occurred through June 7, 2013, the date which these financial statements were available to be issued. The Company disclosed material subsequent events in Note 11.

(A Development Stage Company)

Notes to Financial Statements

Years Ended September 30, 2012 and September 30, 2011 and the

Period from Inception (March 6, 2006) to September 30, 2012

3. PROPERTY AND EQUIPMENT

At September 30, 2012 and 2011, property and equipment consisted of:

	Estimated Useful Life	2012		2011	
	Useful Life			2011	
Furniture and fixtures	5 years	\$	11,791	\$	11,791
Computer equipment	3 years		4,196		4,196
Lab equipment	5 years		3,066		3,066
			19,053		19,053
Less - accumulated depreciation			18,145		14,773
		\$	908	\$	4,280

Depreciation expense for the years ended September 30, 2012 and 2011, and for the period from inception (March 6, 2006) through September 30, 2012 was \$3,372, \$3,624 and \$18,145, respectively.

4. INCOME TAXES

The principal components of the Company's net deferred tax asset consisted of the following at September 30:

		2012	 2011
Net operating loss carryforwards	\$	884,891	\$ 735,621
Research and experimentation credit carryforwards		32,559	32,559
Fixed assets		7,258	5,910
Accrued expenses	_	10,000	
Gross deferred tax assets		934,708	774,090
Deferred tax asset valuation allowance		(934,708)	(774,090)
Net deferred tax asset	\$		\$ -

As of September 30, 2012 and 2011, the Company had federal net operating loss carryforwards of approximately \$2,228,000 and \$1,838,000, respectively, which may be available to offset future taxable income and which would begin to expire in 2026. As of September 30, 2012 and 2011, the Company had federal research and experimentation credit carryforwards of \$32,559 which may be available to offset future income tax liabilities and which would begin to expire in 2028.

As of September 30, 2012 and 2011, the Company had state net operating loss carryforwards of approximately \$2,120,000 and \$1,847,000, respectively, which may be available to offset future taxable income and which would begin to expire in 2012. As of September 30, 2012 and 2011, the Company had state research and experimentation credit carryforwards of \$10,135 which may be available to offset future income tax liabilities and which would begin to expire in 2023.

4. INCOME TAXES (Continued)

As the Company has not yet achieved profitable operations, management believes the tax benefits as of September 30, 2012 and 2011 did not satisfy the realization criteria set forth in FASB ASC Topic 740, *Income Taxes*, and therefore has recorded a valuation allowance for the entire deferred tax asset. The valuation allowance increased in 2012 and 2011 by approximately \$198,220 and \$151,930, respectively. The Company's effective income tax rate differed from the federal statutory rate due to state taxes and the Company's full valuation allowance, the latter of which reduced the Company's effective federal income tax rate to zero.

Ownership changes, as defined in the Internal Revenue Code, may limit the amount of net operating loss carryforwards that can be utilized annually to offset future taxable income. Subsequent ownership changes could further affect the limitation in future years.

5. RELATED PARTY TRANSACTIONS

Notes Payable, Related Party

In February 2009, the Company issued a promissory note to Terrence Norchi, a shareholder and director of the Company. During the period from February 2009 through February 2011, aggregate cash proceeds of \$275,200 were advanced to the Company under the note. The note accrued interest at a rate of 6% per year through December 31, 2009 and 10% per year beginning January 1, 2010. The original maturity date of the note was August 10, 2010. In connection with the note, the Company issued warrants to purchase shares of convertible preferred stock at the purchase price of such stock equal to 20% of the principal balance of the note divided by the purchase price. The warrants expire during the period beginning February 13, 2016 through February 25, 2018. As of June 7, 2013 the Company has not completed, nor are there plans to complete, a preferred equity financing that would resolve the contingency associated with the convertible preferred stock warrants.

Upon maturity of the note on August 10, 2010, the note holder entered into an agreement of forbearance with the Company extending the time to repay the note and accrued interest for an unspecified period of time. Under the terms of the agreement, interest will continue to accrue at 10% per year until the note is paid.

Convertible Notes Payable, Related Parties

Beginning in June 2006 through December 2008, the Company issued convertible notes with related parties for aggregate cash proceeds of \$105,000. The notes accrue interest at various rates ranging from 6% to 10% per year and had an original maturity date of two years from issuance. The notes are convertible into the number of shares of convertible preferred stock upon the closing of a preferred equity financing of at least \$1,000,000 by dividing the principle and accrued interest by the purchase price of the convertible preferred stock ("conversion price").

In connection with the notes, the Company issued warrants to purchase additional shares of convertible preferred stock at the conversion price equal to an aggregate amount of 20%. The warrants expire in June 2016. As of June 7, 2013 the Company has not completed, nor are there plans to complete, a preferred equity financing that would resolve the contingency associated with the convertible preferred stock warrants. At September 30, 2012 and 2011, \$55,000 of the convertible notes with related parties had matured. Each of the holders of the matured notes entered into an agreement of forbearance with the Company extending the time to repay the matured notes and accrued interest for an unspecified amount of time. Under the terms of the agreement, interest will continue to accrue at the rate in effect at the time of maturity. \$50,000 of notes expire January 1, 2013.

5. RELATED PARTY TRANSACTIONS (Continued)

Directors Compensation

In November 2010, the Company entered into an agreement to pay the Chief Executive Officer ("CEO") a cash bonus of \$500,000 upon the raising of capital from a financing of at least \$1,000,000. Additionally, the Company agreed that upon such closing, warrants shall be issued to the CEO allowing the purchase of the number of shares of convertible preferred stock equal to \$100,000 divided by the purchase price per share of the convertible preferred stock. As of June 7, 2013 the Company has not completed, nor are there plans to complete, a preferred equity financing that would resolve the contingency associated with the convertible preferred stock warrants.

6. CONVERTIBLE NOTES PAYABLE

Beginning in March 2006 through September 30, 2012, the Company issued convertible notes for aggregate cash proceeds of \$1,485,000. The notes accrue interest at various rates ranging from 6% to 10% per year and had an original maturity date of two years from issuance. The notes are convertible into the number of shares of convertible preferred stock upon the closing of a preferred equity financing of at least \$1,000,000 by dividing the principal and accrued interest by the purchase price of the convertible preferred stock ("conversion price"). In connection with the notes, the Company issued warrants to purchase additional shares of convertible preferred stock at the conversion price equal to an aggregate amount ranging from 10% to up to 50% of the principal balance of the note. The warrants expire at various dates through January 2015. As of June 7, 2013 the Company has not completed, nor are there plans to complete, a preferred equity financing that would resolve the contingency associated with the convertible preferred stock warrants.

On July 5, 2011, the Company issued a convertible note for cash proceeds of \$250,000. The note accrues interest at 6% per year and matured in one year. The note is convertible into the number of shares of common stock upon the closing of an equity financing of at least \$750,000 by dividing the principal and accrued interest by the purchase price of the stock sold in the equity financing. Upon maturity of the note on July 5, 2012, the note holder entered into an agreement of forbearance with the Company extending time to repay the matured note and the accrued interest for an unspecified period of time. Under the terms of the agreement, interest will continue to accrue at 6% per year until the note is paid or converted.

The Company held \$1,025,000 of notes that had matured during the year ended September 30, 2011. An additional \$275,000 matured during the year ended September 30, 2012, bringing the total to \$1,300,000. Each of the holders of the matured notes entered into an agreement of forbearance with the Company extending the time to repay the matured notes and the accrued interest for an unspecified period of time. Under the terms of the agreement, interest will continue to accrue at the rate in effect at the time of maturity.

Arch Therapeutics, Inc.
(A Development Stage Company)
Notes to Financial Statements

Years Ended September 30, 2012 and September 30, 2011 and the Period from Inception (March 6, 2006) to September 30, 2012

CONVERTIBLE NOTES PAYABLE (Continued)

All notes maturing within twelve months are classified as short-term. As of September 30, 2012, \$235,000 of long-term convertible notes payable are classified as long-term and will all mature during the year ending September 30, 2014.

7. STOCKHOLDERS' EQUITY

6.

In November 2006, the Company issued 660,000 shares of restricted common stock to the founders of the Company for consideration equal to the fair value of the common stock. The shares of common stock issued were subject to vesting and became fully vested in 2010.

In May 2009, the Company issued 303,000 additional shares of common stock to the founders of the Company for consideration at fair market value. Upon issuance of the shares to the founders, the founders entered into restricted stock agreements whereby the shares of common stock issued were subject to vesting and became fully vested in 2010.

In May 2009, the Company issued 46,700 shares of common stock in connection with a license of technology from the Massachusetts Institute of Technology ("MIT") and 11,680 shares of common stock as a gift to the Deshpande Center, an affiliate of MIT. Under the terms of the license, upon the sale of additional shares of common stock or conversion of debt into additional shares of common stock up to a total of \$4,000,000 in the aggregate, the Company is obligated to issue additional shares of common stock to the MIT in an amount to maintain the their equity ownership at 4% on a fully diluted basis.

8. STOCK-BASED COMPENSATION

During 2009, the Company established the 2009 Stock Incentive Plan (the "Plan). Under the Plan, a maximum number of 146,000 shares of the Company's authorized and available common stock may be issued in the form of stock options and other equity interests. Under the terms of the Plan, options and other equity interests may be granted to employees, officers, directors, consultants and advisors of the Company. The exercise price of each stock option shall be the fair market value as determined in good faith by the Board of Directors (the Board) at the time each option is granted.

Beginning in May, 2009 through November 2011, 119,495 shares of common stock subject to vesting were issued under the Plan to consultants at fair market value. An additional 24,140 shares were issued to consultants not subject to vesting terms.

Non-employee shares subject to vesting are revalued at each vesting date and at the end of the reporting period, with all changes in fair value recorded as stock-based compensation expense. There were no changes in fair value during the period from issuance of the shares through September 30, 2012 and accordingly, no expense has been recorded by the Company.

(A Development Stage Company)

Notes to Financial Statements

Years Ended September 30, 2012 and September 30, 2011 and the

Period from Inception (March 6, 2006) to September 30, 2012

8. STOCK-BASED COMPENSATION (Continued)

Stock compensation plan activity for the years ended September 30, 2012 and 2011 follows:

Restricted Stock	2012	2011
Non Vested at October 1	18,769	1,847
Awarded	2,000	63,595
Vested	(9,038)	46,673
Forfeited	<u>-</u>	
Non Vested at September 30	11,731	18,769

Shares available for grant totaled 2,365 and 4,365 for the years ended September 30, 2012 and 2011, respectively. The weighted average restricted stock award date fair value information for the years ended September 30, 2012 and 2011 follows:

	 2012	2011		
Non Vested at October 1	\$ 0.0005	\$	0.0005	
Awarded	0.0005		0.0005	
Vested	0.0005		0.0005	
Forfeited	0.0005		0.0005	
Non Vested at September 30	\$ 0.0005	\$	0.0005	

The weighted average contractual terms (in years) for the years ended September 30, 2012 and 2012 follows:

	2012	2011
Outstanding	7.40	8.38
Expected to Vest	8.65	9.65

9. WARRANTS

During the period from inception (March 6, 2006) through September 30, 2012, the Company has issued a total of 38 warrants, all of which have been attached to various debt instruments and commitments issued by the Company. The warrants issued are convertible into shares of Series A Preferred Stock, \$.01 par value at the conversion price equal to an aggregate amount ranging from 10% to up to 50% of the principal balance of the debt Conversion of all warrants is contingent on the Company completing a Series A Preferred Equity Financing, defined as the sale of financing securities to a third party in which the Company receives gross proceeds from investors of at least \$1,000,000, excluding the conversion of the notes. The warrants carry a ten-year term, and may be net share settled.

9. WARRANTS (Continued)

The Company, to date, has not completed a Series A Preferred Equity Financing, and as such, these warrants are deemed to be contingently convertible. Due to the terms of these warrants, the Company has not recorded any amounts in its financial statements. As of June 7, 2013, the Company has no plans to complete a financing that would result in the contingency being resolved.

10. COMMITMENTS AND CONTINGENCIES

In the ordinary course of business, the Company enters into various agreements containing standard indemnification provisions. The Company's indemnification obligations under such provisions are typically in effect from the date of execution of the applicable agreement through the end of the applicable statute of limitations. The aggregate maximum potential future liability of the Company under such indemnification provisions is uncertain. As of September 30, 2012 and 2011, no amounts have been accrued related to such indemnification provisions.

From time to time, the Company may be exposed to litigation in connection with its operations. The Company's policy is to assess the likelihood of any adverse judgments or outcomes related to legal matters, as well as ranges of probable losses.

In December, 2007, the Company entered into a license agreement with MIT pursuant to which the Company acquired an exclusive world-wide license to develop and commercialize technology related to self-assembling peptide compositions, and methods of making and using such compositions in medical and non-medical applications, including claims that cover the Company's proposed products and methods of use thereof. The license also provides non-exclusive rights to additional intellectual property in the fields that cover the Company's proposed products and methods of use thereof, in order to provide freedom to operate. The license provides the Company a right to sublicense the exclusively licensed intellectual property. The Company has not sublicensed the exclusively licensed intellectual property to any party for any field.

In exchange for the licenses granted in the agreement, the Company has paid MIT license maintenance fees and patent prosecution costs. The Company paid license maintenance fees of \$10,000 to MIT in each of the years ended September 30, 2009 and 2010. There were no payments made for license maintenance fees in the years ended September 30, 2011 and 2012.

Annual license maintenance obligations extend through the life of the patents. The following table reflects the Company's annual license maintenance fee commitments:

Year Ending		
September 30,	_	
2013	\$	25,000
2014		35,000
2015		35,000 45,000
2016		50,000
2017		50,000 50,000
	\$	205,000

10. COMMITMENTS AND CONTINGENCIES (Continued)

For each year that the agreement is in effect after 2017, the annual license maintenance fee commitment would be \$50,000. In addition, MIT is entitled to royalties on applicable future product sales, if any. The annual payments may be applied towards royalties payable to MIT for that year for product sales.

The Company is obligated to indemnify MIT and related parties from losses arising from claims relating to the exercise of any rights granted to the Company under the license, with certain exceptions. The maximum potential amount of future payments the Company could be required to make under this provision is unlimited. The Company considers there to be a low performance risk as of September 30, 2012.

The agreement expires upon the expiration or abandonment of all patents that are issued and licensed to the Company by MIT under such agreement. The Company expects that patents will be issued from presently pending U.S. and foreign patent applications. Any such patent will have a term of 20 years from the filing date of the underlying application. MIT may terminate the agreement immediately, if the Company ceases to carry on its business, if any nonpayment by the Company is not cured or the Company commits a material breach that is not cured. The Company may terminate the agreement for any reason upon six months notice to MIT.

11. SUBSEQUENT EVENTS

During December 2012 and January 2013 the Company issued convertible notes for aggregate cash proceeds of \$250,000. The notes accrue interest at 8% per year and mature two years from issuance. The notes are convertible into the number of shares of convertible preferred stock upon the closing of a preferred equity financing of at least \$1,000,000 by dividing the principle and accrued interest by the purchase price of the convertible preferred stock ("conversion price"). In connection with the notes, the Company issued warrants to purchase additional shares of convertible preferred stock at the conversion price equal to an aggregate amount of 20% of the principle balance of the notes.

On May 10, 2013 the Company entered into a definitive agreement to merge with another company ("Acquirer"). The Acquirer will acquire all of the issued and outstanding shares of the Company in exchange for 2.5 shares of common stock of the Acquirer for each share of the Company and will assume the convertible notes and warrants of the Company in exchange for the issuance to the note holders shares of common stock of the Acquirer equal to the amount of principle and interest accrued through April 30, 2013 divided by 3.64. Pursuant to the agreement, the Acquirer has advanced \$1,250,000 to the Company in exchange for promissory notes. The notes bear interest at 5% per year. At the Closing, the Company shall become wholly-owned by the Acquirer and the notes evidencing the advance from the Acquirer shall be cancelled as intercompany transactions in connection with the merger.

(A Development Stage Company) Balance Sheets March 31, 2013 and September 30, 2012

		March 31, 2013 (Unaudited)		ptember 30, 2012
ASSETS	,	C naudica)		
Current assets:				
Cash and cash equivalents	\$	881	\$	17.139
Prepaid expenses and other current assets	·	1,121	·	3,308
Total current assets		2,002		20,447
Long Term Assets:				
Property and equipment, net		61		908
Total assets	\$	2,063	\$	21,355
LIABILITIES AND STOCKHOLDERS' DEFICIT				
Current liabilities:				
Current maturities of convertible notes payable	\$	1,455,000	\$	1,395,000
Convertible notes payable, related parties	Ψ	105,000	Ψ	105,000
Notes payable, related party		275,200		275,200
Accounts payable		303,464		258,426
Accrued expenses and other liabilities		24,950		49,510
Current portion of accrued interest		413,861		352,755
Accrued interest to related parties		134,917		116,548
Total current liabilities	_	2,712,392		2,552,439
Long-term liabilities:				
Convertible notes payable, net of current maturities		425,000		235,000
Accrued interest, net of current portion		14,999		6,351
Total long-term liabilities		439,999		241,351
Total liabilities		3,152,391		2,793,790
Commitments and contingencies				
Stockholders' deficit:				
Common stock, no par value, 1,275,000 shares authorized, 1,165,015 shares issued and outstanding at March 31, 2013 and September 30, 2012		583		583
Deficit accumulated during the development stage		(3,150,911)		(2,773,018)
Total stockholders' deficit		(3,150,328)	Ξ	(2,772,435)
Total liabilities and stockholders' deficit	\$	2,063	\$	21,355

The accompanying notes are an integral part of these financial statements.

	March	n 31, 2013	March 31, 2012	Period from Inception (March 6, 2006) through March 31, 2013
Other Revenues	\$	<u>-</u>	\$ -	\$ 431,461
Operating expenses:				
General and administrative expenses		278,410	231,097	2,414,375
Research and development expenses	<u> </u>	11,290	17,912	659,062
Total operating expenses	·	289,700	249,009	3,073,437
Operating loss		(289,700)	(249,009)	(2,641,976)
Other (expense) income:				
Interest expense		(88,213)	(75,469)	(567,930)
Other income		20	463	58,995
Total other expense		(88,193)	(75,006)	(508,935)
Net loss	\$	(377,893)	\$ (324,015)	\$ (3,150,911)
		() , , , , , , , , , , , , , , , , , ,		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
Net loss per common share basic and diluted	\$	(0.32)	\$ (0.28)	\$ (3.69)
	Ψ	(0.52)	(3.20)	+ (3.07)
Weighted average number of shares outstanding		1,165,015	1,164,015	854,779
0		1,105,015	1,107,013	054,777

The accompanying notes are an integral part of these financial statements.

	March 31, 2013	March 31, 2012	Period from Inception (March 6, 2006) through March 31, 2013
Cash flows from operating activities:			
Net loss	\$ (377,893)	\$ (324,015)	\$ (3,150,911)
Adjustments to reconcile net loss to cash used in operating activities:			
Depreciation expense	847	1,806	18,992
Noncash interest expense on convertible notes payable	69,754	56,612	428,860
Noncash interest expense on notes payable to related party	18,369	18,561	134,917
Issuance of common stock for services	-	-	253
Changes in operating assets and liabilities:	-	-	-
Prepaid expenses and other current assets	2,187	2,561	(1,121)
Accounts payable	45,038	167,391	303,464
Accrued expenses and other liabilities	(24,559)	(8,481)	24,951
Net cash used in operating activities	(266,257)	(85,565)	(2,240,595)
, 0			
Cash flows from investing activities:			
Purchases of property and equipment	_	_	(19,053)
Net cash used in investing activities			(19,053)
			(17,033)
Cash flows from financing activities:			
Proceeds from issuance of common stock	-	_	330
Proceeds from issuance of notes payable to related party	-	-	275,200
Proceeds from issuance of convertible notes to related parties	-	_	105,000
Proceeds from issuance of convertible notes payable	250,000	60,000	1,880,000
Net cash provided by financing activities	250,000	60,000	2,260,530
Provide Provide Sylvania Salaria	230,000	00,000	2,200,330
Net increase in cash and cash equivalents	(16,258)	(25,565)	882
•			
Cash and cash equivalents, beginning of period	17,139	36,775	
Cash and cash equivalents, end of period	\$ 881	\$ 11,210	\$ 882

There has been no cash paid for interest or income taxes from inception (March 6, 2006) through March 31, 2013.

The accompanying notes are an integral part of these financial statements.

1. BASIS OF PRESENTATION AND DESCRIPTION OF BUSINESS

The accompanying unaudited interim financial statements have been prepared in accordance with accounting principles generally accepted in the United States of America ("US GAAP") for interim financial information and with the instructions to Form 10-Q and Rule 8-03 of Regulation S-X. Accordingly, these interim financial statements do not include all of the information and disclosures required by U.S. GAAP for complete financial statements. In management's opinion, the interim financial statements and accompanying condensed notes reflect all adjustments, consisting of normal and recurring adjustments, that are necessary for a fair presentation of these financial statements.

The results of operations for the interim periods are not necessarily indicative of the results to be expected for other interim periods or for the entire year. This information should be read in conjunction with the audited financial statements and notes thereto included as of and for the year ended September 30, 2012.

Arch Therapeutics, Inc. (the "Company") is a medical device company developing polymers comprising synthetic peptides intended to form gel-like barriers over wounds to stop or control bleeding and seal wounds. The Company is in the development stage and is devoting substantially all of its efforts toward product research and development.

The accompanying unaudited interim financial statements have been prepared assuming that the Company will continue as a going concern. As reflected in the financial statements, the Company has an accumulated deficit, has suffered significant net losses and negative cash flows from operations, and has negative working capital. The Company expects to incur substantial expenditures for the foreseeable future for the research, development and commercialization of its potential products. The Company does not have sufficient cash and cash equivalents to support its current operating plan. The Company will be required to raise additional capital, obtain alternative means of financial support, or both, in order to continue to fund operations. Therefore, there exists substantial doubt about the Company's ability to continue as a going concern. The unaudited interim financial statements do not include any adjustments that might result should the Company be unable to continue as a going concern.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES

Basis of Accounting

The Company is in the development stage and is devoting substantially all of its efforts to raising capital, developing technologies, establishing customer and vendor relationships, and recruiting new employees. Accordingly, the accompanying financial statements are presented under the development stage accounting provisions of the Financial Accounting Standards Board (FASB).

Use of Estimates

Management is required to make estimates and assumptions that affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the dates of the financial statements and the reported amounts of revenue and expenses during the reporting periods. Actual results could differ from those estimates.

2. SUMMARY OF SIGNIFICANT ACCOUNTING POLICIES (Continued)

For a complete summary of our significant accounting policies, refer to Note 2 of our audited financial statements for the fiscal year ended September 30, 2012. There have been no material changes to our significant accounting policies during the six months ended March 31, 2013.

Subsequent Events

The Company evaluated all events or transactions that occurred through June 14, 2013, the date which these financial statements were available to be issued. The Company disclosed material subsequent events in Note 6.

3. RELATED PARTY TRANSACTIONS

Beginning in June 2006 through December 2008, the Company issued convertible notes with related parties for aggregate cash proceeds of \$105,000, of which \$50,000 expired January 1, 2013.

4. COMMITMENTS AND CONTINGENCIES

In the ordinary course of business, the Company enters into various agreements containing standard indemnification provisions. The Company's indemnification obligations under such provisions are typically in effect from the date of execution of the applicable agreement through the end of the applicable statute of limitations. The aggregate maximum potential future liability of the Company under such indemnification provisions is uncertain. As of September 30, 2012 and 2011, no amounts have been accrued related to such indemnification provisions.

From time to time, the Company may be exposed to litigation in connection with its operations. The Company's policy is to assess the likelihood of any adverse judgments or outcomes related to legal matters, as well as ranges of probable losses.

In December, 2007, the Company entered into a license agreement with MIT pursuant to which the Company acquired an exclusive world-wide license to develop and commercialize technology related to self-assembling peptide compositions, and methods of making and using such compositions in medical and non-medical applications, including claims that cover the Company's proposed products and methods of use thereof. The license also provides non-exclusive rights to additional intellectual property in the fields that cover the Company's proposed products and methods of use thereof, in order to provide freedom to operate. The license provides the Company a right to sublicense the exclusively licensed intellectual property. The Company has not sublicensed the exclusively licensed intellectual property to any party for any field.

In exchange for the licenses granted in the agreement, the Company has paid MIT license maintenance fees and patent prosecution costs. The Company paid license maintenance fees of \$10,000 to MIT in each of the years ended September 30, 2009 and 2010. There were no payments made for license maintenance fees in the years ended September 30, 2011 and 2012.

4. COMMITMENTS AND CONTINGENCIES (Continued)

Annual license maintenance obligations extend through the life of the patents. The following table reflects the Company's annual license maintenance fee commitments:

Year	Endi	ng
Sente	mhei	- 30

September 50,		
2013	\$ 25,000	
2014	35,000	
2015	45,000	
2016	50,000	
2017	50,000	
	\$ 205,000	

For each year that the agreement is in effect after 2017, the annual license maintenance fee commitment would be \$50,000. In addition, MIT is entitled to royalties on applicable future product sales, if any. The annual payments may be applied towards royalties payable to MIT for that year for product sales.

The Company is obligated to indemnify MIT and related parties from losses arising from claims relating to the exercise of any rights granted to the Company under the license, with certain exceptions. The maximum potential amount of future payments the Company could be required to make under this provision is unlimited. The Company considers there to be a low performance risk as of September 30, 2012.

The agreement expires upon the expiration or abandonment of all patents that are issued and licensed to the Company by MIT under such agreement. The Company expects that patents will be issued from presently pending U.S. and foreign patent applications. Any such patent will have a term of 20 years from the filing date of the underlying application. MIT may terminate the agreement immediately, if the Company ceases to carry on its business, if any nonpayment by the Company is not cured or the Company commits a material breach that is not cured. The Company may terminate the agreement for any reason upon six months notice to MIT.

5. CONVERTIBLE NOTES PAYABLE

During December 2012 and January 2013 the Company issued convertible notes for aggregate cash proceeds of \$250,000. The notes accrue interest at 8% per year and mature two years from issuance. The notes are convertible into the number of shares of convertible preferred stock upon the closing of a preferred equity financing of at least \$1,000,000 by dividing the principle and accrued interest by the purchase price of the convertible preferred stock ("conversion price"). In connection with the notes, the Company issued warrants to purchase additional shares of convertible preferred stock at the conversion price equal to an aggregate amount of 20% of the principle balance of the notes.

6. SUBSEQUENT EVENTS

On April 30, 2013, the Company repurchased two convertible notes and related warrants (together, the "Purchased Notes") from the original holders for an aggregate amount equal to the outstanding principal and accrued and unpaid interest of \$370,397 (the "Aggregate Amount"). On June 18, 2013, the Purchased Notes were subsequently sold to new holders for the Aggregate Amount.

6. SUBSEQUENT EVENTS (Continued)

On May 10, 2013 the Company entered into a definitive agreement to merge with another company ("Acquirer"). The Acquirer will acquire all of the issued and outstanding shares of the Company in exchange for 2.5 shares of common stock of the Acquirer for each share of the Company and will assume the convertible notes and warrants of the Company in exchange for the issuance to the note holders shares of common stock of the Acquirer equal to the amount of principle and interest accrued through April 30, 2013 divided by 3.64. Pursuant to the agreement, the Acquirer has advanced \$1,250,000 to the Company in exchange for promissory notes. The notes bear interest at 5% per year. At the Closing, the Company shall become wholly-owned by the Acquirer and the notes evidencing the advance from the Acquirer shall be cancelled as intercompany transactions in connection with the merger.

On June 18, 2013, the Company repaid to a shareholder and director of the Company \$373,187 of principal and accrued interest associated with a related party note payable. In connection with the repayment, warrants to purchase shares of convertible preferred stock that had been issued with the related party note payable were terminated.

Arch Therapeutics, Inc. Unaudited Pro Forma Condensed Combined Financial Statements

Introductory Note

On May 10, 2013, Almah, Inc., a Nevada corporation (now known as Arch Therapeutics, Inc., the "Company", "Arch Therapeutics", "we", "us", or "our"), entered into an Agreement and Plan of Merger (the "Merger Agreement") with Arch Biosurgery, Inc. (formerly known as Arch Therapeutics, Inc.), a Massachusetts corporation ("ABS"), and Arch Acquisition Corporation, a Massachusetts corporation and our wholly-owned subsidiary formed for the purpose of the transaction ("Merger Sub"). The Merger Agreement provided for the merger of Merger Sub with and into ABS (the "Merger"), with ABS surviving the Merger as our wholly owned subsidiary, upon the terms and subject to the conditions set forth in the Merger Agreement. Pursuant to the Merger, the former shareholders of ABS received 5,645,212 shares of the Company's common stock and the former holders of ABS convertible debt and warrants received 9,000,000 shares of the Company's common stock. Effective June 5, 2013, the Company changed its name from Almah, Inc. to Arch Therapeutics, Inc. The Merger closed on June 26, 2013.

In contemplation of the Merger, on April 19, 2013, we entered into a financing agreement with Coldstream Summit Ltd., pursuant to which we agreed to issue and sell, and Coldstream agreed to purchase or assist in securing the purchase of 4,000,000 shares of the Company's common stock within the 12 month period following the closing of the Merger (the "Coldstream Financing"). Prior to the closing of the Merger, the Company issued 2,500,000 shares of its common stock in exchange for gross proceeds in the amount of \$1,250,000.

As a result of these transactions we acquired assets, and started operations sufficient to cease being a shell company, as defined in Rule 12b-2. Additional information regarding these transactions and the assets are contained herein.

Prior to the Merger, the Company had an authorized capitalization consisting of 300,000,000 shares of Common Stock, of which, 44,000,000 shares of common stock were issued and outstanding as of June 26, 2013 immediately prior to giving effect to the Merger.

The following unaudited pro forma condensed combined balance sheets are based on the historical balance sheets of the Company and ABS as of March 31, 2013 and September 30, 2012, respectively.

The Arch Therapeutics unaudited pro forma statements of condensed combined statement of operations give effect to the Merger as if it had occurred on the first day of the period presented and the unaudited pro forma condensed combined balance sheet gives effect to the Merger as if it had occurred on September 30, 2012.

(A Development Stage Company)
Pro Forma Condensed Combined Balance Sheet
Almah, Inc. as of March 31, 2013
Arch Biosurgery, Inc. as of March 31, 2013

	Aln	Almah, Inc.		_Almah, Inc		Almah, Inc.		Almah, Inc.		Almah, Inc.		Almah, Inc.		Almah, Inc. Arch Biosurgery, Inc.		Combined Totals		Combined Totals		Combined Totals				Pro Forma djustments	Ref	Adjusted Pro Forma Totals	
ASSETS																											
Current assets:																											
Cash and cash equivalents	\$	52	\$	881	\$	933	\$	1,250,000	[1]	\$	1,250,933																
Prepaid expenses and other current assets		250		1,121		1,371		-			1,371																
Total current assets		302		2,002		2,304		1,250,000			1,252,304																
Long Term Assets:																											
Property and equipment, net				61	_	61					61																
Total assets	\$	302	\$	2,063	\$	2,365	\$	1,250,000		\$	1,252,365																
LIABILITIES AND STOCKHOLDERS' DEFICIT																											
Current liabilities:																											
Current maturities of convertible notes payable	\$	-	\$	1,455,000	\$	1,455,000	\$	(1,455,000)	[2]	\$	-																
Convertible notes payable, related parties		-		105,000		105,000		(105,000)	[2]		-																
Notes payable, related party		5,935		275,200		281,135		-			281,135																
Accounts payable		-		303,464		303,464		-			303,464																
Accrued expenses and other liabilities		1,525		24,950		26,475		-			26,475																
Current portion of accrued interest		-		413,861		413,861		(413,861)	[2]		-																
Accrued interest to related parties		-		134,917		134,917		(42,988)	[2]		91,929																
Total current liabilities		7,460		2,712,392		2,719,852		(2,016,849)			703,003																
Long-term liabilities:																											
Convertible notes payable, net of current maturities		-		425,000		425,000		(425,000)	[2]		-																
Accrued interest, net of current portion		-		14,999		14,999		(14,999)	[2]																		
Total long-term liabilities	_			439,999	_	439,999	_	(439,999)																			
Total liabilities	_	7,460		3,152,391	_	3,159,851	_	(2,456,848)			703,003																
Stockholders' deficit:																											
Common stock		6,030		583		6,613		11,500	[1],[2]		18,113																
Additional paid in capital		34,270		-		34,270		3,647,890	[1],[2],[3]		3,682,160																
Deficit accumulated during the development stage		(47,458)		(3,150,911)		(3,198,369)		47,458	[3]		(3,150,911)																
Total stockholders' deficit		(7,158)		(3,150,328)		(3,157,486)		3,706,848			549,362																
Total liabilities and stockholders' deficit	\$	302	\$	2,063	\$	2,365	\$	1,250,000		\$	1,252,365																

Arch Therapeutics, Inc.
(A Development Stage Company)
Pro Forma Condensed Combined Statement of Operations
Almah, Inc. Six months Ended March 31, 2013
Arch Biosurgery, Inc. Six months Ended March 31, 2013

	Almah, Inc.	Arch Biosurgery, Inc.	Combined Totals	Pro Forma Adjustments	Ref	Adjusted Pro Forma Totals
Other revenues	\$ -	\$ -	\$ -	\$ -		\$ -
Operating expenses:						
General and administrative expenses	5,811	278,411	284,222	-		284,222
Research and development expenses Total operating expenses	5,811	11,290 289,701	11,290 295,512			11,290 295,512
Operating loss	(5,811)	(289,701)	(295,512)			(295,512)
Other (expense) income:						
Interest expense	-	(88,212)	(88,212)	74,489	[2]	(13,723)
Other income		20	20			20
Total other expense		(88,193)	(88,193)	74,489		(13,704)
Net loss	\$ (5,811)	\$ (377,893)	\$ (383,704)	\$ 74,489		\$ (309,215)
Net loss per common share basic and diluted	\$ (0.00)	\$ (0.32)				\$ (0.01)
Weighted average number of shares outstanding	6,030,000	1,165,015				37,015,798

(A Development Stage Company)
Pro Forma Condensed Combined Balance Sheet
Almah, Inc. as of September 30, 2012
Arch Biosurgery, Inc. as of September 30, 2012

	Alm	nah, Inc.	Inc. Arch Biosurgery, Inc.		Combined Totals		Pro Forma Combined Totals Adjustments		Ref	Adjusted Pro Forma Tota	
ASSETS											
Current assets:											
Cash and cash equivalents	\$	4,468	\$	17,139	\$	21,607	\$	1,250,000	[1]	\$	1,271,607
Prepaid expenses and other current assets		250		3,308		3,558		-			3,558
Total current assets		4,718		20,447		25,165		1,250,000			1,275,165
Long Term Assets:											
Property and equipment, net		<u> </u>		908	_	908	_				908
Total assets	\$	4,718	\$	21,355	\$	26,073	\$	1,250,000		\$	1,276,073
LIABILITIES AND STOCKHOLDERS' DEFICIT											
Current liabilities:											
Current maturities of convertible notes payable	\$	-	\$	1,395,000	\$	1,395,000	\$	(1,395,000)	[2]	\$	-
Convertible notes payable, related parties		-		105,000		105,000		(105,000)	[2]		-
Notes payable, related party		61		275,200		275,261		-			275,261
Accounts payable		-		258,426		258,426		-			258,426
Accrued expenses and other liabilities		6,005		49,510		55,515		-			55,515
Current portion of accrued interest		-		352,755		352,755		(352,755)	[2]		-
Accrued interest to related parties		-		116,548		116,548		(38,341)	[2]		78,207
Total current liabilities		6,066		2,552,439		2,558,505	_	(1,891,096)			667,409
Long-term liabilities:											
Convertible notes payable, net of current maturities		-		235,000		235,000		(235,000)	[2]		-
Accrued interest, net of current portion				6,351		6,351		(6,351)	[2]		<u> </u>
Total long-term liabilities				241,351	_	241,351		(241,351)			
Total liabilities		6,066		2,768,790	_	2,799,856		(2,132,447)			667,409
Stockholders' deficit:											
Common stock		6,030		583		6,613		11,500	[1],[2]		18,113
Additional paid in capital		34,270		-		34,270		3,329,299	[1],[2],[3]		3,363,569
Deficit accumulated during the development stage		(41,648)		(2,773,018)		(2,814,666)		41,648	[3]		(2,773,018)
Total stockholders' deficit		(1,348)		(2,772,435)		(2,773,783)	_	3,382,447			608,664
Total liabilities and stockholders' deficit	\$	4,718	\$	21,355	\$	26,073	\$	1,250,000		\$	1,276,073

(A Development Stage Company)
Pro Forma Condensed Combined Statement of Operations
Almah, Inc. Year Ended September 30,2012
Arch Biosurgery, Inc. Year Ended September 30,2012

Other revenues \$ - \$ - \$ - \$ - \$ - \$ - \$ - \$ - \$ - \$ -
General and administrative expenses 36,611 333,503 370,114 - 370,114 Research and development expenses 87,021 87,021 - 87,021 Total operating expenses 36,611 420,524 457,135 - 457,135 Operating loss (36,611) (420,524) (457,135) - (457,135) Other (expense) income: Interest expense - (156,865) 129,670 [2] (27,195) Other income - 478 478 - 478
Research and development expenses 87,021 87,021 - 87,021 Total operating expenses 36,611 420,524 457,135 - 457,135 Operating loss (36,611) (420,524) (457,135) - (457,135) Other (expense) income: Interest expense - (156,865) 129,670 [2] (27,195) Other income - 478 478 - 478
Total operating expenses 36,611 420,524 457,135 - 457,135 Operating loss (36,611) (420,524) (457,135) - (457,135) Other (expense) income: Interest expense - (156,865) 129,670 [2] (27,195) Other income - 478 478 - 478
Operating loss (36,611) (420,524) (457,135) - (457,135) Other (expense) income: Interest expense - (156,865) 129,670 [2] (27,195) Other income - 478 478 - 478
Other (expense) income: Interest expense - (156,865) (156,865) 129,670 [2] (27,195) Other income - 478 478 - 478
Other (expense) income: Interest expense - (156,865) (156,865) 129,670 [2] (27,195) Other income - 478 478 - 478
Interest expense - (156,865) (156,865) 129,670 [2] (27,195) Other income - 478 478 - 478
Interest expense - (156,865) (156,865) 129,670 [2] (27,195) Other income - 478 478 - 478
Other income - 478 478 - 478
470 470
Total other expense - (156,387) - 129,670 129,670
Net loss \$ (36,611) \$ (576,911) \$ (613,522) \$ 129,670 \$ (483,852)
6 (616)211 6 (616)221 6 125)310
Net loss per common share basic and diluted $\qquad (0.01) \qquad (0.50)$
1 Net ross per common shall basic and diffice 1 (0.03)
with the second
Weighted average number of shares outstanding $4,684,082$ $1,165,015$ $17,015,798$

Note 1. Basis of Pro Forma Presentation

The unaudited pro forma condensed combined financial information included herein has been prepared pursuant to the rules and regulations of the United States Securities and Exchange Commission. The unaudited pro forma condensed combined financial information of Arch Therapeutics, Inc. is based on the historical balance sheets of Almah, Inc. and Arch Biosurgery, Inc. as of March 31, 2013 and September 30, 2012, respectively, and the pro forma condensed statements of operations for the six months ended March 31, 2013 and the year ended September 30, 2012, respectively, have been prepared after giving effect to the adjustments and assumptions described below.

Arch Therapeutics employs accounting policies that are in accordance with generally accepted accounting principles. In management's opinion, all material adjustments necessary to reflect fairly the unaudited pro forma financial position and results of operations of Arch Therapeutics have been made. The ongoing activity presented in this unaudited pro forma combined financial information represents Arch Therapeutics' financial position, after giving effect to the Merger.

Note 2. Pro Forma Adjustments

The accompanying unaudited pro forma condensed combined statements of operations have been prepared as if the Merger was completed on the first day of the period presented and the combined balance sheet has been prepared as if the Merger had occurred as of the date presented (September 30, 2012) and reflect the pro forma adjustments as presented below.

- [1] To record the issuance of 2,500,000 shares of common stock for \$1,250,000 in connection with the Coldstream Financing.
- [2] To record the issuance of 9,000,000 shares of common stock to convertible debt holders in the Merger with Arch Biosurgery, Inc. in cancellation of the debt and accrued interest.
- [3] To eliminate the Accumulated Deficit of Almah, Inc.